UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of The Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): June 24, 2014

REXFORD INDUSTRIAL REALTY, INC.

(Exact name of registrant as specified in its charter)

Maryland (State or other jurisdiction of incorporation)

001-36008

46-2024407 (IRS Employer Identification No.)

11620 Wilshire Boulevard, Suite 1000, Los Angeles, California (Address of principal executive offices)

90025 (Zip Code)

Registrant's telephone number, including area code: (310) 966-1680

N/A (Former name or former address, if changed since last report.)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see General Instructions A.2.):

- Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)
- Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
- Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))
- Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Item 2.01 Completion of Acquisition or Disposition of Assets.

On June 27, 2014, Rexford Industrial Realty, Inc. (the "Company"), through its operating partnership, Rexford Industrial Realty, L.P. (the "Operating Partnership"), completed the acquisition of an industrial portfolio consisting of nine properties, aggregating approximately 817,166 square feet (the "Portfolio"), from a third-party seller (the "Seller").

The purchase price for the Portfolio was approximately \$88.5 million, exclusive of brokerage commissions and closing costs. The Company funded the acquisition as follows: (i) \$48.5 million with a new loan secured by certain properties in the Portfolio, as described in more detail in Item 2.03 of this Current Report on Form 8-K (this "Current Report") and (ii) cash from drawing on the Company's revolving credit facility in an amount which, when combined with the brokerage commissions, closing costs and proration adjustments, was approximately \$40.2 million.

The table below sets forth relevant information with respect to the properties in the Portfolio acquired:

Property Address	City	Number of Buildings	Asset Type	Year Built	Rentable Square Feet	Number of Leases	Occupancy(1)		Annualized Base Rent(2)	Percentage of Total Annualized Base Rent(3)	Annualized Base Rent per Square Foot(4)	Ownership Interest
Fluperty Address	City	Dulluligs	Warehouse / Distribution /	Tear Duit	rect	Leases	Occupancy(1)	-	Dase Rent(2)	Dasc Rein(3)	 oquate Fuu(4)	Interest
14955-14971 E Salt Lake Ave.	Industry	1	Manufacturing	1979	126.036	4	100%	\$	853,632	15.7%	\$ 6.77	100%
3880 W. Valley Blvd.	Pomona	1	Manufacturing / Distribution	1980	108,703	1	100%	\$	705,888	13.0%	\$ 6.49	100%
5235 E. Hunter Ave.	Anaheim	1	Manufacturing / Office	1987	119,692	3	100%	\$	842,448	15.5%	\$ 7.04	100%
9340 Cabot Drive	San Diego	1	Manufacturing / Distribution	1975/6	86,564	2	84%	\$	576,900	10.6%	\$ 6.66	100%
9404 Cabot Drive	San Diego	1	Distribution	1975/6	46,846	1	100%	\$	452,436	8.3%	\$ 9.66	100%
9455 Cabot Drive	San Diego	1	Distribution	1975/6	96,840	1	84%	\$	651,240	12.0%	\$ 6.72	100%
9755 Distribution Ave.	San Diego	1	Warehouse / Distribution	1974	47,666	2	100%	\$	336,792	6.2%	\$ 7.07	100%
9855 Distribution Ave.	San Diego	1	Warehouse / Distribution	1983	60,819	2	100%	\$	537,756	9.9%	\$ 8.84	100%
1601 Alton Parkway	Irvine	1	R&D/Manufacturing	1974	124,000	1	40%	\$	467,892	8.6%	\$ 3.77	100%
•		9	•		817,166	17	87%	\$	5,424,984	100%	\$ 7.61	

⁽¹⁾ Represents the occupancy at such property as of June 27, 2014.
(2) Calculated as monthly contracted base rent per the terms of the lease(s) at such property for June 2014, multiplied by 12. Excludes rent abatements.
(3) Calculated as annualized see rent for such property divided by annualized base rent for the total Portfolio as of June 27, 2014.
(4) Calculated as annualized base rent for such property divided by leased square feet for such property as of June 27, 2014.

The table below sets forth a summary schedule of lease expirations for leases in place in the Portfolio as of the date of acquisition, plus available space, for each of the calendar years noted below commencing with December 31, 2014. The information set forth in the table assumes that tenant exercise no renewal options and no early termination rights.

	Year of Lease Expiration	Number of Leases Expiring	Total Rentable Square Feet(1)	Annualized Base Rent(2)	Percentage of Annualized Base Rent ⁽³⁾	Ann	ualized Base Rent per Square Foot ⁽⁴⁾
Available		4	104,482	\$ =	0.0%	\$	-
2014		-	-	-	0.0%		-
2015		6	147,962	1,030,932	19.0%		6.97
2016		7	329,534	2,415,792	44.5%		7.33
2017		4	109,475	946,584	17.4%		8.65
2018		-	-	-	0.0%		-
2019		3	104,726	843,816	15.6%		8.06
2020		-	-	-	0.0%		-
2021		-	-	-	0.0%		-
2022		1	20,987	187,860	3.5%		8.95
Total Portfolio)	25	817,166	\$ 5,424,984	100.0%	\$	7.61

(1) Represents the contracted square footage upon expiration.
(2) Calculated as monthly contracted base rent per the terms of such lease(s) for June 2014, multiplied by 12. Excludes rent abatements.
(3) Calculated as annualized base rent set forth in this table divided by annualized base rent for the total Portfolio as of June 27, 2014.
(4) Calculated as annualized base rent for such leases divided by leased square feet for such leases at each of the properties so impacted by the lease expirations as of June 27, 2014.

We believe that existing occupancy and expiring leases in the Portfolio both present the Company with value-add opportunities to reposition certain properties in order to deliver high-demand product capable of achieving higher rents. The Company expects to undertake a value-add capital improvement program at the Alton property in the near future, including dividing a larger space into three dock-high units with new office areas. The Company expects that this capital improvement program will take approximately six to eight months to complete, and has budgeted approximately \$3.0 million for capital expenditures to complete these value-add improvements. At the 9455 Cabot property, the Company expects to undertake a capital improvement program to complete dividing of a vacant space. The Company expects these improvements to take approximately three to five months to complete, and has budgeted approximately \$550,000 for the work. Except for the capital improvement programs noted above, and other than normally recurring capital expenditures and tenant improvements, the Company has no present plans with respect to renovation, improvement or redevelopment of the Portfolio. The Company expects to finance these capital improvement programs from cash on hand and proceeds drawn from its revolving credit facility.

Creation of a Direct Financial Obligation or an Obligation under an Off-Balance Sheet Arrangement of a Registrant.

On June 24, 2014, Rexford Industrial – SDLAOC, LLC, a wholly-owned subsidiary (the "Borrower") of the Operating Partnership entered into a Loan and Security Agreement (the "Loan") by and among the Borrower with JPMorgan Chase Bank, N.A., as Administrative Agent, the Lenders referenced therein, and J.P. Morgan Securities LLC, as Sole Bookrunner and Sole Lead Arranger, for amount equal to \$48.5 million, to partially fund the acquisition of the Portfolio described in Item 2.01 of this Current Report. The Loan has an initial three-year term with one 24-month extension option available, subject to certain conditions, and matures on June 24, 2017. The Loan bears interest, at the Borrower's option, at either (i) LIBOR plus 1.55% or (ii) the alternate base rate plus 0.55%.

The Loan is secured by the first priority deed of trust on eight of the nine properties in the Portfolio and is guaranteed by the Operating Partnership.

The Loan includes a series of customary covenants that the Borrower must comply with, in addition to a performance covenant that is tested annually and requires the achievement of a minimum in-place debt yield of 9.25% by the Properties.

The Loan may be voluntarily prepaid in whole or in part at any time without premium or penalty.

The Loan contains usual and customary events of default including defaults in the payment of principal, interest or fees, defaults in compliance with the covenants set forth in the Loan, cross-defaults to certain other indebtedness, and bankruptcy and other insolvency defaults. If an event of default occurs and is continuing under the Loan, the unpaid principal amount together with all accrued unpaid interest and other amounts owing in respect thereof, may be declared immediately due and payable.

The foregoing summary of the Loan does not purport to be complete and is qualified in its entirety by reference to the Loan, a copy of which is filed as Exhibit 10.1 to this Current Report and is incorporated herein by reference.

On June 26, 2014, the Company elected to draw down a portion of the previously established \$200.0 million revolving credit facility. The Company elected to borrow \$39.5 million at one-month LIBOR plus the applicable margin (the "Facility Borrowing"). The Facility Borrowing was used to fund a portion of the remaining balance of the purchase price of the Portfolio, which transaction funded and closed on June 27, 2014, as disclosed in detail in Item 2.01 of this Current Report.

Upon the one-month maturity date of the Facility Borrowing, the Company may elect to either repay all or a portion of the Facility Borrowing, or convert the Facility Borrowing to a new borrowing at either the then applicable rate plus an applicable margin.

After giving effect to the acquisition of the Portfolio and the related borrowing under the Company's revolving credit facility, there is approximately \$99.9 million drawn under that revolving facility, and an additional \$100.0 million under the related senior unsecured term loan facility. The current borrowing capacity under the Company's revolving credit facility is \$200.0 million; accordingly, after giving effect to the acquisition of the Portfolio, the Company will have approximately \$100.1 million available under the revolving credit facility to fund the above-mentioned capital improvement programs, additional acquisitions and working capital needs.

Item 7.01 Regulation FD Disclosure.

On June 30, 2014, the Company issued a press release announcing the acquisition of the Property. A copy of the press release is furnished with this Current Report as Exhibit 99.1 and is incorporated herein by reference.

The information included in this Current Report under this Item 7.01 (including Exhibit 99.1 hereto) is being "furnished" and shall not be deemed to be "filed" for the purposes of Section 18 of the Securities Exchange Act of 1934, as amended, or otherwise subject to the liabilities of Section 18, nor shall it be incorporated by reference into a filing under the Securities Act of 1933, as amended, or the Securities Exchange Act of 1934, as amended, or the Securities Exchange Act of 1934, as amended, or the Securities Exchange Act of 1934, as amended, or the Securities Exchange Act of 1934, as amended, or the Securities Exchange Act of 1934, as amended, or the Securities Exchange Act of 1934, as amended, or the Securities Exchange Act of 1934, as amended, or the Securities Exchange Act of 1934, as amended, or the Securities Exchange Act of 1935, as amended, or the Securities Exchange Act of 1936, as amended, or the Securities Exchange Act of 1936, as amended, or the Securities Exchange Act of 1937, as amended, or the Securities Exchange Act of 1936, as amended, or the Securities Exchange Act of 1937, as amended, or the Securities Exchange Act of 1936, as amended, or the Securities Exchange Act of 1937, as amended, or the Securities Exchange Act of 1937, as amended, or the Securities Exchange Act of 1938, as amended, or the Securities Exchange Act of 1938, as amended, or the Securities Exchange Act of 1938, as amended, or the Securities Exchange Act of 1938, as amended, or the Securities Exchange Act of 1938, as amended, or the Securities Exchange Act of 1938, as amended, or the Securities Exchange Act of 1938, as amended, or the Securities Exchange Act of 1938, as amended, or the Securities Exchange Act of 1938, as amended, or the Securities Exchange Act of 1938, as amended, or the Securities Exchange Act of 1938, as amended, or the Securities Exchange Act of 1938, as amended, or the Securities Exchange Act of 1938, as amended, or the Securities Exchange Act of 1938, as amended, or the Securities Exchange Act of

Item 9.01 Financial Statements and Exhibits

- (a) Financial statements of businesses acquired. To the extent required by this item, audited financial statements will be filed as an amendment on Form 8-K/A to this report as soon as practicable but no later than 71 calendar days after the latest date on which this initial Form 8-K is required to be filed.
- (b) Pro Forma Financial Information. To the extent required by this item, pro forma financial information will be filed as an amendment on Form 8-K/A to this report as soon as practicable but no later than 71 calendar days after the latest date on which this initial Form 8-K is required to be filed.

(d) Exhibits.

- Loan and Security Agreement, dated as of June 24, 2014, by and among Rexford Industrial SDLAOC, LLC, as borrower, JPMorgan Chase Bank, N.A., as Administrative Agent, the Lenders referenced therein, and J.P. Morgan Securities, LLC, as Sole Bookrunner and Sole Lead Arranger.
- 99.1 Press Release dated June 30, 2014.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, as amended, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

June 30, 2014

June 30, 2014

Rexford Industrial Realty, Inc.

/s/ Michael S. Frankel

Michael S. Frankel Co-Chief Executive Officer (Principal Executive Officer)

Rexford Industrial Realty, Inc.

/s/ Howard Schwimmer

Howard Schwimmer Co-Chief Executive Officer (Principal Executive Officer)

EXHIBIT INDEX

Exhibit
NumberDescription10.1Loan and Security Agreement, dated as of June 24, 2014, by and among Rexford Industrial – SDLAOC, LLC, as borrower, JPMorgan Chase Bank, N.A., as Administrative Agent, the Lenders referenced therein, and J.P. Morgan Securities, LLC, as Sole Bookrunner and Sole Lead Arranger.99.1Press Release dated June 30, 2014

J.P.Morgan

LOAN AND SECURITY AGREEMENT

DATED AS OF JUNE 24, 2014

BY AND AMONG

REXFORD INDUSTRIAL – SDLAOC, LLC,

AS BORROWER,

JPMORGAN CHASE BANK, N.A.,

AS ADMINISTRATIVE AGENT,

THE LENDERS REFERENCED HEREIN,

AND

J.P. MORGAN SECURITIES LLC AS SOLE BOOKRUNNER AND SOLE LEAD ARRANGER

JPMORGAN REAL ESTATE BANKING

	<u>Page</u>
DEFINED TERMS	1
Definitions	1
CONDITIONS TO DISBURSEMENT	18
Generally	18
Conditions to Closing and Advance	19
LOAN TERMS	21
Loan and Advance	21
Funding of Loan Advance	22
Interest Elections	22
Repayment of Loans; Evidence of Debt	24
Prepayment of Loans	24
Fees	25
Interest	25
Alternate Rate of Interest	25
Increased Costs	26
Break Funding Payments	27
Taxes	28
Payments Generally; Pro Rata Treatment; Sharing of Set-offs	32
Mitigation Obligations; Replacement of Lenders	33
Extension Option	34
Reconveyance of Property	35
GENERAL COVENANTS	37
Inspection	37
Liens, Taxes, and Governmental Claims	37
Leases	38
Operations of Borrower	40
Management	43
Appraisals	43
Operating and Reserve Accounts	43
	Definitions CONDITIONS TO DISBURSEMENT Generally Conditions to Closing and Advance LOAN TERMS Loan and Advance Funding of Loan Advance Funding of Loan Advance Interest Elections Repayment of Loans; Evidence of Debt Prepayment of Loans; Fees Interest Alternate Rate of Interest Increased Costs Freak Funding Payments Taxes Payments Generally; Pro Rata Treatment; Sharing of Set-offs Mitigation Obligations; Replacement of Lenders Extension Option Reconveyance of Property GENERAL COVENANTS Inspection Liens, Taxes, and Governmental Claims Lease Opperations of Borrower Management Appraisals

Prohibited Distributions

<u>Page</u> 43

1.09	Borrower's Right to Contest Legal Requirements	43
4.10	Government Regulation	44
4.11	Financial Information and Other Deliveries	44
4.12	Hazardous Substances	45
4.13	ERISA	46
4.14	Compliance with Laws	47
4.15	Sanctions Laws and Regulations	47
4.16	Debt Yield	47
4.17	Borrower's Funds Account	48
ARTICLE V	COLLATERAL ASSIGNMENT AND SECURITY AGREEMENT	49
5.01	Project Documents and Swap Agreements	49
5.02	Additional Instruments	50
ARTICLE VI	REPRESENTATIONS AND WARRANTIES	50
5.01	Representations and Warranties	50
5.02	Nature of Representations and Warranties	53
ARTICLE VII	INSURANCE AND CONDEMNATION	53
7.01	Insurance and Casualty	53
7.02	Condemnation and Other Awards	58
ARTICLE VIII	DEFAULTS	59
3.01	Defaults	59
ARTICLE IX	ACCELERATION AND REMEDIES	63
9.01	Acceleration	63
9.02	Other Remedies	63
9.03	Curing of Defaults	64
ARTICLE X	THE ADMINISTRATIVE AGENT	64
10.01	Appointment	64
10.02	Capacity as Lender	64
10.03	Duties and Obligations	65
10.04	Reliance	65
10.05	Sub-Agents	65
10.06	Resignation	66
10.07	Independent Credit Analysis	66

		<u>Page</u>
10.08	Lender Actions Against Collateral	67
10.09	Lender Reply Period	67
10.10	Foreclosure	67
10.11	Defaulting Lender	69
10.12	Compliance with Flood Laws	71
10.13	Borrower's Rights	71
ARTICLE XI	MISCELLANEOUS	71
11.01	Notices	71
11.02	Waivers; Amendments	73
11.03	Expenses; Indemnity; Damage Waiver	75
11.04	Successors and Assigns	76
11.05	Survival	80
11.06	Counterparts; Integration; Effectiveness	80
11.07	Severability	81
11.08	Right of Setoff	81
11.09	Governing Law; Jurisdiction; Consent to Service of Process	81
11.10	WAIVER OF JURY TRIAL	82
11.11	WAIVER OF SPECIAL DAMAGES	84
11.12	Headings	84
11.13	Confidentiality	84
11.14	Interest Rate Limitation	85
11.15	USA Patriot Act	85
11.16	Administrative Agent Approvals	86
11.17	Replacement Documentation	86
11.18	Swap Agreements	86

EXHIBITS

Exhibit F-4

Exhibit I

Schedule 1.01 Lenders

Exhibit A-1 Legal Description - 9404 Cabot Property Exhibit A-2 Legal Description – 9455 Cabot Property Exhibit A-3 Legal Description – 9340 Cabot Property Exhibit A-4 Legal Description – 9855 Distribution Property Exhibit A-5 $Legal\ Description-9755\ Distribution\ Property$ Exhibit A-6 Legal Description – 14955-14791 E. Salt Lake Property

Exhibit A-7 Legal Description – 5235 E. Hunter Property Exhibit A-8 Legal Description - 3880 W. Valley Property

Exhibit B Promissory Note Exhibit C Assignment and Assumption

Exhibit D Limited Recourse and Guaranty Provisions

Exhibit E Approved Lease Terms

Form of U.S. Tax Compliance Certificate (For Foreign Lenders That Are Not Partnerships For U.S. Federal Income Tax Purposes)

Exhibit F-1

Form of U.S. Tax Compliance Certificate (For Foreign Participants That Are Not Partnerships For U.S. Federal Income Tax Purpose Form of U.S. Tax Compliance Certificate (For Foreign Participants That Exhibit F-2

Physical Condition Reports

Exhibit F-3

Are Partnerships For U.S. Federal Income Tax Purposes)
Form of U.S. Tax Compliance Certificate (For Foreign Lenders That

Are Partnerships For U.S. Federal Income Tax Purposes)

Exhibit G Form of Compliance Certificate Exhibit H Environmental Reports

-ii-

LOAN AND SECURITY AGREEMENT

THIS LOAN AND SECURITY AGREEMENT (the "Agreement") dated as of this 24th day of June, 2014, is by and among REXFORD INDUSTRIAL – SDLAOC, LLC, a Delaware limited liability company ("Borrower"), JPMORGAN CHASE BANK, N.A., ("JPMorgan") in its capacity as Administrative Agent (as hereinafter defined), and the Lenders (as hereinafter defined).

RECITALS

WHEREAS, Borrower is acquiring all of that certain real property located at (a) 9404 Cabot Drive, San Diego, California, more particularly described in Exhibit A-1 attached hereto and incorporated herein by this reference (the "9404 Cabot Property"), (b) 9404 Cabot Drive, San Diego, California, more particularly described in Exhibit A-2 attached hereto and incorporated herein by this reference (the "9455 Cabot Property"), (c) 9340 Cabot Drive, San Diego, California, more particularly described in Exhibit A-3 attached hereto and incorporated herein by this reference (the "9350 Distribution Avenue, San Diego, California, more particularly described in Exhibit A-4 attached hereto and incorporated herein by this reference (the "9755 Distribution Avenue, San Diego, California, more particularly described in Exhibit A-5 attached hereto and incorporated herein by this reference (the "9755 Distribution Property"), (e) 9755 Distribution Avenue, San Diego, California, more particularly described in Exhibit A-6 attached hereto and incorporated herein by this reference (the "44955 E. Salt Lake Property"), (g) 5235 E. Hunter Avenue, Anaheim, California, more particularly described in Exhibit A-8 attached hereto and incorporated herein by this reference (the "5355 E. Hunter Property"), and (h) 3880 W. Valley Boulevard, Pomona, California, more particularly described in Exhibit A-8 attached hereto and incorporated herein by this reference (the "3850 W. Valley Property"); and

WHEREAS, Borrower has requested, and the Lenders have agreed to provide, financing to Borrower to fund part of the cost of acquiring the Property on the terms and conditions set forth herein.

AGREEMENT

NOW, THEREFORE, for good and valuable consideration, the receipt and adequacy of which is hereby acknowledged, the parties hereto agree as follows:

ARTICLE I DEFINED TERMS

1.01 Definitions.

The following terms shall have the following meanings:

"3880 W. Valley Property" has the meaning set forth in the Recitals.

"5235 E. Hunter Property" has the meaning set forth in the Recitals.

1

- "9340 Cabot Property" has the meaning set forth in the Recitals.
- "9404 Cabot Property" has the meaning set forth in the Recitals.
- "9455 Cabot Property" has the meaning set forth in the Recitals.
- "9755 Distribution Property" has the meaning set forth in the Recitals.
- "9855 Distribution Property" has the meaning set forth in the Recitals.
- "14955 E. Salt Lake Property" has the meaning set forth in the Recitals.

the next 1/16 of 1%) equal to (a) the LIBO Rate for such Interest Period multiplied by (b) the Statutory Reserve Rate.

"ABR," when used in reference to any Loan or Advance, refers to whether such Loan, or the Loans comprising such Advance, are bearing interest at a rate determined by reference to the Alternate Base Rate.

"Adjusted LIBO Rate" means, with respect to any Eurodollar Advance for the relevant Interest Period, or for any Floating Rate Advance, an interest rate per annum (rounded upwards, if necessary, to

- "Administrative Agent" means JPMorgan, in its capacity as administrative agent for the Lenders hereunder.
- "Administrative Questionnaire" means an Administrative Questionnaire in a form supplied by Administrative Agent.
- "Advance" means a borrowing hereunder, (a) made by the Lenders on the same Borrowing Date, or (b) converted or continued by the Lenders on the same date of conversion or continuation, consisting, in either case, of the aggregate amount of the several Loans of the same Type and, in the case of Eurodollar Loans, for the same Interest Period.
- "Affiliate" means, with respect to a specified Person, another Person that directly, or indirectly through one or more intermediaries, Controls or is Controlled by or is under common Control with the Person specified.
- "Aggregate Commitment" means, as of any date of determination, the aggregate of the Commitments of all the Lenders, as such amount may have been reduced pursuant to Section 3.14(b) hereof. The Aggregate Commitment shall be reduced on a dollar-for-dollar basis by payments of principal actually received by the Agent. As of the date hereof, the Aggregate Commitment is Forty-Eight Million Five Hundred Thousand and No/100 Dollars (\$48,500,000.00).
 - "Agreement" has the meaning assigned to it in the Preamble.
- "Alternate Base Rate" means, for any day, a rate per annum equal to the greatest of (a) the Prime Rate in effect on such day, (b) the Federal Funds Effective Rate in effect on such day plus ½ of 1% and (c) the Adjusted LIBO Rate for a one month Interest Period on such day

(or if such day is not a Business Day, the immediately preceding Business Day) plus 1% (without any rounding). Any change in the Alternate Base Rate due to a change in the Prime Rate, the Federal Funds Effective Rate or the Adjusted LIBO Rate shall be effective from and including the effective date of such change in the Prime Rate, the Federal Funds Effective Rate or the Adjusted LIBO Rate, respectively. If the Alternate Base Rate is being used as an alternate rate of interest pursuant to Section 3.08 hereof, then the Alternate Base Rate shall be the greater of clause (a) and (b) above and shall be determined without reference to clause (c) above.

"Annual Debt Service" means, as of any date of determination, annual debt service on a loan with a principal amount equal to the Aggregate Commitment on such date of determination, assuming (a) a fixed rate of interest per annum equal to the greater of (i) the Eurodollar Rate as of such date of determination for a loan in such amount with an Interest Period equal to one (1) month, or (ii) the annual yield to maturity as of such date of determination of a ten (10) year United States Treasury Note in the amount of such loan plus two percent (2.0%) per annum, and (b) amortization of such loan in equal annual payments of principal and interest over a period of thirty (30) years.

"Applicable Margin" means 0.55% with respect to the Floating Rate and 1.55% with respect to the Eurodollar Rate.

"Applicable Percentage" means, with respect to any Lender, the percentage of the Aggregate Commitment represented by such Lender's Commitment; provided that in the case of Section 10.11 when a Defaulting Lender shall exist, "Applicable Percentage" means the percentage of the Aggregate Commitment (disregarding any Defaulting Lender's Commitment) represented by such Lender's Commitment. If the Commitments have terminated or expired, the Applicable Percentages shall be determined based upon the Commitments most recently in effect, giving effect to any assignments.

"Appraisal" means a written statement setting forth an opinion of the market value of the Mortgaged Property that (i) has been independently and impartially prepared by a qualified appraiser directly engaged by Administrative Agent, (ii) complies with all applicable federal and state laws and regulations dealing with appraisals or valuations of real property, and (iii) has been reviewed as to form and content and approved by Administrative Agent, in its reasonable discretion.

"Appraised Value" means, with respect to each Property, "as-is" value of such Property, as determined by the Required Lenders based upon their review of the Appraisals.

"Approved Fund" means any Person (other than a natural person) that is engaged in making, purchasing, holding or investing in bank loans and similar extensions of credit in the ordinary course of its business and that is administered or managed by (a) a Lender, (b) an Affiliate of a Lender or (c) an entity or an Affiliate of an entity that administers or manages a Lender.

"Approved Lease" means any Lease which (a) is substantially in the form of the Form Lease and (b) if a Major Lease, satisfies the requirements set forth on Exhibit E attached hereto. A Lease shall be substantially in the form of the Form Lease if the variations from the Form Lease (taken as a whole) do not materially detract from the rights and protections of the

lessor set forth in the Form Lease.

"Approved Manager" means Rexford Industrial Realty, L.P., a Maryland limited partnership, or any other reputable and creditworthy property manager, subject to the prior written approval of Administrative Agent, which written approval may be evidenced by e-mail confirmation, not to be unreasonably withheld, with a portfolio of properties comparable to the Mortgaged Property under active management.

"Assignment and Assumption" means an assignment and assumption entered into by a Lender and an assignee (with the consent of any party whose consent is required by Section 11.04 hereof), and accepted by Administrative Agent, in the form of Exhibit C attached hereto or any other form approved by Administrative Agent.

"Bankruptcy Event" means, with respect to any Person, such Person becomes the subject of a bankruptcy or insolvency proceeding, or has had a receiver, conservator, trustee, administrator, custodian, assignee for the benefit of creditors or similar Person charged with reorganization or liquidation of its business, appointed for it, or, in the good faith determination of the Administrative Agent, has taken any action in furtherance of, or indicating its consent to, approval of, or acquiescence in, any such proceeding or appointment, provided that a Bankruptcy Event shall not result solely by virtue of any ownership interest, or the acquisition of any ownership interest, in such Person by a Governmental Authority or instrumentality thereof, provided, further, that such ownership interest does not result in or provide such Person with immunity from the jurisdiction of courts within the United States or from the enforcement of judgments or writs of attachment on its assets or permit such Person (or such Governmental Authority or instrumentality), to reject, repudiate, disavow or disaffirm any contracts or agreements made by such Person.

"Board" means the Board of Governors of the Federal Reserve System of the United States of America.

"Borrower" has the meaning set forth in the preamble.

"Borrower Financing Statement" means the UCC financing statement covering the security interests in personal property granted by Borrower to Administrative Agent for the benefit of the Lenders, in the Loan Documents for filling with the Secretary of State of the State of Delaware.

"Borrower's Funds Account" means a deposit account maintained by Borrower with Administrative Agent pursuant to Section 4.17 hereof.

"Borrowing Date" means a date on which an Advance is made hereunder.

"Business Day" means any day that is not a Saturday, Sunday or other day on which commercial banks in New York City are authorized or required by law to remain closed; provided that, when used in connection with a Eurodollar Advance, the term "Business Day" shall also exclude any day on which banks are not open for dealings in dollar deposits in the London interbank market.

"Cabot Project" means the 9340 Cabot Property, the 9404 Cabot Property and the 9455 Cabot Property, collectively.

"Capital Lease Obligations" of any Person means the obligations of such Person to pay rent or other amounts under any lease of (or other arrangement conveying the right to use) real or personal property, or a combination thereof, which obligations are required to be classified and accounted for as capital leases on a balance sheet of such Person under GAAP, and the amount of such obligations shall be the capitalized amount thereof determined in accordance with GAAP.

"Certification of Non-Foreign Status" means an affidavit, signed under penalty of perjury by an authorized officer of Borrower, stating (a) that Borrower is not a "foreign corporation," "foreign partnership," "foreign trust," or "foreign estate," as those terms are defined in the Code and the regulations promulgated thereunder, (b) Borrower's U.S. employer identification number, and (c) the address of Borrower's principal place of business. Such affidavit shall be consistent with the requirements of the regulations promulgated under Section 1445 of the Code, and shall otherwise be in form and substance acceptable to Administrative Agent.

"Change in Law" means the occurrence after the date of this Agreement or, with respect to any Lender, such later date on which such Lender becomes a party to this Agreement, of: (a) the adoption of any law, rule, regulation or treaty, (b) any change in any law, rule, regulation or treaty or in the interpretation or application thereof by any Governmental Authority or (c) compliance by any Lender (or, for purposes of Section 3.09(b), by any lending office of such Lender or by such Lender's holding company, if any) with any request, guideline or directive (whether or not having the force of law) of any Governmental Authority made or issued after the date of this Agreement; provided that notwithstanding anything herein to the contrary, (i) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines or directives thereunder, issued in connection therewith or in implementation thereof, and (ii) all requests, rules, guidelines and directives promulgated by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities, in each case pursuant to Basel III, shall be deemed to be a "Change in Law," regardless of the date enacted, adopted, issued or implemented.

"Charges" has the meaning set forth in Section 11.13 hereof.

"Closing Date" is the earlier of the date of the disbursement of the Loan or the date all the conditions to such disbursement have been satisfied, including recording the Deed of Trust.

"Code" means the Internal Revenue Code of 1986, as amended from time to time.

"Commitment" means, for each Lender, the obligations of such Lender to make Loans not exceeding the amount set forth on Schedule 1.01 or as set forth in any assignment agreement that has become effective pursuant to Section 11.04 hereof, as such amount may be modified from time to time pursuant to the terms hereof.

"Connection Income Taxes" means Other Connection Taxes that are imposed on or measured by net income (however denominated) or that are franchise Taxes or branch profits Taxes.

"Control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ability to exercise voting power, by contract or otherwise. "Controlling" and "Controlled" have meanings correlative thereto.

"Credit Party" means the Administrative Agent or any other Lender.

"Debt Service Coverage Ratio" means, as of any determination date, the ratio of NOI as of such determination date to Annual Debt Service as of such determination date.

"Debt Yield Ratio" means, as of any date of determination, the ratio, expressed as a percentage, of (a) NOI as of such determination date, to (b) the Aggregate Commitment in effect as of such

determination date

"Debtor Relief Laws" means any applicable liquidation, conservatorship, bankruptcy, moratorium, rearrangement, insolvency, fraudulent conveyance, reorganization, or similar laws affecting the rights, remedies or recourse of creditors generally, including without limitation, the United States Bankruptcy Code and all amendments thereto, as are in effect from time to time during the term of the Loans.

"Deed of Trust" means, individually and collectively, each Deed of Trust, Assignment of Leases and Rents, Security Agreement and Fixture Filing of even date herewith executed by Borrower in favor of Administrative Agent, for the benefit of the Lenders, as amended from time to time, covering a Property.

"Default" has the meaning assigned to it in Section 8.01 hereof.

"Defaulting Lender" means any Lender that (a) has failed, within two Business Days of the date required to be funded or paid, to (i) fund any portion of its Loan, or (ii) pay over to any Credit Party any other amount required to be paid by it hereunder, unless in the case of clause (i) above, such Lender notifies the Administrative Agent in writing fat such failure is the result of such Lender's good faith determination that a condition precedent to funding (specifically identified and including the particular Default, if any) has not been satisfied; (b) has notified Borrower or any Credit Party in writing, or has made a public statement, to the effect that it does not intend or expect to comply with any of its funding obligations under this Agreement (unless such writing or public statement indicates that such position is based on such Lender's good faith determination that a condition precedent to funding (specifically identified and including the particular Default, if any) cannot be satisfied) or generally under other agreements in which it commits to extend credit; (c) has failed, within three Business Days after request by a Credit Party, acting in good faith, to provide a certification in writing from an authorized officer of such Lender that it will comply with its obligations to fund under this Agreement, provided that such Lender shall cease to be a Defaulting Lender pursuant to this clause (c), upon such Credit Party's receipt of such certification in form and substance satisfactory to it and the Administrative Agent; or (d) has become the subject of a Bankruptey

Event.

"Designated Persons" means a person or entity (a) listed in the annex to, or otherwise subject to the provisions of, any Executive Order; (b) named as a "Specially Designated National and Blocked Person" ("SDN") on the most current list published by OFAC at its official website or any replacement website or other replacement official publication of such list (the "SDN List") or is otherwise the subject of any Sanctions Laws and Regulations; (c) in which an entity or person on the SDN List has 50% or greater ownership interest or that is otherwise controlled by an SDN.

"dollars" or "\$" refers to lawful money of the United States of America.

"Electronic Signature" means an electronic sound, symbol, or process attached to, or associated with, a contract or other record and adopted by a person with the intent to sign, authenticate or accept

such contract or record.

"Electronic System" means any electronic system, including e-mail, e-fax, Intralinks®. ClearPar®, Dropbox, ShareFile, and any other Internet or extranet-based site, whether such electronic system is owned, operated or hosted by the Administrative Agent or any other Person, providing for access to data protected by passcodes or other security system.

"Embargoed Person" has the meaning set forth in Section 6.01(m).

"Employee Benefit Plan" means an employee benefit plan as defined in Section 3(3) of ERISA, maintained, sponsored by or contributed to by Borrower or any ERISA Affiliate.

"Environmental Indemnity Agreement" means, individually and collectively, each separate Environmental Indemnity Agreement of even date herewith executed by Borrower and Guarantor in favor of Administrative Agent, for the benefit of the Lenders, as amended from time to time.

"Environmental Laws" means any local, state or federal law, rule (having the effect of law), regulation or order (having the effect of law) relating to the manufacture, storage, use, handling, discharge, transport, disposal, treatment or clean-up of hazardous or toxic substances or materials, including, without limitation, "CERCLA", "RCRA", or state superlien or environmental clean-up statutes.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended from time to time.

"ERISA Affiliate" means Borrower or any corporation, trade or business that along with Borrower is treated as a single employer under Sections 414(b), 414(c), 414(m) or 414(o) of the Code.

"Eurodollar Advance" means an Advance which, except as otherwise provided in Section 3.07(c) hereof, bears interest at the applicable Eurodollar Rate.

"Eurodollar Loan" means a Loan which, except as otherwise provided in Section 3.07(c) hereof, bears interest at the applicable Eurodollar Rate.

"Eurodollar Rate" means, with respect to a Eurodollar Advance for the relevant Interest Period, a per annum rate of interest equal to the Adjusted LIBO Rate for such Interest Period plus the Applicable

Margin.

"Excess Cash Flow" means, with respect to each period of determination, an amount equal to (a) the aggregate income, revenues, reimbursements and receipts of any kind whatsoever generated during such period from the Property, less (b) the aggregate expenses incurred during such period in connection with the operation of the Property, less (c) the interest, principal fees and other charges actually paid by Borrower to Administrative Agent with respect to the Loan during such period.

"Excluded Taxes" means any of the following Taxes imposed on or with respect to a Recipient or required to be withheld or deducted from a payment to a Recipient: (a) Taxes imposed on or measured by net income (however denominated), franchise Taxes, and branch profits Taxes, in each case, (i) imposed as a result of such Recipient being organized under the laws of, or having its principal office or, in the case of any Lender, its applicable lending office located in, the jurisdiction imposing such Tax (or any political subdivision thereof) or (ii) that are Other Connection Taxes, (b) in the case of a Lender, U.S. Federal withholding Taxes imposed on amounts payable to or for the account of such Lender with respect to an applicable interest in a Loan or Commitment pursuant to a law in effect on the date on which (i) such Lender acquires such interest in the Loan or Commitment (other than pursuant to an assignment request by Borrower under Section 3.13(b)) or (ii) such Lender changes its lending office, except in each case to the extent that, pursuant to Section 3.11, amounts with respect to such Taxes were payable either to such Lender's assignor immediately before such Lender acquired the applicable interest in a Loan or Commitment or to such Lender immediately before it changed its lending office, (c) Taxes attributable to such Recipient's failure to comply with Section 3.11(f) and (d) any U.S. Federal withholding Taxes imposed under FATCA.

"Executive Order" has the meaning assigned to it in the definition of Sanctions Law and Regulations.

"Extended Maturity Date" has the meaning set forth in Section 3.14(a) hereof.

"FATCA" means Sections 1471 through 1474 of the Code, as of the date of this Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with), any current or future regulations or official interpretations thereof and any agreement entered into pursuant to Section 1471(b)(1) of the Code.

"Federal Funds Effective Rate" means, for any day, the weighted average (rounded upwards, if necessary, to the next 1/16 of 1%) of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers, as published on the next succeeding Business Day by the Federal Reserve Bank of New York, or, if such rate is not so published for any day that is a Business Day, the average (rounded upwards, if

necessary, to the next 1/16 of 1%) of the quotations for such day for such transactions received by the Administrative Agent from three Federal funds brokers of recognized standing selected by it.

- "Fee Letter" means that certain letter agreement of even date herewith "Fee Letter" by and among JPMorgan Chase Bank, N.A. and Borrower.
- "Floating Rate" means, for any day, a rate per annum equal to the Alternate Base Rate for such day plus the Applicable Margin for such day.
- "Floating Rate Advance" means an Advance which, except as otherwise provided in Section 3.07(c) hereof, bears interest at the Floating Rate.
- "Floating Rate Loan" means a Loan which, except as otherwise provided in Section 3.07(c) hereof, bears interest at the Floating Rate.
- "Flood Laws" means the National Flood Insurance Act of 1968, the Flood Disaster Protection Act of 1973, the National Flood Insurance Reform Act of 1994, the Biggert-Waters Flood Insurance Act of 2012, as such statues may be amended or re-codified from time to time, any substitution therefor, any regulations promulgated thereunder, and all other Legal Requirements relating to flood insurance.
 - "Foreign Assets Control Regulations" has the meaning assigned to it in Section 6.01(m).
- "Foreign Lender" means (a) if Borrower is a U.S. Person, a Lender, with respect to such Borrower, that is not a U.S. Person, and (b) if Borrower is not a U.S. Person, a Lender, with respect to such Borrower, that is resident or organized under the laws of a jurisdiction other than that in which Borrower is resident for tax purposes.
 - "Form Lease" means the form of lease for the Improvements to be reviewed and approved by Administrative Agent.
 - "Future Commitment" has the meaning assigned to it in Section 10.11(d) hereof.
 - "GAAP" means generally accepted accounting principles in the United States of America.
- "Governmental Authority" means the government of the United States of America, any other nation or any political subdivision thereof, whether state or local, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government.
- "Guarantee" of or by any Person (the "guarantor") means any obligation, contingent or otherwise, of the guarantor guaranteeing or having the economic effect of guaranteeing any Indebtedness or other obligation of any other Person (the "primary obligor") in any manner, whether directly or indirectly, and including any obligation of the guarantor, direct

or indirect, (a) to purchase or pay (or advance or supply funds for the purchase or payment of) such Indebtedness or other obligation or to purchase (or to advance or supply funds for the purchase of) any security for the payment thereof, (b) to purchase or lease property, securities or services for the purpose of assuring the owner of such Indebtedness or other obligation of the payment thereof, (c) to maintain working capital, equity capital or any other financial statement condition or liquidity of the primary obligor so as to enable the primary obligor to pay such Indebtedness or other obligation; provided, that the term Guarantee shall not include endorsements for collection or deposit in the ordinary course of business.

"Guarantor" means REXFORD INDUSTRIAL REALTY, LP, a Maryland limited partnership.

"Hazardous Substances" means and includes all hazardous and toxic substances, wastes or materials, hydrocarbons (including naturally occurring or man-made petroleum and hydrocarbons), flammable explosives, urea formaldehyde insulation, radioactive materials, biological substances and any other kind and/or type of pollutants or contaminates (including, without limitation, asbestos and raw materials which include hazardous constituents), sewage sludge, industrial slag, solvents and/or any other similar substances, or materials which are included under or regulated by any Environmental Laws; provided, however, that "Hazardous Substances" shall not include (a) materials customarily used in the construction and demolition of buildings, or (b) cleaning materials and office products customarily used in the operation of properties such as the Mortgaged Property, to the extent such materials described in the preceding clauses (a) and (b) are stored, handled, used and disposed of in compliance with all Environmental Laws.

"Improvements" means all buildings, structures and improvements now or hereafter located on any portion of the Property.

"Indebtedness" of any Person means, without duplication, (a) all obligations of such Person for borrowed money or with respect to deposits or advances of any kind, (b) all obligations of such Person evidenced by bonds, debentures, notes or similar instruments, (c) all obligations of such Person upon which interest charges are customarily paid, (d) all obligations of such Person under conditional sale or other title retention agreements relating to property acquired by such Person, (e) all obligations of such Person in respect of the deferred purchase price of property or services (excluding current accounts payable incurred in the ordinary course of business), (f) all Indebtedness of others secured by (or for which the holder of such Indebtedness has an existing right, contingent or otherwise, to be secured by) any Lien on property owned or acquired by such Person, whether or not the Indebtedness secured thereby has been assumed, (g) all Guarantees by such Person of Indebtedness of others, (h) all Capital Lease Obligations of such Person, (i) all obligations, contingent or otherwise, of such Person in respect of bankers' acceptances, and (k) all Swap Obligations. The Indebtedness of any Person shall include the Indebtedness of any other entity (including any partnership in which such Person is a general partner) to the extent such Person is liable therefor as a result of such Person's ownership interest in or other relationship with such entity, except to

the extent the terms of such Indebtedness provide that such Person is not liable therefor.

- "Indemnified Taxes" means (a) Taxes, other than Excluded Taxes, imposed on or with respect to any payment made by or on account of any obligation of Borrower under any Loan Document and (b) to the extent not otherwise described in (a), Other Taxes.
 - "Indemnitee" has the meaning set forth in Section 11.03(b) hereof.
 - "Initial Maturity Date" means June 24, 2017.
 - "Interest Election Request" means a request by Borrower to convert or continue an Advance in accordance with Section 3.03 hereof.
 - "Interest Payment Date" means the fifth (5th) day of each month.
- "Interest Period" means with respect to any Eurodollar Advance, the period commencing on the date of such Eurodollar Advance and ending on the numerically corresponding day in the calendar month that is one, two, three or six months thereafter, as Borrower may elect; provided, that (a) if any Interest Period would end on a day other than a Business Day, such Interest Period shall be extended to the next succeeding Business Day and (ii) any Interest Period shall in the next calendar month, in which case such Interest Period shall end on the next preceding Business Day and (ii) any Interest Period that commences on the last Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the last calendar month of such Interest Period) shall end on the last Business Day of the last calendar month of such Interest Period. For purposes hereof, the date of an Advance initially shall be the date on which such Advance is made and thereafter shall be the effective date of the most recent conversion or continuation of such Advance.
 - "IRS" means the United States Internal Revenue Service.
 - "JPMorgan" has the meaning set forth in the preamble.
 - "Lease" means any lease or other agreement for the use and occupancy of all or any portion of the Improvements, whether now in existence or hereafter arising.
- "Legal Requirements" means any and all judicial decisions, statutes, rulings, directions, rules, regulations, permits, certificates or ordinances of any Governmental Authority in any way applicable to Borrower or the Property or Improvements, including, without limitation, the ownership, division, use, occupancy, possession, operation, maintenance, alteration, repair or reconstruction thereof.
 - "Lender Reply Period" has the meaning assigned to it in Section 10.09 hereof.
- "Lenders" means the Persons listed on Schedule 1.01 and any other Person that shall have become a party hereto pursuant to an Assignment and Assumption, other than any such Person that ceases to be a party hereto pursuant to an Assignment and Assumption.
 - "Lessee" means a tenant under a Lease.

"LIBO Rate" means, with respect to any Eurodollar Loan for any Interest Period, the London interbank offered rate as administrated by the ICE Benchmark Association (or any other Person that takes over the administration of such rate) for U.S. Dollars for a period equal in length to such Interest Period as displayed on page LIBOR01 of the Reuters Screen (or on any successor or substitute page on such screen, or on the appropriate page of such other information service that publishes such rate from time to time in place of Reuters; in each case as the "Screen Rate") at approximately 11:00 a.m., London time, two Business Days prior to the commencement of such Interest Period; provided, that, if the Screen Rate shall not be available for such Interest Period then, such Loan shall accrue interest at the Floating Rate and be made as a Floating Rate Loan. The LIBO Rate shall never be less than zero percent (0%).

"Lien" means, with respect to any asset, (a) any mortgage, deed of trust, lien, pledge, hypothecation, encumbrance, charge or security interest in, on or of such asset, (b) the interest of a vendor or a lessor under any conditional sale agreement, capital lease or title retention agreement (or any financing lease having substantially the same economic effect as any of the foregoing) relating to such asset and (c) in the case of securities, any purchase option, call or similar right of a third party with respect to such securities.

"Limited Guaranty" means the Limited Guaranty of even date herewith executed by Guarantor in favor of Administrative Agent, for the benefit of the Lenders, as amended from time to time, evidencing guaranties by the Guarantor with respect to certain acts of Borrower, including without limitation, as described on Exhibit D.

"Loan" or "Loans" means, with respect to any Lender(s), any loans made by such Lender(s) pursuant to this Agreement (or any conversion or continuation thereof).

"Loan Documents" means this Agreement, the Notes, the Deeds of Trust, the Limited Guaranty, the Environmental Indemnity Agreement, the Fee Letter and any and all other documents now or hereafter executed by Borrower, Guarantor or any other guarantor of the Obligations or any portion thereof evidencing, guarantying, securing or otherwise pertaining to the Obligations; provided. however, that Swap Agreements between Borrower and any Lender or Affiliate of any Lender shall not constitute Loan Documents.

"Loan-to-Value Ratio" means the ratio, expressed as a percentage, of (a) the Aggregate Commitment to (b) the aggregate Appraised Value of the Property.

"Major Lease" means any Lease demising more than thirty-three percent (33%) of the net rentable square footage in the applicable Improvements.

"Material Borrower Indebtedness" has the meaning set forth in Section 8.01(t) hereof.

"Material Guarantor Indebtedness" has the meaning set forth in Section 8.01(q) hereof.

"Maturity Date" means the Initial Maturity Date as such date may be extended pursuant to Section 3.14 hereof.

"Maximum Rate" has the meaning set forth in Section 11.13 hereof.

"Minimum Release Price" means, with respect to each Property, the greatest of (a) a sum such that, after giving effect to the requested release of such Property and the payment of the subject release price, (i) the then Remaining Property shall have an aggregate Loan-to-Value Ratio of not more than sixty-five percent (65%) (as determined by Administrative Agent and if required by Administrative Agent, based on a new Appraisal timely obtained by Administrative Agent, and (ii) each then Remaining Property shall have a Loan-to-Value Ratio (based on the Par Loan Amount allocated to such Remaining Property pursuant to the table set forth below) of not more than sixty-five percent (65%) (as determined by Administrative Agent, based on a new Appraisal timely obtained by Administrative Agent), (b) a sum such that, after giving effect to the requested release of such Property and the payment of the subject release price, the then Remaining Property shall satisfy, on an aggregate bases, (i) a Debt Service Coverage Ratio of not less than 1.25:1.00, and (ii) a Debt Yield Ratio of at least 9.25%, and (c) the "Par Loan Amount" specified in the chart below with respect to such Property:

<u>Property</u> 3880 W. Valley Property	<u>-</u> \$	<u>Par Loan Amount</u> 6,299,117.00
5235 E. Hunter Property	\$	7,354,299.00
Cabot Project	\$	19,185,126.00
9755 Distribution Property	\$	3,574,829.00
9855 Distribution Property	\$	5,052,083.00
14955 E. Salt Lake Property	\$	7.034.546.00

Upon Borrower's request and the delivery to Administrative Agent of an Appraisal setting forth an opinion of the "as-is" value of each of the 9340 Cabot Property, the 9404 Cabot Property and the 9455 Cabot Property and the 9455 Cabot Property, which allocation shall allocate the Par Loan Amount specified above for the Cabot Project to the 9340 Cabot Property, the 9404 Cabot Property and the 9455 Cabot Property based on the relative "as-is" value of each such Property, which allocation shall be conclusive and binding absent manifest error. Administrative Agent shall provide prompt written notice of such allocation to Borrower and Lenders, and thereafter such allocation to the 9340 Cabot Property, the 9404 Cabot Property and the 9455 Cabot Property shall, in each case, be deemed to be the Par Loan Amount applicable to such Property for the purposes of clause (c) above.

"Minimum Rental Rate" means a per square foot, monthly rental rate, for the applicable rental basis specified in the chart below, in an amount not less than the "Minimum Rental Rate" specified in the chart below with respect to such Property:

<u>Property</u>		e Minimum Rental Rate
3880 W. Valley Property	(triple basis) \$0.54	(modified gross basis) N/A
5235 E. Hunter Property	\$0.51	N/A
9340 Cabot Property	\$0.69	N/A
9404 Cabot Property	\$0.69	N/A
9455 Cabot Property	\$0.69	N/A
9755 Distribution Property	\$0.66	N/A
9855 Distribution Property	\$0.71	N/A
14955 E. Salt Lake Property	y N/A	\$0.62

[&]quot;Mortgaged Property" means the Property, the Improvements and all other property of Borrower subject to a Lien or security interest granted by any of the Loan Documents.

"NOI" means, as of any date of determination, means the aggregate income, revenues, reimbursements and receipts of any kind whatsoever of the Remaining Property generated from executed approved (or deemed approved). Leases in-place during the trailing twelve-month period prior to the calculation date (excluding income from month-to-month Leases, Leases expiring within three months of the calculation date, Leases that are in monetary default, Leases with tenants that have vacated the applicable Property with respect to such Lease, and Leases with tenants that are in bankruptcy, provided, however, that, so long as any such income from a tenant in bankruptcy is not otherwise excluded as provided in this definition, such income shall be included if the applicable Lease has been assumed by such tenant pursuant to timely and appropriate proceedings approved by the applicable bankruptcy court), less, without duplication, (i) the aggregate expenses incurred during such twelve-month period in connection with the operation of the Remaining Property, but expressly excluding capital expenses, other non-recurring extraordinary expenses, depreciation and amortization, income taxes, debt service on the Loan, amounts funded into reserves and escrows maintained with Administrative Agent pursuant to the Loan Documents, any and all expenses (including, without limitation, legal, accounting and other professional fees, expenses and disbursements) incurred in connection with the making or administration of the Loan or the sale, exchange, transfer, financing or refinancing of any Property or in connection with the recovery of insurance proceeds or condemnation awards relating to any Property.

[&]quot;Net Casualty Proceeds" shall have the meaning set forth in Section 7.01(g)(ii) hereof.

[&]quot;Net Condemnation Proceeds" shall have the meaning set forth in Section 7.02 hereof.

"Non-Defaulting Lender" means any Lender, as determined by Administrative Agent, that is not a Defaulting Lender.

"Notes" means the Promissory Notes executed by Borrower in favor of each of the Lenders, substantially in the form of Exhibit B hereto, as amended from time to time.

"Obligations" means (i) all unpaid principal of and accrued and unpaid interest on the Loans, all accrued and unpaid fees and all expenses, reimbursements, indemnities and other indebtedness, liabilities, or obligations of Borrower to the Lenders or to any Lender, Administrative Agent or any indemnified party arising under the Loan Documents, and (ii) all Swap Obligations under Swap Agreements with one or more Lenders or their respective Affiliates, provided that at or prior to the time that any transaction relating to such Swap Obligation is executed, the Lender counterparty thereto (other than JPMorgan or its Affiliates) shall have delivered written notice to Administrative Agent that such a transaction has been or will be entered into and, subject to Administrative Agent's prior approval in its reasonable discretion, such Swap Obligation shall constitute an Obligation entitled to the benefits of the liens and guaranties provided by the Loan Documents.

"OFAC" means Office of Foreign Assets Control of the United States Department of the Treasury.

"Other Connection Taxes" means, with respect to any Recipient, Taxes imposed as a result of a present or former connection between such Recipient and the jurisdiction imposing such Tax (other than connections arising from such Recipient having executed, delivered, become a party to, performed its obligations under, received payments under, received or perfected a security interest under, engaged in any other transaction pursuant to or enforced any Loan Document, or sold or assigned an interest in any Loan or Loan Document).

"Other Taxes" means all present or future stamp, court or documentary, intangible, recording, filing or similar Taxes that arise from any payment made under, from the execution, delivery, performance, enforcement or registration of, from the receipt or perfection of a security interest under, or otherwise with respect to, any Loan Document, except any such Taxes that are Other Connection Taxes imposed with respect to an assignment (other than an assignment made pursuant to Section 3.13).

"Participant" has the meaning set forth in Section 11.04(c)(i) hereof.

"Participant Register" has the meaning assigned to such term in Section 11.04(c)(i).

"Permitted Encumbrances" means (a) Liens and security interests granted pursuant to the Loan Documents, (b) the items set forth on Schedule B of the Title Policy, (c) customary easements entered into by Borrower in connection with the development and operation of the Mortgaged Property which Administrative Agent has determined would have no material adverse effect on the use or value of the Mortgaged Property, (d) documents required to be recorded by applicable law which have no material adverse effect on the use or value of the Mortgaged Property, (e) Approved Leases, (f) subordination, non-disturbance and attornment agreements executed by Administrative Agent with respect to Approved Leases, and (g)

memoranda of Approved Leases.

- "Permitted Indebtedness" means (a) the Obligations, (b) unsecured letters of credit or guarantees required by Governmental Authorities in connection with the construction of the Improvements, (c) trade debt incurred in the ordinary course of operation of the Mortgaged Property in such amounts as are normal and reasonable under the circumstances, provided that such debt is not evidenced by a note and is paid when due and provided in any event that the outstanding principal balance of such debt shall not exceed at any one time five percent (5%) of the outstanding Obligations, (d) equipment leases entered into in the ordinary course of the operation of the Mortgaged Property and (e) Approved Leases.
 - "Person" means any natural person, corporation, limited liability company, trust, joint venture, association, company, partnership, Governmental Authority or other entity.
 - "Physical Conditions Report" means the physical condition reports described on Exhibit I attached hereto.
 - "Plan Assets" means the assets of an employee benefit plan within the meaning of 29 C.F.R. 2510.3-101.
- "Prime Rate" means the rate of interest per annum publicly announced from time to time by JPMorgan, as its prime rate; each change in the Prime Rate shall be effective from and including the date such change is publicly announced as being effective. The Prime Rate is a reference rate and is not necessarily the lowest rate.
- "Project Documents" means any agreement relating to the ownership, financing, or operation of the Improvements to which Borrower is a party or beneficiary, whether now existing or hereafter arising; provided, however, that Project Documents shall not include the Loan Documents, any Swap Agreement, or the Leases.
- "Property" means, collectively, the 3880 W. Valley Property, the 9340 Cabot Property, the 9404 Cabot Property, the 9455 Cabot Property, the 5235 E. Hunter Property, the 9755 Distribution Property, and the 14955 E. Salt Lake Property; provided, however, that, for the avoidance of doubt, each of the foregoing shall cease to be a "Property" upon the reconveyance thereof pursuant to Section 3.15 below
- "Qualified Financial Institution" means a financial institution with a long term corporate debt rating of at least "A" from Standard and Poor's Rating Group or a comparable rating by a rating agency acceptable to Administrative Agent.
 - "Recipient" means (a) the Administrative Agent and (b) any Lender, as applicable.
 - "Register" has the meaning set forth in Section 11.04(b)(iv) hereof.
 - "Related Parties" means, with respect to any specified Person, such Person's Affiliates and the respective directors, officers, employees, agents and advisors of such Person and such Person's

Affiliates.

16

- "Remaining Property" means, after giving effect to the reconveyance of any Property pursuant to Section 3.15 below, the Property that then remains subject to the lien of a Deed of Trust
- "Report" means the environmental reports and assessments described in $\underline{\textit{Exhibit } H}$ attached hereto.
- "Required Lenders" means Lenders (other than Defaulting Lenders) in the aggregate having at least 66 2/3% of the Aggregate Commitment or, if the Aggregate Commitment has been terminated, Lenders in the aggregate holding at least 66 2/3% of the aggregate unpaid principal amount of the outstanding Advances.
- "Requirement of Law" means, as to any Person, any law, treaty, rule or regulation or determination of an arbitrator or a court or other Governmental Authority, in each case applicable to or binding upon such Person or any of its property or to which such Person or any of its property is subject.
- "Sanctions Laws and Regulations" means any sanctions, prohibitions or requirements imposed by any executive order (an "Executive Order") or by any sanctions program administered by the U.S. Department of the Treasury Office of Foreign Assets Control ("OFAC").
- "Statutory Reserve Rate" means a fraction (expressed as a decimal), the numerator of which is the number one and the denominator of which is the number one minus the aggregate of the maximum reserve percentages (including any marginal, special, emergency or supplemental reserves) expressed as a decimal established by the Board to which the Administrative Agent is subject with respect to the Adjusted LIBO Rate for eurocurrency funding (currently referred to as "Eurocurrency Liabilities" in Regulation D of the Board). Such reserve percentages shall include those imposed pursuant to such Regulation D. Eurodollar Loans shall be deemed to constitute eurocurrency fundings and to be subject to such reserve requirements without benefit of or credit for proration, exemptions or offsets that may be available from time to time to any Lender under such Regulation D or any comparable regulation. The Statutory Reserve Rate shall be adjusted automatically on and as of the effective date of any change in any reserve percentage.
 - "Survey" has the meaning set forth in Section 2.02(b)(v) hereof.
- "Swap Agreement" means any agreement with respect to any swap, forward, spot, future, credit default or derivative transaction or option or similar agreement involving, or settled by reference to, one or more rates, currencies, commodities, equity or debt instruments or securities, or economic, financial or pricing indices or measures of economic, financial or pricing risk or value or any similar transaction or any combination of these transactions; provided that such Swap Agreement must specifically state that the obligations of Borrower thereunder are to be secured by the Deed of Trust.
- "Swap Obligations" of a Person means any and all obligations of such Person, whether absolute or contingent and howsoever and whensoever created, arising, evidenced or acquired (including all renewals. extensions and modifications thereof and substitutions

 $therefor), under (a) \ any \ and \ all \ Swap \ Agreements, \ and \ (b) \ any \ and \ all \ cancellations, \ buy \ backs, \ reversals, \ terminations \ or \ assignments \ of \ any \ Swap \ Agreement \ transaction.$

"Taxes" means all present or future taxes, levies, imposts, duties, deductions, withholdings (including backup withholding), assessments, fees or other charges imposed by any Governmental Authority, including any interest, additions to tax or penalties applicable thereto.

"Title Company" means Chicago Title Insurance Company.

"Title Policy" means, individually and collectively, an ALTA Lender's Policy of Title Insurance in form and substance satisfactory to Administrative Agent issued by the Title Company in the amount of the Aggregate Commitment insuring each Deed of Trust as a first priority lien on the portion of the Property and the Improvements covered thereby, containing such endorsements and with such re-insurance as Administrative Agent may request, excepting only such items as shall be acceptable to Administrative Agent.

"Transactions" means the execution, delivery and performance by Borrower of this Agreement and the other Loan Documents, the borrowing of Loans and the use of the proceeds thereof.

"Type" means, with respect to any Advance, its nature as a Floating Rate Advance or a Eurodollar Advance and with respect to any Loan, its nature as a Floating Rate Loan or Eurodollar Loan.

"Unmatured Default" means the occurrence of an event which with notice or lapse of time or both would constitute a Default.

"U.S. Person" means a "United States person" within the meaning of Section 7701(a)(30) of the Code.

"U.S. Tax Compliance Certificate" has the meaning assigned to such term in Section 3.12(f)(ii)(B)(3).

ARTICLE II CONDITIONS TO DISBURSEMENT

2.01 Generally.

Each Lender severally agrees, on the terms and conditions set forth in this Agreement, to make Loans to Borrower in a single Advance on the Closing Date in an amount equal to the aggregate amount of its Commitment. No Lender shall be responsible for the failure of any other Lender to perform its obligations to make Loans hereunder, and the Commitment of any Lender shall not be increased or decreased as a result of the failure by any other Lender to perform its obligation to make Loans hereunder.

Conditions to Closing and Advance.

Borrower agrees that, in addition to all other conditions set forth herein, the making of the Loan and the Advance is conditioned upon the fulfillment of each of the following conditions, subject, 2.02 however, to the right of Administrative Agent to waive any one or more of such conditions in whole or in part: Loan Documents and Certain Third Party Documents. Administrative Agent shall have received on the date hereof the following documents fully executed and in form (a)

and substance satisfactory to Administrative Agent: The Notes; The Deed of Trust; (ii) (iii) (iv) The Limited Guaranty;
The Environmental Indemnity Agreement; (v) (vi) Intentionally Omitted; Borrower Financing Statement; (vii) The agreement of Borrower's managing member to cause Borrower to comply with the provisions of Section 4.04 hereof; and Administrative Agent's Disbursement and Rate Management Signature Authorization and Instruction Form.

<u>Additional Closing Deliveries</u>. Administrative Agent shall have received the following on the date hereof in form and substance satisfactory to Administrative Agent: (viii) (b) (i) An opinion or opinions from counsel for Borrower and Guarantor;
Current UCC, tax and judgment searches made in such places as Administrative Agent may specify, covering Borrower and showing no filings (ii) relating to, or which could relate to, the Mortgaged Property other than those made hereunder; Evidence of the insurance required under Section 7.01 hereof: (iii)

(iv) A commitment to issue the Title Policy, together with copies of all documentation evidencing exceptions raised therein; An ALTA survey of each portion of the Property certified in a manner acceptable to Administrative Agent (the "Survey");

A certificate of a secretary or assistant secretary of Rexford Industrial Realty, Inc., a Maryland corporation, certifying as to (i) the operating (v) (vi)

agreements for Borrower, (ii) the authorizing resolutions of Borrower, and (iii) incumbency and specimen signatures of signatories for Borrower, together with (x) a copy of the Certificate of Formation for Borrower, certified by the Delaware Secretary of State as of a recent date, (y) a certificate of good standing as of a recent date for Borrower from the Delaware Secretary of State, and (z) a certificate of good standing as of a recent date for Borrower from the California Secretary of State;

(vii) A certificate of a secretary or assistant secretary of Rexford Industrial Realty, Inc., a Maryland corporation, certifying as to (i) the partnership agreement of Guarantor, (ii) the authorizing resolutions of Guarantor, and (iii) incumbency and specimen signatures of signatories for Guarantor, together with (y) a copy of the Certificate of Limited Partnership for Guarantor, certified by the State Department of Assessments and Taxation of Maryland ("SDAT") as of a recent date, and (z) a certificate of good standing as of a recent date for Guarantor from the SDAT; (viii)

A copy of the contract to purchase the Property and the related closing statement;

(ix) Evidence indicating whether the Improvements or any part thereof are or will be located within a one hundred year flood plain or other area identified by Administrative Agent has having high or moderate risk of flooding or identified as a special flood hazard area as defined by the Federal Emergency Management Agency, and, if so, a flood notification form signed by Borrower and evidence that the flood insurance required by Article VII of this Agreement is in place for the Improvements and contents, if applicable, all in form, substance and amount satisfactory to Administrative Agent and the Lenders;

> An Appraisal showing the Loan-to-Value Ratio to be no more than sixty-five percent (65%); (x)

(xi) Evidence satisfactory to Administrative Agent showing that the Aggregate Commitment does not exceed sixty-five percent (65%) of the aggregate purchase price paid by Borrower for the acquisition of the Property;

(xii)

test reports with respect to the Property made by or for Borrower; (xiii) An environmental report with respect to the Property prepared by an environmental consultant acceptable to Administrative Agent;
(xiv) A so-called probable maximum loss report ("PML Report"), which shall address (A) the scenario expected loss that is likely to be sustained by each Property in the event of an earthquake or other seismic casualty at or affecting such Property, and (B) likelihood and likely intensity of an earthquake or other seismic casualty at or affecting each Property;

A soils report with respect to each Property prepared by an engineer acceptable to Administrative Agent, and copies of all other inspection and

(xv)

The Physical Conditions Report;

Evidence indicating compliance by the Improvements with applicable zoning requirements (without requirement for a variance); Evidence that all utilities and municipal services required for the construction and operation of the Improvements are available at the Property; (xvi) (xvii) (xviii) The most recent available financial statements of Borrower, if any, and Guarantor; (xix) Copies of all Leases executed with respect to the Improvements; A fully executed subordination, non-disturbance and attornment agreement in form and substance acceptable to Administrative Agent for each (xx) Lease designated by Administrative Agents

A Certification of Non-Foreign Status; (xxi) (xxii)

A signed IRS Form W-8 or W-9 as applicable; and Such other information and documents as Administrative Agent may reasonably require.

Fees and Expenses. All fees and reimbursement of expenses due the Administrative Agent and the Lenders shall be paid prior to or out of the Advance.

Representations and Warranties. The representations and warranties made hereunder or under any of the other Loan Documents, or in any certificate or other document (c) (d)

executed by Borrower or Guarantor and delivered to Administrative Agent pursuant to or in connection with this Agreement, shall be true and correct in all material respects as of the Borrowing Date except to the extent any such representation or warranty is made as of a specified date, in which case such representation or warranty shall have been true and correct on and as of such specified date.

ARTICLE III LOAN TERMS

3.01 Loan and Advance.

(a)

(xxiii)

Generally. The Advance of the Loan is subject to satisfaction of the conditions to disbursement contained in Article II of this Agreement, as well as the terms of this

(b) Type. Subject to Section 3.08 hereof, (i) each Advance shall be comprised entirely of Floating Rate Loans or Eurodollar Loans as Borrower may request in accordance herewith and shall bear interest at the applicable Floating Rate or Eurodollar Rate, respectively. Each Lender at its option may make any Eurodollar Loan by causing any domestic or foreign branch or Affiliate of such Lender to make such

Loan; $\underline{provided}\,\text{that any exercise}$ of

Article.

- such option shall not affect the obligation of Borrower to repay such Loan in accordance with the terms of this Agreement.

 (c) Eurodollar Loan Amounts. At the commencement of each Interest Period for any Eurodollar Loan, such Loan shall be in an aggregate amount that is not less than \$1,000,000.00. Loans of more than one Type may be outstanding at the same time; provided that there shall not at any time be more than a total of three (3) Eurodollar Loans outstanding.
- (d) No Interest Period to Expire After Maturity Date. Notwithstanding any other provision of this Agreement, Borrower shall not be entitled to request, or to elect to

convert or continue, any Advance if the Interest Period requested with respect thereto would end after the Maturity Date.

3.02 Funding of Loan Advance.

- Generally. Each Lender shall make the Loan to be made by it hereunder on the proposed date thereof by wire transfer of immediately available funds by 11:00 a.m., Dallas, Texas time, to the account of Administrative Agent most recently designated by it for such purpose by notice to the Lenders. Administrative Agent will make such Loans available to Borrower by depositing the amounts so received, in like funds, with the Title Company in accordance with escrow instructions satisfactory to Administrative Agent or promptly crediting the amounts so received, in like funds, to an account of Borrower maintained with Administrative Agent and designated by Borrower in Administrative Agent's Disbursement and Rate Management Signature Authorization and Instruction Form.
- Advance Fundings. Unless Administrative Agent shall have received notice from a Lender prior to the proposed date of the Advance that such Lender will not make (b) available to Administrative Agent such Lender's share of such Advance, Administrative Agent may assume that such Lender has made such share available on such date in accordance with paragraph (a) of this Section and may, in reliance upon such assumption, make available to Borrower a corresponding amount. In such event, if a Lender has not in fact made its share of the Advance available to Administrative Agent, then the applicable Lender and Borrower severally agree to pay to Administrative Agent forthwith on demand such corresponding amount with interest thereon, for each day from and including the date such amount is made available to Borrower to but excluding the date of payment to Administrative Agent, at (i) in the case of such Lender, the greater of the Federal Funds Effective Rate and a rate determined by Administrative Agent in accordance with banking industry rules on interbank compensation or (ii) in the case of Borrower, the interest rate applicable to Floating Rate Loans. If such Lender pays such amount to Administrative Agent, then such amount shall constitute such Lender's Loan included in such Advance.
 - Interest Elections.
- (a) Generally. The Advance shall initially be a Floating Rate Advance unless Borrower has elected that the Advance be a Eurodollar Advance. Thereafter, Borrower may elect to convert such Advance to a different Type or to continue such Advance and, in the case of a Eurodollar Advance, may elect Interest Periods therefor, all as provided in this Section.

Borrower may elect different options with respect to different portions of the affected Advance, in which case each such portion shall be allocated ratably among the Lenders holding the Loans comprising such Advance, and the Loans comprising each such portion shall be considered a separate Advance.

(b) Interest Election Request. To make an election pursuant to this Section, Borrower shall notify Administrative Agent of such election by electronic communication as provided in Section 11.01 (a) in the case of a Eurodollar Advance, not later than 11:00 a.m., Dallas, Texas time, one (1) Business Day before the date of the proposed Advance, not later than 11:00 a.m., Dallas, Texas time, one (1) Business Day before the date of the proposed Advance, not later than 11:00 a.m., Dallas, Texas time, one (1) Business Day before the date of the proposed Advance, not later than 11:00 a.m., Dallas, Texas time, one (1) Business Day before the date of the proposed Advance, not later than 11:00 a.m., Dallas, Texas time, one (1) Business Day before the date of the proposed Advance, not later than 11:00 a.m., Dallas, Texas time, one (1) Business Day before the date of the proposed Advance, not later than 11:00 a.m., Dallas, Texas time, one (1) Business Day before the date of the proposed Advance, not later than 11:00 a.m., Dallas, Texas time, one (1) Business Day before the date of the proposed Advance, not later than 11:00 a.m., Dallas, Texas time, one (1) Business Day before the date of the proposed Advance, not later than 11:00 a.m., Dallas, Texas time, one (1) Business Day before the date of the proposed Advance, not later than 11:00 a.m., Dallas, Texas time, one (1) Business Day before the date of the proposed Advance, not later than 11:00 a.m., Dallas, Texas time, one (1) Business Day before the date of the proposed Advance, not later than 11:00 a.m., Dallas, Texas time, one (1) Business Day before the date of the proposed Advance, not later than 11:00 a.m., Dallas, Texas time, one (1) Business Day before the date of the proposed Advance, not later than 11:00 a.m., Dallas, Texas time, one (1) Business Day before the date of the proposed Advance, not later than 11:00 a.m., Dallas, Texas time, one (1) Business Day before the date of the proposed Advance, not later than 11:00 a.m., Dallas, Texas time, one (1) Business Day

Required Information. Each Interest Election Request shall specify the following information in compliance with Section 3.01 hereof:
the Advance to which such Interest Election Request applies and, if different options are being elected with respect to different portions thereof, the

portions thereof to be allocated to each resulting Advance (in which case the information to be specified pursuant to clauses (iii) and (iv) below shall be specified for each resulting Advance);

the effective date of the election made pursuant to such Interest Election Request, which shall be a Business Day; (ii)

(iii) the Type of Advance, i.e., a Floating Rate Advance or a Eurodollar Advance; and if the resulting Advance is a Eurodollar Advance, the Interest Period to be applicable thereto after giving effect to such election, which shall be a (iv)

If no Type of Advance is specified in the Interest Election Request or if any such Interest Election Request requests a Eurodollar Advance but does not specify an Interest Period, then Borrower shall be deemed to have requested a Floating Rate Advance.

Notice to Lenders. Promptly following receipt of an Interest Election Request, Administrative Agent shall advise each Lender of the details thereof and of such (d)

Lender's portion of each resulting Advance.

(c)

(e) Failure to Elect; Default. If Borrower fails to deliver a timely Interest Election Request with respect to a Eurodollar Advance prior to the end of the Interest Period applicable thereto, then, unless such Advance is repaid as provided herein, at the end of such Interest Period such Advance shall be converted to a Floating Rate Advance. Notwithstanding any contrary provision hereof, if a Default has occurred and is continuing and Administrative Agent, at the request of the Required Lenders, so notifies Borrower, then, so long as a Default is continuing (i) no outstanding Advance may be converted to or continued as a Eurodollar

Advance and (ii) unless repaid, each Eurodollar Advance shall be converted to a Floating Rate Advance at the end of the Interest Period applicable thereto.

3.04 Repayment of Loans; Evidence of Debt.

(a) Repayment at Maturity. Borrower hereby unconditionally promises to pay to Administrative Agent for the account of each Lender the then unpaid principal amount of each Loan and all unpaid accrued interest on the Maturity Date.

(b) Lender Accounting. Each Lender shall maintain in accordance with its usual practice an accounting of the indebtedness of Borrower to such Lender resulting from each Loan made by such Lender, including the amounts of principal and interest payable and paid to such Lender from time to time hereunder.

(c) Administrative Agent Accounting. Administrative Agent shall maintain an accounting of (i) the amount of each Loan made hereunder, the Type thereof and the Interest Period applicable thereto, (ii) the amount of any principal or interest due and payable or to become due and payable from Borrower to each Lender hereunder and (iii) the amount of any sum received by Administrative Agent hereunder for the account of the Lender's share thereof.

(d) Prima Facie Evidence. The entries made in the accounting maintained pursuant to paragraph (b) or (c) of this Section shall be prima facie evidence of the existence and amounts of the obligations recorded therein; provided that the failure of any Lender or Administrative Agent to maintain such accounting or any error therein shall not in any manner affect the obligation of Borrower to repay the Loans in accordance with the terms of this Agreement.

(e) Notes. The Loans made by each Lender shall be evidenced by the Note executed by Borrower in favor of such Lender.

Prepayment of Loans.

Borrower shall have the right at any time and from time to time to prepay any Advance in whole or in part, subject to prior notice in accordance with this Section 3.05. Borrower shall notify
Administrative Agent by electronic communication as provided in Section 11.01 of any prepayment hereunder (i) in the case of prepayment of a Eurodollar Advance, not later than 11:00 a.m., Dallas, Texas time, three (3) Business Days
before the date of prepayment, or (ii) in the case of prepayment of a Floating Rate Advance, not later than 11:00 a.m., Dallas, Texas time, one (1) Business Day before the date of prepayment. Each such notice shall be irrevocable and shall
specify the prepayment date and the principal amount of each Advance or portion thereof to be prepaid. Promptly following receipt of any such notice, Administrative Agent shall advise the Lenders of the contents thereof. Each partial
prepayment of any Advance shall be in an amount that would be permitted in the case of an Advance of the same Type as provided in Section 3.01 hereof. Prepayments shall be accompanied by accrued interest on the amount prepaid, plus
any other break funding payments or fees required by Section 3.10 or other amounts required by Section 3.07 hereof.

3.06 (a) Loan Fee. Borrower agrees to pay to Administrative Agent for the account of each Lender a loan fee in the amount separately agreed upon between Borrower and

Administrative Agent Fees Non-Refundable. All fees payable hereunder shall be paid on the dates due, in immediately available funds, to Administrative Agent for distribution, in the case of

the loan fee, to the Lenders. Fees paid shall not be refundable under any circumstances

3.07 Interest.

Floating Rate Loans. The Loans comprising each Floating Rate Advance shall bear interest at the Floating Rate. (b)

Eurodollar Loans. The Loans comprising each Eurodollar Advance shall bear interest at the Eurodollar Rate for the Interest Period in effect for such Advance.

Default Rate. Notwithstanding the foregoing, to the extent permitted under applicable law, upon the occurrence of a Default, and after maturity, the Loans shall bear

(c) interest, after as well as before judgment at a rate per annum equal to two percent (2%) plus the rate otherwise applicable to such Loans as provided in the preceding paragraphs of this Section. (d) Payment of Accrued Interest, Accrued interest on each Loan shall be payable in arrears on each Interest Payment Date for such Loan; provided that (i) interest accrued

pursuant to paragraph (c) of this Section shall be payable on demand, (ii) in the event of any repayment or prepayment or prepayment or any Loan, accrued interest on the principal amount repaid or prepaid shall be payable on the date of such repayment or prepayment and (iii) in the event of any conversion of any Eurodollar Loan prior to the end of the current Interest Period therefor, accrued interest on such Loan shall be payable on the effective date of such conversion.

(e) Computation of Interest. All interest hereunder shall be computed on the basis of a year of 360 days, and shall be payable for the actual number of days elapsed

(including the first day but excluding the last day). The applicable Alternate Base Rate, Adjusted LIBO Rate or LIBO Rate shall be determined by Administrative Agent, and such determination shall be conclusive absent manifest error.

3.08

Alternate Rate of Interest.

If prior to the commencement of any Interest Period for a Eurodollar Advance, (a) Administrative Agent determines (which determination shall be conclusive absent manifest error) that adequate and reasonable means do not exist for ascertaining the Adjusted LIBO Rate or the LIBO Rate, as applicable, for such Interest Period, or (b) Administrative Agent is advised by the Required Lenders that the Adjusted LIBO Rate or the LIBO Rate, as applicable, for such Interest Period will not adequately and fairly reflect the cost to such Lenders (or Lender) of making or maintaining their Loans (or its Loan) included in such Advance for such Interest Period, then Administrative Agent shall give notice thereof to Borrower and the Lenders by electronic communication as provided in Section 11.01 as promptly as practicable thereafter and,

until Administrative Agent notifies Borrower and the Lenders that the circumstances giving rise to such notice no longer exist, (i) any Interest Election Request that requests the conversion of any Advance to, or continuation of any Advance as, a Eurodollar Advance shall be ineffective, and (ii) if such determination is made prior to the Advance on the Closing Date, such Advance shall be made as a Floating Rate Advance.

3.09 Increased Costs.

- (a) Increased Costs of Making or Maintaining Loans. If any Change in Law shall (i) impose, modify or deem applicable any reserve, special deposit or similar requirement (including any compulsory loan requirement, insurance charge or other assessment) against assets of, deposits with or for the account of, or credit extended by, any Lender (except any such reserve requirement reflected in the Adjusted LIBO Rate), (ii) impose on any Lender or the London interbank market any other condition, cost or expense (other than Taxes) affecting this Agreement or Loans made by such Lender or (iii) subject any Recipient to any Taxes (other than (A) Indemnified Taxes, (B) Taxes described in clauses (b) through (d) of the definition of Excluded Taxes and (C) Connection Income Taxes) on its loans, loan principal, letters of credit, commitments, or other obligations, or its deposits, reserves, other liabilities or capital attributable thereto, and the result of any of the foregoing shall be to increase the cost to such Lender or such other Recipient of making or maintaining any Loan (or of maintaining its obligation to make any such Loan) or to reduce the amount of any sum received or receivable by such Lender or such other Recipient fereunder (whether of principal, interest or otherwise), then Borrower will pay to such Lender or such other Recipient such additional amount or amounts as will compensate such Lender or such other Recipient for such additional costs incurred or reduction suffered.
- (b) Capital Adequacy. If any Lender determines that any Change in Law regarding capital or liquidity requirements has or would have the effect of reducing the rate of return on such Lender's capital or on the capital of such Lender's holding company, if any, as a consequence of this Agreement or the Loans made by such Lender to a level below that which such Lender or such Lender's holding company could have achieved but for such Change in Law (taking into consideration such Lender's policies and the policies of such Lender's holding company with respect to capital adequacy and liquidity), then from time to time Borrower will pay to such Lender such additional amount or amounts as will compensate such Lender or such Lender's holding company for any such reduction suffered.
- (c) Certificate of Amounts Due. A certificate of a Lender setting forth the amount or amounts necessary to compensate such Lender or its holding company, as the case may be, as specified in paragraph (a) or (b) of this Section shall be delivered to Borrower and shall be conclusive absent manifest error. Borrower shall pay such Lender the amount shown as due on any such certificate within ten (10) days after receipt thereof. Notwithstanding anything in this Section 3.09, from and after the date such certificate is delivered to Borrower shall have the right (but not the obligation) to voluntarily prepay the outstanding principal amount of the Loan, in whole, without prepayment of any charge, premium or fee (other than amounts due, if any, pursuant to Section 3.10 below in connection with the prepayment of any Eurodollar Loans).

(d) <u>Delay in Demand For Compensation</u>. Failure or delay on the part of any Lender to demand compensation pursuant to this Section shall not constitute a waiver of such Lender's right to demand such compensation; <u>provided</u> that Borrower shall not be required to compensate a Lender pursuant to this Section for any increased costs or reductions incurred more than 270 days prior to the date that such Lender notifies Borrower of the Change in Law giving rise to such increased costs or reductions and of such Lender's intention to claim compensation therefor; <u>provided further</u> that, if the Change in Law giving rise to such increased costs

3.10 Break Funding Payments.

or reductions is retroactive, then the 270-day period referred to above shall be extended to include the period of retroactive effect thereof.

In the event of (a) the payment of any Eurodollar Loan other than on the last day of an Interest Period applicable thereto (including as a result of a Default), (b) the conversion of any Eurodollar Loan other than on the last day of the Interest Period applicable thereto, (c) the failure to borrow, convert, continue or prepay any Eurodollar Loan on the date specified in any notice delivered pursuant hereto, or (d) the assignment of any Eurodollar Loan other than on the last day of the Interest Period applicable thereto as a result of a request by Borrower pursuant to Section 3.13 hereof, then, in any such event, Borrower shall pay to Administrative Agent an administrative fee of \$250.00 and compensate each Lender for the loss, cost and expense attributable to such event. Such loss, cost or expense to any Lender shall be deemed to include an amount determined by such Lender to be the excess, if any, of (i) the amount of interest which would have accrued on the principal amount of such Eurodollar Loan had such event not occurred, at the Adjusted LIBO Rate that would have been applicable to such Eurodollar Loan, for the period from the date of such event to the last day of the then current Interest Period therefor (or, in the case of a failure to borrow, convert or continue, for the period that would have been the Interest Period for such Eurodollar Loan), over (ii) the amount of interest which would accrue on such principal amount for such period, for the beriod from other banks in the eurodollar market. A certificate of any Lender setting forth in reasonable detail any amount or amounts that such Lender is entitled to receive pursuant to this Section shall be delivered to Borrower and shall be conclusive absent manifest error. Borrower shall pay such Lender the amount shown as due on any such certificate within ten (10) days after receipt thereof.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

By its signature below, Borrower waives any right under California Civil Code Section 2954.10 or otherwise to prepay any Eurodollar Loan, in whole or in part, without payment of the amounts described above. Borrower acknowledges that prepayment of any Eurodollar Loan may result in Lenders' incurring additional losses, costs, expenses and liabilities, including lost revenues and lost profits. Borrower therefore agrees to pay described above. Borrower acknowledges that prepayment of any Eurodollar Loan may result in Leithers incluming admittorian issess, costs, expenses and nationates, including tost revenues and rost profits. Borrower interest in the Mortgaged Property giving Administrative Agent the right to accelerate the maturity of the Loans as provided in the Deed of Trust. Borrower agrees that Lenders' willingness to offer Eurodollar Loans to Borrower is sufficient and independent consideration, given individual weight by Lenders, for this waiver.

REXFORD INDUSTRIAL – SDLAOC, LLC,

a Delaware limited liability company

By:Rexford Industrial Realty, L.P., a Maryland limited partnership, its Managing Member

> By:Rexford Industrial Realty, Inc., a Maryland corporation, its General Partner

> > By: /s/ Howard

Name: <u>Howard Schwimmer</u> Title: Co-Chief Executive Officer

3.11 Taxes.

(a) Payments Free of Taxes. Any and all payments by or on account of any obligation of Borrower under any Loan Document shall be made without deduction or withholding for any Taxes, except as required by applicable law. If any applicable law (as determined in the good faith discretion of an applicable withholding agent) requires the deduction or withholding of any Tax from any such payment by a withholding agent, then the applicable withholding agent shall be entitled to make such deduction or withholding and shall timely pay the full amount deducted or withheld to the relevant Governmental Authority in accordance with applicable law and, if such Tax is an Indemnified Tax, then the sum payable by Borrower shall be increased as necessary so that after such deduction or withholding has been made (including such deductions and withholdings applicable to additional sums payable under this Section 3.11) the applicable Recipient receives an amount equal to the sum it would have received had no such deduction or withholding been made.

(b) Payment of Other Taxes by Borrower. Borrower shall timely pay to the relevant Governmental Authority in accordance with applicable law, or at the option of the

Administrative Agent timely reimburse it for, Other Taxes. Evidence of Payments. As soon as practicable after any payment of Taxes by Borrower to a Governmental Authority pursuant to this Section 3.11, Borrower shall deliver to the Administrative Agent the original or a certified copy of a receipt issued by such Governmental Authority evidencing such payment, a copy of the return reporting such payment or other evidence of such payment reasonably

satisfactory to the Administrative Agent. (d) Indemnification by Borrower. Borrower shall indemnify each Recipient, within 10 days after demand therefor, for the full amount of any Indemnified Taxes (including Indemnified Taxes imposed or asserted on or attributable to amounts payable under this Section) payable or paid by such Recipient or required to be withheld or deducted from a payment to such Recipient and any reasonable expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to Borrower by a Lender (with a copy to the Administrative Agent), or by the Administrative Agent on its own behalf or on behalf or a Lender, shall be conclusive absent manifest error.

Indemnification by the Lenders. Each Lender shall severally indemnify the Administrative Agent, within 10 days after demand therefor, for (i) any Indemnified Taxes attributable to such Lender (but only to the extent that Borrower has not already indemnified the Administrative Agent for such Indemnified Taxes and without limiting the obligation of Borrower to do so), (ii) any Taxes attributable to such Lender's failure to comply with the provisions of Section 11.04(c)(i) relating to the maintenance of a Participant Register and (iii) any Excluded Taxes attributable to such Lender, in each case, that are payable or paid by the Administrative Agent in connection with any Loan Document, and any reasonable expenses arising therefrom or with respect thereto, whether or not such Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to any Lender by the Administrative Agent shall be conclusive absent manifest error. Each Lender hereby authorizes the Administrative Agent to set off and apply any and all amounts at any time owing to such Lender under any Loan Document or otherwise payable by the Administrative Agent to the Lender from any other source against any amount due to the Administrative Agent under this paragraph (e).

(i) Any Lender that is entitled to an exemption from or reduction of withholding Tax with respect to payments made under any Loan Document shall deliver to Borrower and the Administrative Agent, at the time or times reasonably requested by Borrower or the Administrative Agent, such properly completed and executed documentation reasonably requested by Borrower or the Administrative Agent as will permit such payments to be made without withholding or at a reduced rate of withholding. In addition, any Lender, if reasonably requested by Borrower or the Administrative Agent, shall deliver such other documentation prescribed by applicable

law or reasonably requested by Borrower or the Administrative Agent as will enable Borrower or the Administrative Agent to determine whether or not such Lender is subject to backup withholding or information reporting requirements. Notwithstanding anything to the contrary in the preceding two sentences, the completion, execution and submission of such documentation (other than such documentation set forth in Section 3.11(f)(ii)(A), (ii)(B) and (ii)(D) below) shall not be required if in the Lender's reasonable judgment such completion, execution or submission would subject such Lender to any material unreimbursed cost or expense or would materially prejudice the legal or commercial position of such Lender.

Without limiting the generality of the foregoing, in the event that Borrower is a U.S. Person,

(A) any Lender that is a U.S. Person shall deliver to Borrower and the Administrative Agent on or prior to the date on which such Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of Borrower or the Administrative Agent), executed originals of IRS Form W-9 certifying that such Lender is exempt from U.S. Federal backup withholding tax;

(B) any Foreign Lender shall, to the extent it is legally entitled to do so, deliver to Borrower and the Administrative Agent (in such number of copies as shall be requested by the recipient) on or prior to the date on which such Foreign Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of Borrower or the Administrative Agent), whichever of the following is applicable:

(1) in the case of a Foreign Lender claiming the benefits of an income tax treaty to which the United States is a party (x) with respect to payments of interest under any Loan Document, executed originals of IRS Form W-8BEN establishing an exemption from, or reduction of, U.S. Federal withholding Tax pursuant to the "interest" article of such tax treaty and (y) with respect to any other applicable payments under any Loan Document, IRS Form W-8BEN establishing an exemption from, or reduction of, U.S. Federal withholding Tax pursuant to the "business profits" or "other income" article of such tax treaty; (2)

executed originals of IRS Form W-8ECI;

in the case of a Foreign Lender claiming the benefits of the exemption for portfolio interest under Section 881(c) of the Code, (x) a certificate substantially in the form of Exhibit F-1 to the effect that such Foreign Lender is not a "bank" within the meaning of Section 881(c)(3)(A) of the Code, a "10 percent shareholder" of Borrower within the meaning of Section 881(c)(3)(B) of the Code, or a "controlled foreign corporation" described in Section 881(c)(3)(C) of the Code (a "U.S. Tax")

Compliance Certificate") and (y) executed originals of IRS Form W-8BEN; or

(4) to the extent a Foreign Lender is not the beneficial owner, executed originals of IRS Form W-8IMY, accompanied by IRS Form W-8ECI, IRS Form W-8BEN, a U.S. Tax Compliance Certificate substantially in the form of Exhibit F-2 or Exhibit F-3, IRS Form W-9, and/or other certification documents from each beneficial owner, as applicable; provided that if the Foreign Lender is a partnership and one or more direct or indirect partners of such Foreign Lender are claiming the portfolio interest exemption, such Foreign Lender may provide a U.S. Tax Compliance Certificate substantially in the form of Exhibit F-4 on behalf of each such direct and indirect partner;

any Foreign Lender shall, to the extent it is legally entitled to do so, deliver to Borrower and the Administrative Agent (in such number of copies as shall be requested by the recipient) on or prior to the date on which such Foreign Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of Borrower or the Administrative Agent), executed originals of any other form prescribed by applicable law as a basis for claiming exemption from or a reduction in U.S. Federal withholding Tax, duly completed, together with such supplementary documentation as may be prescribed by applicable law to permit Borrower or the Administrative Agent to determine the withholding or deduction required to be made; and

(D) if a payment made to a Lender under any Loan Document would be subject to U.S. Federal withholding Tax imposed by FATCA if such Lender were to fail to comply with the applicable reporting requirements of FATCA (including those contained in Section 1471(b) or 1472(b) of the Code, as applicable), such Lender shall deliver to Borrower and the Administrative Agent at the time or times prescribed by law and at such time or times reasonably requested by Borrower or the Administrative Agent as may be necessary for Borrower and the Administrative Agent to comply with their obligations under FATCA and to determine that such Lender has complied with such Lender's obligations under FATCA or to determine the amount to deduct and withhold from such payment. Solely for purposes of this clause (D), "FATCA" shall include any amendments made to FATCA after the date of this Agreement.

Each Lender agrees that if any form or certification it previously delivered expires or becomes obsolete or inaccurate in any respect, it shall update such form or certification or promptly notify Borrower and the Administrative Agent in writing of its legal inability to do so.

(g) Treatment of Certain Refunds. If any party determines, in its sole discretion exercised in good faith, that it has received a refund of any Taxes as to which it has been indemnified pursuant to this Section 3.11 (including by the payment of additional amounts pursuant to this Section 3.11), it shall pay to the indemnifying party an amount equal to such refund (but only to the extent of indemnity payments made under this Section 3.11 with respect to the Taxes giving rise to such refund), net of all out-of-pocket expenses (including Taxes) of such indemnified party and without interest (other than any interest paid by the relevant Governmental Authority with respect to such refund). Such indemnifying party, upon the request of such indemnified party, shall repay to such indemnified party the amount paid over pursuant to this paragraph (g) (plus any penalties, interest or other charges imposed by the relevant Governmental Authority) in the event that such indemnified party is required to repay such refund to such Governmental Authority. Notwithstanding anything to the contrary in this paragraph (g), in no event will the indemnified party be required to pay any amount to an indemnifying party pursuant to this paragraph (g) the payment of which would place the indemnified party in a less favorable net after-Tax position than the indemnified party would have been in if the Tax subject to indemnification and giving rise to such refund had not been deducted, withheld or otherwise imposed and the indemnification payments or additional amounts with respect to such Tax had never been paid. This paragraph shall not be construed to require any indemnified party to make available its Tax returns (or any other information relating to its Taxes that it deems confidential) to the indemnifying party

(h) Survival; Defined Terms. Each party's obligations under this Section 3.11 shall survive the resignation or replacement of the Administrative Agent or any assignment of rights by, or the replacement of, a Lender, the termination of the Commitments and the repayment, satisfaction or discharge of all obligations under any Loan Document. For purposes of this Section 3.11, the term "applicable law" includes

Payments Generally; Pro Rata Treatment; Sharing of Set-offs.

Payments Generally. Borrower shall make each payment required to be made by it hereunder (whether of principal, interest or fees, or of amounts payable under Section 3.09, 3.10 or 3.11, or otherwise) prior to 11:00 a.m., Dallas, Texas time, on the date when due, in immediately available funds, without set-off or counterclaim. Any amounts received after such time on any date may, in the discretion of Administrative Agent, be deemed to have been received on the next succeeding Business Day for purposes of calculating interest thereon. All such payments shall be made to Administrative Agent at its offices at 700 N Pearl Street, 13th Floor, TX1-2625, Dallas, TX 75201, Attention: Commercial Real Estate Loan Administration, except that payments pursuant to Sections 3.09, 3.10, 3.11 and 11.03 hereof shall be made directly to the Persons entitled thereto. Administrative Agent shall distribute any such payments received by it for the account of any other Person to the appropriate recipient promptly following receipt thereof. If any payment hereunder shall be due on a day that is not a Business Day, the date for payment shall be extended to the next succeeding Business Day, and, in the case of any payment accruing interest, interest thereon shall be payable for the period of such extension. All payments hereunder shall be made in U.S. dollars.

cation of Insufficient Funds. If at any time insufficient funds are received by and available to Administrative Agent to pay fully all amounts of principal, interest

and fees then due hereunder, such funds shall be applied (i) first, towards payment of fees, indemnities and expense reimbursements then due hereunder to the parties entitled thereto; (ii) second, towards payment of interest then due hereunder, ratably among the parties entitled thereto in accordance with the amounts of interest then due to such parties, and (iii) third, towards payment of principal then due hereunder and Swap Obligations then due on a pari passu basis, ratably among the parties entitled thereto in accordance with the amounts of principal and Swap Obligations then due to such parties.

- (c) Allocation of Payments. If, except as otherwise expressly provided herein, any Lender shall, by exercising any right of set-off or counterclaim or otherwise, obtain payment in respect of any principal of or interest on any of its Loans resulting in such Lender receiving payment of a greater proportion of the aggregate amount of its Loans and accrued interest thereon than the proportion received by any other similarly situated Lender, then the Lender receiving such greater proportion shall purchase (for cash at face value) participations in the Loans of other Lenders to the extent necessary so that the benefit of all such payments shall be shared by all such Lenders ratably in accordance with the aggregate amount of principal of and accrued interest on their respective Loans; provided that (i) if any such participations are purchased and all or any portion of the payment giving rise thereto is recovered, such participations shall be rescinded and the purchase price restored to the extent of such recovery, without interest, and (ii) the provisions of this paragraph shall not be construed to apply to any payment made by Borrower pursuant to and in accordance with the express terms of this Agreement or any payment obtained by a Lender as consideration for the assignment of or sale of a participation in any of its Loans to any assignee or participant, other than to Borrower or any subsidiary or Affiliate thereof (as to which the provisions of this paragraph shall apply). Borrower consents to the foregoing and agrees, to the extent it may effectively do so under applicable law, that any Lender acquiring a participation pursuant to the foregoing arrangements may exercise against Borrower rights of set-off and counterclaim with respect to such participation as fully as if such Lender were a direct creditor of Borrower in the amount of such participation.
- (d) Advance Payments. Unless Administrative Agent shall have received notice from Borrower prior to the date on which any payment is due to Administrative Agent for the account of the Lenders hereunder that Borrower will not make such payment, Administrative Agent may assume that Borrower has made such payment on such date in accordance herewith and may, in reliance upon such assumption, distribute to the Lenders the amount due. In such event, if Borrower has not in fact made such payment, then each of the Lenders severally agrees to repay to Administrative Agent forthwith on demand the amount so distributed to such Lender with interest thereon, for each day from and including the date such amount is distributed to it but excluding the date of payment to Administrative Agent, at the greater of the Federal Funds Effective Rate and a rate determined by Administrative Agent in accordance with banking industry rules on interbank compensation.

3.13

Mitigation Obligations; Replacement of Lenders.

Mitigation of Increased Costs. If any Lender requests compensation under Section 3.09 hereof, or if Borrower is required to pay any Indemnified Taxes or additional amounts to any Lender or any Governmental Authority for the account of any Lender pursuant to

Section 3.11 hereof, then such Lender shall use reasonable efforts to designate a different lending office for funding or booking its Loans hereunder or to assign its rights and obligations hereunder to another of its offices, branches or affiliates, if, in the judgment of such Lender, such designation or assignment (i) would eliminate or reduce amounts payable pursuant to Section 3.09 or 3.11 hereof, as the case may be, in the future and (ii) would not subject such Lender to any unreimbursed cost or expense and would not otherwise be disadvantageous to such Lender. Borrower hereby agrees to pay all reasonable costs and expenses incurred by any Lender in connection with any such designation or assignment.

(b) Replacement of Lenders. If any Lender requests compensation under Section 3.09 hereof, or if Borrower is required to pay any Indemnified Taxes or additional amounts to any Lender or any Governmental Authority for the account of any Lender pursuant to Section 3.11 hereof, or if any Lender defaults in its obligation to fund Loans hereunder, then Borrower may, at its sole expense and effort, upon notice to such Lender and Administrative Agent, require such Lender to assign and delegate, without recourse (in accordance with and subject to the restrictions contained in Section 11.04 hereof), all its interests, rights (other than its existing rights to payments pursuant to Section 3.09 or 3.11) and obligations under this Agreement to an assignee that shall assume such obligations (which assignee may be another Lender, if a Lender accepts such assignment); provided that (i) Borrower shall have received the prior written consent of Administrative Agent, (ii) such Lender shall have received payment of an amount equal to the outstanding principal of its Loans, accrued interest thereon, accrued fees and all other amounts payable to it hereunder, from the assignee (to the extent of such outstanding principal and accrued interest and fees) or Borrower (in the case of all other amounts); provided however, that in the case of Borrower's replacement of a Defaulting Lender for failure to fund Loans hereunder, the assignee or Borrower, as the case may be, shall holdback from such amounts payable to such Lender and pay directly to Administrative Agent, any payments due to Administrative Agent or the Non-Defaulting Lenders by Defaulting Lender under this Agreement, and (iii) in the case of any such assignment resulting from a claim for compensation under Section 3.09 hereof or payments required to be made pursuant to Section 3.11 hereof, such assignment will result in a reduction in such compensation or payments. A Lender shall not be required to make any such assignment and delegation if, prior thereto, as a result of a waive

3.14 Extension Option.

(a) Extension of Maturity. At the written request of Borrower made at least sixty (60) but not more than one hundred twenty (120) days prior to the Initial Maturity Date, the Maturity Date shall be extended to the second anniversary of the Initial Maturity Date (the "Extended Maturity Date") provided that the following conditions are satisfied:

(i) Administrative Agent shall have received evidence indicating whether the Improvements or any part thereof are or will be located within a one hundred year flood plain or other area identified by Administrative Agent has having high or moderate risk of flooding or identified as a special flood hazard area as defined by the Federal Emergency Management Agency, and, if so, a flood notification form signed by Borrower and evidence that the flood insurance required by ARTICLE VII of this

Agreement is in place for the Improvements and contents, if applicable, all in form, substance and amount satisfactory to Administrative Agent and the Lenders.

(ii) Borrower has demonstrated to the satisfaction of Administrative Agent that, as of the Initial Maturity Date, the Debt Service Coverage Ratio of the then Remaining Property, on an aggregate bases, is at least 1.25:1.00; (iii) Borrower has demonstrated to the satisfaction of Administrative Agent that, as of the Initial Maturity Date, the Debt Yield Ratio of the then Remaining Property, on an aggregate bases, is at least 9.25%; (iv) Borrower has delivered to Administrative Agent evidence acceptable to Administrative Agent (which evidence shall be a new Appraisal of each then Remaining Property timely obtained by Administrative Agent) that, as of the Initial Maturity Date, the Loan-to-Value Ratio of the then Remaining does not exceed sixty-five percent (65%); (v) On or before the Initial Maturity Date, Administrative Agent shall have received, on behalf of the Lenders, an extension fee in an amount equal to two-tenths of one percent (0.20%) of the Aggregate Commitment; No Default or Unmatured Default shall have occurred and be continuing on the Initial Maturity Date; If requested by Administrative Agent, Administrative Agent shall have received a CLTA 110.5E (or equivalent) endorsement to each Title Policy. (vii) All representations and warranties made hereunder or under any of the other Loan Documents shall be true and correct in all material respects as (viii) of the Initial Maturity Date, except to the extent such representation and warranty is made as of a specified date, in which case such representation and warranty shall have been true and correct as of such specified date; and (ix) Administrative Agent has received satisfactory documentation evidencing the extension executed by Borrower and consented to by the Guarantor and Borrower shall have paid all of Administrative Agent's costs and expenses in connection with such extension in accordance with Section 11.03. (b) <u>Voluntary Prepayment.</u> In the event the conditions to extension set forth in clauses (ii), (iii) or (iv) of Section 3.14(a) hereof, as applicable, are not satisfied, Borrower may satisfy such requirement(s) prior to the Initial Maturity Date by making a voluntary paydown of the Loan, subject, in each case, to the satisfaction of any conditions to prepayment, including the payment of any prepayment fee or Reconveyance of Property.

Reconveyance of Property. From time to time, upon the request of Borrower, Administrative Agent shall reconvey and release all, but not less than all, of a

premium

Property from the Lien of the applicable Deed of Trust (the "Released Property") upon Borrower's satisfaction of all of the following conditions precedent:

(i) Borrower shall have submitted to Administrative Agent a written request that Administrative Agent reconvey and release the Released Property from the Lien of the applicable Deed of Trust;

No Default or Unmatured Default (as therein defined) has occurred and is continuing under any Loan Document; Administrative Agent has been paid, in immediately available funds, the reasonable costs of preparing and delivering the reconveyance and of any (ii) (iii)

title insurance endorsements reasonably required by Administrative Agent; nistrative Agent has been paid, in immediately available funds, the Minimum Release Price applicable to the Released Property; (iv)

(v) The release of the Released Property will not result in the loss by any Remaining Property of access to a public street, the use of necessary easements or utilities, or any parking availability which is necessary for the use and operation of any such Remaining Property for its intended purpose;

If applicable, the Released Property shall be subject to such reciprocal easement agreements and other agreements as Administrative Agent may reasonably require to ensure that all Remaining Property enjoys adequate ingress and egress, access to a public street, utility access, parking, and such other rights and benefits as Administrative Agent may reasonably

(vii) Borrower's rights under any reciprocal easement agreement or other agreement required pursuant to Section 3.15(a)(vi) above shall, in each case, be subject to the lien of each other Deed of Trust covering the Remaining Property;

(viii) The Released Property shall constitute a separate legal lot in accordance with applicable California law, including without limitation, the California Subdivision Map Act;

(ix) After giving effect to the requested release, the 9755 Distribution Property shall not represent more than twenty-five percent (25%) of the aggregate "as-is" value of the Remaining Property (as determined by Administrative Agent and if required by Administrative Agent, based on a new Appraisal timely obtained by Administrative Agent); and (x) All escrow, closing and recording costs have been paid at no expense to Administrative Agent.

The foregoing conditions precedent contained in this Section 3.15(a) are solely for the benefit of Administrative Agent and Lenders and, subject to the terms and

conditions of this Agreement, may be waived in writing by Administrative Agent and in no other manner.

Swap Obligations. Notwithstanding the foregoing provisions of this Section 3.15, if after giving effect to the release and reconveyance of the Lien of the applicable (b) Deed of Trust encumbering a Property, no other Property will be subject to the Lien of any Deed of Trust, and if the Deed of Trust to be released secures any Swap Obligations of Borrower or any other party under a Swap Agreement with one or more Lenders or their respective Affiliates, Administrative Agent shall not be required to reconvey the lien of such Deed of Trust unless Borrower shall provide to Administrative Agent or the applicable Lender or Affiliate of Lender such substitute security for such Swap Obligations as Administrative Agent or the applicable Lender or Affiliate of Lender may require in its sole discretion.

ARTICLE IV GENERAL COVENANTS

4.01 Inspection.

(a) Inspections. Administrative Agent, through its officers, agents and employees, shall have the right, subject to the rights of tenants, at all reasonable times, on reasonable prior notice and at Administrative Agent's sole risk (i) to enter upon the Mortgaged Property and inspect the Mortgaged Property, and (ii) to examine the books, records, accounting data and other documents pertaining to the Mortgaged Property. Borrower will cooperate with Administrative Agent and its representatives and consultants.

(b) Exculpation. It is expressly understood and agreed that Administrative Agent is under no duty to inspect the Mortgaged Property, and that any such inspection by or on behalf of Administrative Agent is for the sole purpose of protecting the interests of Administrative Agent and the Lenders with respect to the Mortgaged Property. Failure to inspect the Mortgaged Property or any part thereof shall not constitute a waiver of any of Administrative Agent's rights hereunder. Inspection not followed by notice of Default shall not constitute a waiver of any Default then existing. It is further understood and agreed that any consents or approvals of Lender hereunder are for the sole purpose of protecting the interests of Lender under the Loan Documents and Borrower shall have no right to rely on such approvals for Borrower's purposes.

(a) Liens. Borrower shall pay, satisfy and obtain the release of all other claims and Liens affecting or purporting to affect the title to, or which may be or appear to be Liens on, the Mortgaged Property or any part thereof (other than the Permitted Encumbrances), and all costs, charges, interest and penalties on account thereof, including without limitation the claims of all Persons supplying labor or materials to the Mortgaged Property, and to give Administrative Agent, upon demand, evidence satisfactory to Administrative Agent of the payment, satisfaction or release thereof. Notwithstanding the foregoing, nothing herein contained shall require Borrower to pay any claims or Liens which Borrower in good faith

disputes and which Borrower, at its own expense, is currently and diligently contesting, provided that Borrower complies with the provisions of Section 4.02(c) hereof.

(b) Taxes. Borrower agrees to pay or cause to be paid, prior to the date they would become delinquent if not paid, any and all taxes, assessments and governmental charges whatsoever levied upon or assessed or charged against the Mortgaged Property, including all water and sewer taxes, assessments and other charges, fines, impositions and rents, if any. If requested by Administrative Agent, Borrower shall give to Administrative Agent a receipt or receipts, or certified copies thereof, evidencing every such payment by Borrower, not later than forty-five (45) days after such payment is made. Notwithstanding the foregoing, nothing herein contained shall require Borrower to pay any taxes, assessments or governmental charges which Borrower in good faith disputes and which Borrower, at its own expense, is currently and diligently contesting, provided that applicable law

allows non-payment thereof during the pendency of such contest, and provided further that Borrower complies with the provisions of Section 4.02(c) hereof.

(c) Contest. Borrower shall not be required to pay any taxes, assessments, governmental charges, or claims, or Liens being contested in accordance with the provisions of Section 4.02(a) or (b) hereof, as the case may be, so long as (i) Borrower diligently prosecutes such dispute or contest to a prompt determination in a manner not prejudicial to Administrative Agent or the Lenders and promptly pays all amounts ultimately determined to be owing, and (ii) Borrower provides security for the payment of such tax, assessment or governmental charge, or claim or Lien (together with interest and penalties relating thereto) in an amount and in form and substance satisfactory to Administrative Agent. If Borrower shall fail to pay any such amounts ultimately determined to be owing or to proceed diligently to prosecute such dispute or contest as provided herein, then, upon the expiration of ten (10) days after written notice to Borrower by Administrative Agent of Administrative Agent of Administrative Agent, and addition to any other right or remedy of Administrative Agent, Administrative Agent may, but shall not be obligated to, discharge the same, and the cost thereof shall be reimbursed by Borrower to Administrative Agent. The payment by Administrative Agent of any delinquent tax, assessment or governmental charge, or any claim or Lien which Administrative Agent in good faith believes might be prior hereto, shall be conclusive between the parties as to the legality and amount so paid, and Administrative Agent shall be subrogated to all rights, equities and liens discharged by any such expenditure to the fullest extent permitted by law.

4.03 Leases.

Affirmative Covenants. Borrower shall (i) duly and punctually observe, perform and discharge in all material respects the obligations, terms, covenants, conditions and (a) warranties of Borrower as landlord under the Leases, (ii) give prompt notice to Administrative Agent of any material failure on the part of Borrower to observe, perform and discharge the same or of any written claim made by any Lessee under a Major Lease of any such material failure by Borrower, (iii) enforce the performance of each and every material obligation, term, covenant, condition and agreement in the Leases to be performed by any lessee or any guarantor (including, without limitation, the tenant's obligation to pay rent as and when due), (iv) appear in and defend any action or proceeding arising under, occurring out of or in any manner connected with the Leases or the obligations, duties or

lessee thereunder, do so in the name and on behalf of Administrative Agent upon request by Administrative Agent, but at the expense of Borrower, and pay all costs and expenses of Administrative Agent, including reasonable attorneys' fees and disbursements, in any action or proceeding in which Administrative Agent, (v) at the request of Administrative Agent, use commercially reasonable efforts to cause each lessee under a Lease to execute and deliver to Administrative Agent a subordination, non-disturbance and attornment agreement in form and substance reasonably acceptable to Administrative Agent promptly upon the execution of such Lease, (vi) at the request of Administrative Agent, in confirmation of the assignment and transfer contemplated by the applicable Deed of Trust, execute and deliver to Administrative Agent assignments and transfers of all future Leases upon the same terms and conditions as contained in such Deed of Trust, and (vii) make, execute and deliver to Administrative Agent and attains and other documents and instruments which Administrative Agent may reasonably deem advisable to carry out the true purposes and intent of the assignment set forth in the applicable Deed of Trust. Administrative Agent shall cooperate in good faith with any request made in connection with a Lease approved or deemed approved by Administrative Agent assignments and end attornment agreement prepared on Administrative Agent assignments and autornment agreement with only such changes thereto as Administrative Agent may approve in its sole and absolute discretion.

(b) Negative Covenants. Unless Borrower first obtains the written consent of Administrative Agent (which consent shall not be unreasonably withheld), Borrower shall not

(b) <u>Negative Covenants</u>. Unless Borrower first obtains the written consent to Administrative Agent (which consent shall not be unreasonably withheld), Borrower shall not (c) cancel, terminate or consent to any surrender of any Major Lease unless such Major Lease is immediately replaced with a Major Lease with comparable or better terms or unless compelled to do so under court order, (ii) materially and adversely modify or alter the terms of any Major Lease unless, after giving effect to any such modification or alteration, such Major Lease satisfies the requirements for "deemed" approval as set forth in Exhibit E, (ii) waive or release any lessee or any guarantors under a Major Lease from any material obligations or conditions to be performed by such lessee or guarantors, (iv) enter into any Lease of all or any portion of any Property unless such Lease is an Approved Lease, (v) renew or extended the term of any Major Lease, except as required under the terms of a Major Lease, except as required under the terms of a Major Lease, consent to any subletting of the premises under any Major Lease, to any assignment of any such Major Lease by the lessee thereunder, or to any assignment of runther subletting of any sublease of any such Major Lease unless, after giving effect to such sublease or assignment, such Major Lease (and any and all subleases thereof) satisfies the requirements for an Approved Lease, (vii) receive or collect any rents from any lessee for a period of more than one month in advance, (viii) further pledge, transfer, mortgage or otherwise encumber or assign future payments of rents, or (ix) waive, excuse, condone, discount, set off, compromise, or in any manner release or discharge any lessee under any Lease of and from any material obligations, covenants, conditions and agreements to be kept, observed and performed by such lessee, including the obligation to pay rents thereunder, in the manner and at the time and place specified therein. Borrower may deliver to Administrative Agent's preliminary,

case, a "Preliminary Response"). Administrative Agent shall review any such request promptly and shall use commercially reasonable efforts to respond to such request within five (5) Business Days. Borrower expressly acknowledges, however, that any such Preliminary Response provided by Administrative Agent of any such terms or any such proposed Lease or amendment. Administrative Agent's failure to disapprove any matter as to which Administrative Agent s prior approval is required under this Section 4.03(b) within a seven (7) Business Day period after Administrative Agent shall have received a written request for approval specifically referencing this Section 4.03(b) and all documents and materials reasonably requested by Administrative Agent in connection with any such matter (including, without limitation, complete copies of any proposed Lease or amendment to any Lease, and any and all financial statements and materials of any proposed tenant, subtenant and any lease guarantor to the extent available to Borrower) shall be deemed to constitute Administrative Agent's approval thereof.

Operations of Borrower. 4.04 (a)

Without limitation of any other provisions of this Agreement or any other Loan Document, Borrower hereby represents, warrants, covenants and agrees that it has not engage in any business or activity other than the acquisition, development, construction, ownership, leasing, operation and maintenance of the

(i) Mortgaged Property, and activities incidental thereto;

(ii) acquire or own any material asset other than the Property, the Improvements, and such incidental personal property as may be necessary for the

construction and operation of the Improvements

merge into or consolidate with any Person or dissolve, terminate or liquidate in whole or in part, transfer or otherwise dispose of all or (iii) substantially all of its assets or change its legal structure, without in each case the prior written consent of Administrative Agent;
(iv) fail to preserve its existence as an entity duly organized, validly existing and in good standing (if applicable) under the laws of the jurisdiction of

its organization or formation, or without the prior written consent of Administrative Agent, amend, modify, terminate or fail to comply with the provisions of Borrower's organizational documents; (v) own any subsidiary or make any investment in or acquire the obligations or securities of any other Person without the prior written consent of

Administrative Agent;

commingle its assets with the assets of any of its partner(s), members, shareholders, Affiliates, or of any other Person or transfer any assets to any (vi) such Person other than distributions on account of equity interests in Borrower permitted hereunder and properly accounted for;

and shall not:

incur any Indebtedness other than Permitted Indebtedness;

(vii) (viii) allow any Person to pay its debts and liabilities or fail to pay its debts and liabilities solely from its own assets;

(ix) fail to maintain its records, books of account and bank accounts separate and apart from those of the shareholders, partners, members, principals and Affiliates of Borrower, the affiliates of a shareholder, partner or member of Borrower, and any other Person or fail to prepare and maintain its own financial statements in accordance with GAAP and susceptible to audit, or if such financial statements are consolidated, fail to cause such financial statements to contain footnotes disclosing that the Mortgaged Property is actually owned by Borrower;

(x) enter into any contract or agreement with any shareholder, partner, member, principal or Affiliate of Borrower, any Guarantor or any shareholder, partner, member, principal or Affiliate thereof, except upon terms and conditions that are intrinsically fair and substantially similar to those that would be available on an arms-length basis with third parties other than any

shareholder, partner, member, principal or Affiliate of Borrower or Guarantor, or any shareholder, partner, member, principal or Affiliate thereof;

(xi) seek dissolution or winding up, in whole or in part;

(xii) (xiii)

fail to correct any known misunderstandings regarding the separate identity of Borrower;
hold itself out to be responsible or pledge its assets or credit worthiness for the Indebtedness of another Person or allow any Person to hold itself out to be responsible or pledge its assets or credit worthiness for the Indebtedness of Borrower (except pursuant to the Loan Documents);

(xiv) make any loans or advances to any third party, including any shareholder, partner, member, principal or Affiliate of Borrower, or any shareholder,

partner, member, principal or Affiliate thereof;

(xv) fail to file its own tax returns (except to the extent that Borrower is permitted or required to file a consolidated tax return un Requirements or is a tax disregarded entity that is not required to file a tax return under applicable Legal Requirements) or to use separate contracts, purchase orders, stationery, invoices and checks; fail to file its own tax returns (except to the extent that Borrower is permitted or required to file a consolidated tax return under applicable Legal

(xvi) fail either to hold itself out to the public as a legal entity separate and distinct from any Person or to conduct its business solely in its own name in order not (i) to mislead others as to the entity with which such other party is transacting business, or (ii) to suggest that Borrower is responsible for the Indebtedness of any third party (including any shareholder, partner, member, principal or affiliate of Borrower, or any shareholder, partner, member, principal or Affiliate thereof);

(xvii) fail to allocate fairly and reasonably among Borrower and any third party (including, without limitation, any Guarantor) any overhead for common employees, shared office space or other overhead and administrative expenses;

(xviii) allow any Person to pay the salaries of its own employees or fail to maintain a sufficient number of employees for its contemplated business operations; provided, however, that the foregoing shall not require any parent or member of Borrower to make any additional capital contributions to Borrower;
(xix) fail to maintain adequate capital for the normal obligations reasonably foreseeable in a business of its size and character and in light of its

contemplated business operations; provided, however, that the foregoing shall not apply to the extent that there is insufficient revenue generated by the Property;
(xx) file a voluntary petition or otherwise initiate proceedings seeking liquidation, reorganization or other relief under any Federal, state or foreign

Debtor Relief Laws, for Borrower or any general partner, manager or managing member of Borrower, or consent to the institution of, or fail to contest in a timely and appropriate manner, and proceeding or petition under Debtor Relief Laws against Borrower or any general partner, manager or managing member of Borrower, or file a petition seeking or consenting to reorganization or relief of Borrower or seek or consent to the appointment of any trustee, receiver, conservator, assignee, sequestrator, custodian, liquidator (or other similar official) of Borrower or any general partner, manager or managing member of Borrower or of all or any substantial part of the properties and assets of Borrower or any general partner, manager or managing member of Borrower, or make any general assignment for the benefit of creditors of Borrower or any general partner, manager or managing member of Borrower, or admit in writing the inability of Borrower or any general partner, manager or managing member of Borrower to pay its debts generally as they become due or declare or effect a moratorium on Borrower or any general partner, manager or managing member of Borrower debt or take any action in furtherance of any such action;
(xxi) hold itself out as or be considered as a department or division of (x) any shareholder, partner, principal, member or Affiliate of Borrower, (y) any

Affiliate of a shareholder, partner, principal, member or Affiliate of Borrower, or (z) any other Person, or allow any Person to identify Borrower as a department or division of that Person; or (xxii) conceal assets from any creditor, or enter into any transaction with the intent to hinder, delay or defraud creditors of Borrower or the creditors of

The foregoing provisions of this Section 4.04 shall not operate to prohibit Borrower from entering into Swap Agreements otherwise permitted under this Agreement.

4.05 Management.

Borrower at all times shall provide for the competent and responsible management and operation of the Mortgaged Property. At all times, Borrower shall cause the Mortgaged Property to be managed by Borrower, an Affiliate of Borrower or an Approved Manager. All management contracts affecting the Mortgaged Property shall be terminable upon thirty (30) days' written notice without penalty or charge (except for unpaid accrued management fees). Any management contract or contracts affecting the Property must be approved in writing by Administrative Agent prior to the execution of the same, which approval shall not be unreasonably withheld, conditioned or delayed. Borrower shall cause any Approved Manager to execute and deliver to Administrative Agent an assignment and subordination of property management agreement in form and substance reasonably acceptable to Administrative Agent.

Appraisals.

Administrative Agent shall have the right to order new Appraisals of the Mortgaged Property from time to time. Each Appraisal is subject to review and approval by Administrative Agent. Borrower agrees upon demand to pay to Administrative Agent the cost and expense for such Appraisals and a fee for Administrative Agent's review of each Appraisal. Borrower's obligation to pay such cost and expense shall be limited to one Appraisal per year, unless the Appraisal is ordered after the occurrence of a Default or is required by Legal Requirement or is required in connection with Borrower's exercise of an extension option pursuant to Section 3.14.

4.07 Operating and Reserve Accounts.

Borrower shall maintain all operating and reserve accounts for the Mortgaged Property at JPMorgan.

4.08 Prohibited Distributions

After the occurrence and during the continuation of a Default or Unmatured Default, Borrower shall not make any dividend or distribution to its members, or make any other payment to Persons holding a direct or indirect ownership interest in Borrower or engage in any transaction that has a substantially similar effect.

4.09 Borrower's Right to Contest Legal Requirements.

Notwithstanding any provision of this Agreement or any of the other Loan Documents to the contrary, no Default or Unmatured Default shall occur hereunder as a result of the failure of Borrower or the Property or Improvements to comply with any Legal Requirement, including, without limitation, Environmental Laws, so long as the following conditions are satisfied:

(a) Borrower is contesting the applicability of such Legal Requirement to Borrower or the Property or Improvements in good faith and has so notified Lender;

(b) (c) Borrower has properly commenced and is diligently pursuing such contest; the contest will not materially impair the ability to ultimately comply with the contested Legal Requirement should the contest not be successful; (d) Borrower demonstrates to Administrative Agent's satisfaction that Borrower has the financial capability to undertake and pay for such contest and any corrective or remedial action then or thereafter likely to be necessary

Neither Administrative Agent nor any Lender is at risk for any material liability due to Borrower's non-compliance with such Legal Requirement; and (e) (f) Borrower's non-compliance with such Legal Requirement will not result in a Lien (other than a Permitted Encumbrance) or charge on the Property or the Improvements, the enforcement of which is not stayed by such contest or insured or bonded over to the satisfaction of Administrative Agent.

4.10 Government Regulation.

Borrower shall not (a) be or become subject at any time to any law, regulation, or list of any Governmental Authority (including, without limitation, the U.S. Office of Foreign Asset Control list) that prohibits or limits the Lenders from making any advance or extension of credit to Borrower's identity as may be requested by any Lender at any time to enable such Lender to verify Borrower's identity as may be requested by any Lender at any time to enable such Lender to verify Borrower's identity or to comply with any applicable law or regulation, including, without limitation, Section 326 of the USA Patriot Act of 2001, 31 U.S.C. Section

Financial Information and Other Deliveries. 4.11 Borrower. (i) Within forty-five (45) days after the end of each of Borrower's fiscal quarters, Borrower shall deliver to Administrative Agent (A) an operating statement (showing actual to budgeted results) and a lease status report (including a rent roll) for the Property and Improvements, each dated as of the last day of such fiscal quarter, and (B) a balance sheet, statement of

operations and statement of cash flow for Borrower, each dated as of the last day of such fiscal quarter, in form and substance satisfactory to Administrative Agent and certified by the chief financial officer of Borrower. (ii) Within ninety (90) days after the end of each of Borrower's fiscal years. Borrower shall deliver to Administrative Agent a balance sheet, statement of operations and statement of cash flow for Borrower, each dated as of the last day of such fiscal year, in form and substance satisfactory to Administrative Agent and certified by the chief financial officer of Borrower.

(iii) Within forty-five (45) days after each Test Date (as defined in Section 4.16 below) a compliance certificate in the form attached hereto as Exhibit G evidencing Borrower's satisfaction of the Minimum Required Debt Yield (as defined in Section 4.16 below).

(iv) Borrower shall promptly deliver to Administrative Agent written notice of (A) the occurrence of any Default or Unmatured Default or the occurrence of an event which would make any representation or warranty contained herein untrue or misleading in any material respect as of the date of such event, and (B) the occurrence of any monetary or material non-monetary default under any Major Lease.

(v) Borrower shall deliver to Administrative Agent such other information and materials with respect to Borrower, the Mortgaged Property, the Guarantor, or compliance with the terms of this Agreement, as the Administrative Agent or any Lender may reasonably request.

(i) Within ninety (90) days after the end of each of Guarantor's fiscal quarters, Borrower shall deliver to Administrative Agent a balance sheet, statement of operations and statement of cash flow for Guarantor, each dated as of the last day of such fiscal quarter, in form and substance satisfactory to Administrative Agent and certified by the chief financial officer of Guarantor;

(ii) Within one hundred twenty (120) days after the end of each of Guarantor's fiscal quarters, Borrower shall deliver to Administrative Agent a balance

sheet, a statement of operations and a statement of cash flow for Guarantor, each dated as of the last day of such fiscal year, in form and substance satisfactory to Administrative Agent and audited by an independent a

(m) Borrower shall deriver to Administrative Agent such diller information and materials with respect to Guarantor as Administrative Agent s request.

4.12 Hazardous Substances.

(b)

Borrower warrants, represents and covenants as follows:

Guarantor.

(a) Report; Compliance with Environmental Laws. Borrower has caused the preparation of the Report, and except as disclosed in the Report, to the actual knowledge of Borrower, Borrower and the Property are in full compliance with all Environmental Laws.

(b) No Liens. Notices or Actions. Except as disclosed in the Report, to Borrower's knowledge, neither Borrower nor the Property are subject to any private or governmental Lien or judicial or administrative notice or action pending, or threatened in writing, relating to Hazardous Substances or the environmental condition of the Property.

(c) No Hazardous Substances; Compliance with Environmental Laws. Except as disclosed in the Report, to Borrower's actual knowledge, no Hazardous Substances are located on or have been stored, processed or disposed of on or released or discharged from (including ground water contamination) the Property, and no above or underground storage tanks exist on the Property. Borrower shall not allow any Hazardous Substances to be stored, located, discharged, possessed, managed, processed or otherwise handled on the Property and shall comply with all Environmental Laws affecting the Property.

(d) Notice. Borrower shall immediately notify Administrative Agent should Borrower become aware of (i) any Hazardous Substance or other environmental problem or liability with respect to the Property or (ii) any Lien, action, or notice of the nature described in Section 4.15(b) above.

4.13

ERISA.

(a) Plan Assets: Compliance; No Material Liability. Borrower hereby covenants and agrees that (i) Borrower shall not use any Plan Assets to repay or secure the Obligations, (ii) no assets of Borrower or Guarantor are or will be Plan Assets, (iii) each Employee Benefit Plan will be in material compliance with all applicable requirements of ERISA and the Code except to the extent any defects can be remedied without material liability to Borrower under Revenue Procedure 2008-50 or any similar procedure, and (iv) Borrower will not have any material liability under Title IV of ERISA or Section 412 of the Code with respect to any Employee Benefit Plan.

Transfer of Interests. In addition to the prohibitions set forth in this Agreement and the other Loan Documents, and not in limitation thereof, Borrower hereby covenants (b) and agrees that Borrower shall not assign, sell, pledge, encumber, transfer, hypothecate or otherwise dispose of its interests or rights (direct or indirect) in any Loan Document or any portion of the Mortgaged Property or attempt to do any of the foregoing or suffer any of the foregoing, or permit any party with a direct or indirect interest or right in any Loan Document or any portion of the Mortgaged Property to do any of the foregoing, if such action would cause this Agreement, any of the other Loan Documents, or the Obligations or the exercise of any of Administrative Agent's or any Lender's rights in connection therewith, to constitute a prohibited transaction under ERISA or the Code (unless Borrower furnishes to Administrative Agent a legal opinion satisfactory to Administrative Agent that the transaction is exempt from the prohibited transaction provisions of ERISA and the Code) or would otherwise result in the Mortgaged Property, or assets of Borrower or Guarantor being Plan Assets.

Indemnity. Borrower hereby agrees to indemnify Administrative Agent, each Lender, their respective affiliates, and each of their directors, officers and employees against all losses, claims, damages, penalties, judgments, liabilities and expenses (including, without limitation, all reasonable expenses of litigation or preparation therefor whether or not Administrative Agent, any Lender or any Affiliate is a party thereto) which any of them may actually pay or incur by reason of the investigation, defense and settlement of claims and in obtaining any prohibited transaction exemption under ERISA or the Code necessary in Administrative Agent's or any Lender's judgment by reason of the inaccuracy of the representations and warranties set forth in Section 6.01(i) hereof or a breach of the provisions set

 $for th in this \ Section \ 4.13. \ The \ obligations \ of \ Borrower \ under \ this \ Section \ 4.13 \ shall \ survive \ the \ termination \ of \ this \ Agreement \ the \ this \ Section \ 4.13 \ shall \ survive \ the \ termination \ of \ this \ Agreement \ this \ Section \ 4.13 \ shall \ survive \ the \ termination \ of \ this \ Agreement \ this \ Section \ 4.13 \ shall \ survive \ the \ termination \ of \ this \ Agreement \ this \ Section \ 4.13 \ shall \ survive \ the \ termination \ of \ this \ Agreement \ this \ Section \ 4.13 \ shall \ survive \ the \ termination \ of \ this \ Agreement \ this \ Section \ 4.13 \ shall \ survive \ the \ termination \ of \ this \ Agreement \ this \ Section \ 4.13 \ shall \ survive \ the \ termination \ of \ this \ Section \ this \ Section$

4.14 Compliance with Laws.

Borrower shall comply with all Requirements of Law applicable to it or its property (including, without limitation, the Mortgaged Property).

4.15 Sanctions Laws and Regulations.

(a) The Borrower shall not, directly or indirectly, use the proceeds of the Loans, or lend, contribute or otherwise make available such proceeds to any Subsidiary, joint venture partner or other person or entity (i) to fund any activities or business of or with any Designated Person, or in any country or territory, that at the time of such funding is the subject of any sanctions under any Sanctions Laws and Regulations, or (ii) in any other manner that would result in a violation of any Sanctions Laws and Regulations by any party to this Agreement.

(b) None of the funds or assets of the Borrower that are used to pay any amount due pursuant to this Agreement shall constitute funds obtained from transactions with or relating to Designated Persons or countries which are the subject of sanctions under any Sanctions Laws and Regulations.

4.16 Debt Yield.

(a) Borrower will not permit the Debt Yield Ratio to be less than 9.25% (the "Minimum Required Debt Yield"), which shall be tested as of June 30 of each year commencing with June 30, 2015 (each a "Test Date"). If, as of any Test Date, the Debt Yield Ratio is less than the Minimum Required Debt Yield, then Borrower shall deposit all Excess Cash Flow generated by the Property during each month into the Borrower's Funds Account, commencing with the month immediately preceding the applicable Test Date and continuing each month thereafter (the "Cash Flow Sweep") until the occurrence of a Cash Flow Sweep Termination Event (as defined below). Required deposits of Excess Cash Flow shall be made within ten (10) Business Days after the end of each month.

(b) If, after commencement of any Cash Flow Sweep as provided above, a Cash Flow Sweep Termination Event shall occur, then (i) the Cash Flow Sweep then in effect shall terminate, (ii) except as otherwise provided below with respect to future Test Dates, Borrower shall have no further obligation to deposit Excess Cash Flow in the Borrower's Funds Account, and (iii) upon Borrower's written request, so long as no Default or Unmatured Default shall have occurred and be continuing, Administrative Agent shall release to Borrower any and all funds then on deposit in the Borrower's Funds Account. Notwithstanding any suspension of the requirement for the Excess Cash Flow to be deposited or the release of funds from the Borrower's Funds Account, the requirements of this Section 4.16 shall remain in full force and effect with respect to any future Test Date. As used herein, "Cash Flow Sweep Termination Event" shall mean the occurrence of either of the following: (1) Borrower shall have paid to Administrative Agent a Remargin Payment as provided in Section 4.16(c) below, or (2) the Minimum Required Debt Yield shall be satisfied as of the Test Date immediately following the commencement of the applicable Cash Flow Sweep.

(c) If the Debt Yield Ratio is less than the Minimum Required Debt Yield for two consecutive Test Dates, Borrower shall, within ten (10) Business Days following the second of such Test Dates, pay to Administrative Agent (each such payment, a "Remargin Payment") for application to the Obligations an amount sufficient to cause the Minimum Required Debt Yield to be satisfied as of such second Test Date. At Borrower's request in connection with any required Remargin Payment, Lender shall apply the amount then on deposit in the Borrower's Funds Account to any such required Remargin Payment, provided, however, if the amount then on deposit in the Borrower's Funds Account is not sufficient to pay in full the required Remargin Payment, Borrower shall pay the deficiency to Lender from other sources as and when due under this Section 4.16.

4.17 Borrower's Funds Accounts.

The Borrower's Funds Account shall be an interest bearing account, with all accrued interest to become part of the funds on deposit in such Borrower's Funds Account. Borrower agrees that it shall include all interest and earnings on any such deposit as its income (and, if Borrower is a partnership or other pass-through entity, the income of its partners, members or beneficiaries, as the case may be), and shall be the owner of all funds on deposit in the Borrower's Funds Account for federal and applicable state and local tax purposes. Administrative Agent shall have the exclusive right to manage and control all funds in the Borrower's Funds Account, but Administrative Agent shall have no fiduciary duty with respect to such funds (but Administrative Agent shall be liable for its own gross negligence or willful misconduct in respect to the funds from time to time held in the Borrower's Funds Account). Borrower grants to Administrative Agent a security interest in any Borrower's Funds Account established and all funds now or hereafter deposited into any such account, and any proceeds thereof, as security for the Obligations. Such security interest shall be governed by the Uniform Commercial Code of the State of California, and Administrative Agent shall have available to it all of the rights and remedies available to a secured party (including accrued interest thereon) to the payment of the foregoing, upon the occurrence and during the continuance of a Default, Administrative Agent may apply all or any portion of the funds on deposit in any Borrower's Funds Account (including accrued interest thereon) to the payment of the Obligations. The Borrower's Funds Account may be established and held in such name or names as Administrative Agent shall deem appropriate, including in the name of Administrative Agent. Borrower hereby constitutes and appoints Administrative Agent and any officer or agent of Administrative Agent and lawfull attorneys-in-fact with full power of substitution during the existence of a Default to op

$\frac{\text{ARTICLE V}}{\text{COLLATERAL ASSIGNMENT AND SECURITY AGREEMENT}}$

5.01

Project Documents and Swap Agreements.

<u>Collateral Assignment and Security Agreement.</u> As additional security for the Obligations, Borrower hereby sells, assigns, transfers and sets over to Administrative (a)

Agent, for the benefit of the Lenders, and grants to Administrative Agent, for the benefit of the Lenders, a security interest in, all of its right, title and interest in and to the Project Documents and any Swap Agreements.

(b) Performance: Enforcement. Borrower shall perform and observe in a timely manner all material covenants, conditions, obligations and agreements on the part of Borrower to be performed or observed under the Project Documents and any Swap Agreements. Borrower shall not waive, excuse, condone or in any manner release or discharge any party to a Project Document or any Swap Agreement from any material covenants, conditions, obligations or agreements to be performed or observed by such party under such Project Document or Swap Agreement, as applicable, but shall, at its sole cost and expense, enforce and secure the performance of all material covenants, conditions, obligations and agreements to be observed by all parties under the Project Documents and any Swap Agreements.

Remedies Upon Default. Upon the occurrence and during the continuance of a Default, Administrative Agent shall have the right but not the obligation, and Borrower (c) hereby authorizes Administrative Agent to enforce Borrower's rights under the Project Documents and any Swap Agreements and to receive the performance of any other Person that is a party to the Project Documents and any Swap

(d) Notices of Default. Borrower shall send to Administrative Agent any written notice of material default or breach of or under the Project Documents or any Swap

Agreements that Borrower sends to (such notice to Administrative Agent to be sent simultaneously therewith) or receives from (such notice to Administrative Agent to be sent immediately upon receipt by Borrower thereof) any Person that is a party to any Project Document or Swap Agreement.

Power of Attorney. Effective upon the occurrence and during the continuance of a Default, Borrower hereby irrevocably constitutes and appoints Administrative Agent as its attorney-in-fact, coupled with an interest, to demand, receive and enforce Borrower's rights with respect to the Project Documents and any Swap Agreements, to give appropriate receipts, releases and satisfactions for and on behalf of Borrower and to do any and all acts in the name of Borrower or in the name of Administrative Agent with the same force and effect as if Borrower had performed such acts.

(f) License. Provided no Default has occurred and is continuing, Borrower shall have the right under a license granted hereby to exercise its rights under the Project Documents and any Swap Agreements. The license granted hereby shall be revoked at Administrative Agent's option upon written notice from Administrative Agent to Borrower after the occurrence and during the continuance of a Default.

49

(g) No Assumption of Liability. Administrative Agent does not hereby assume any of Borrower's obligations or duties under the Project Documents or any Swap Agreements, including, without limitation, the obligation to pay for services rendered thereunder.

(h) Validity and Enforceability of Project Documents and Swap Agreements. Borrower represents and warrants that, to Borrower's actual knowledge, the Project Documents and any Swap Agreements are valid, binding and enforceable (subject to Debtor Relief Laws and general equitable principles), are in full force and effect, and there are no material breaches or defaults thereunder and no events have occurred which with notice and/or lapse of time will constitute a material breach or default thereunder by Borrower or any Affiliate of Borrower. Borrower represents and warrants that it has full power, right and authority to execute and enter into the Project Documents and any Swap Agreements.

(i) <u>Prior Conveyance or Limiting Actions.</u> Borrower represents and warrants that it has not previously conveyed, transferred or assigned the Project Documents or any Swap Agreements or any right, title or interest therein and has not executed any other instrument which might prevent or limit Administrative Agent from operating under the terms and provisions of the assignment contemplated hereby, and Borrower covenants and agrees not to do any of the foregoing.

(j) Execution and Amendment of Project Documents. Borrower shall not enter into any other Project Document or Swap Agreement, or alter, amend or change in any material respect, or terminate or cancel, any Project Document or Swap Agreement, in each case without obtaining Administrative Agent's prior written consent. Administrative Agent may require, as a condition to its approval of a Project Document or Swap Agreement, the execution by the contracting party of an agreement, in form and substance acceptable to Administrative Agent, whereby said contracting party (i) acknowledges the provisions of this Section 5.01, (ii) subordinates its claims against Borrower to payment in full of the Obligations and to the rights of Administrative Agent under the Loan Documents and (iii) agrees that upon the occurrence and during the continuance of a Default, Administrative Agent has the right (but not the obligation) to terminate the subject Project Document.

5.02 Additional Instruments.

At Administrative Agent's request, Borrower shall execute and deliver to Administrative Agent any and all assignments and other documents and instruments reasonably necessary to confirm the collateral assignments contemplated by this ARTICLE V.

ARTICLE VI REPRESENTATIONS AND WARRANTIES

6.01 Representations and Warranties.

As a material inducement to Administrative Agent and the Lenders to enter into this Agreement, and as an express condition to each Advance made hereunder, Borrower hereby represents and warrants,

as follows:

(a) Existence; Power and Authority. Borrower is a limited liability company duly formed and validly existing in the State of Delaware and in good standing under the laws of the States of California and Delaware, with requisite power and authority to (i) incur the Obligations, and (ii) execute, deliver and perform this Agreement and the other Loan Documents to which it is a party.

(b) Authorization; No Conflict. Borrower's execution and delivery to Administrative Agent of this Agreement and the other Loan Documents and the full and complete performance of the provisions thereof (i) are authorized by Borrower's operating agreement; (ii) have been duly authorized by all requisite member actions; (iii) do not require the approval or consent of any Governmental Authority having jurisdiction over Borrower or any of the Mortgaged Property; and (iv) will not result in any breach of, or constitute a default under, or result in the creation of any Lien (other than those contained in any of the Loan Documents) upon any property or assets of Borrower under any indenture, mortgage, deed of trust, bank loan or credit agreement or other instrument or agreement to which Borrower is a party or by which Borrower or any of the Mortgaged Property is bound.

(c) Title. Borrower is the sole legal and beneficial owner of the fee simple interest in the Mortgaged Property free and clear of all Liens other than the Permitted

(d) Financial Statements. Any and all balance sheets, statements of income or loss, and financial statements heretofore furnished to Administrative Agent with respect to Borrower and Guarantor are true and correct in all material respects as of the dates thereof, and fully and accurately present the financial condition of the subjects thereof as of the dates thereof, and no material adverse change has occurred in the financial condition reflected therein since the dates of the most recent thereof. Neither Borrower nor Guarantor has any contingent liabilities, liabilities for taxes, unusual forward or long-term commitments or unrealized or anticipated losses from any unfavorable commitments which are reasonably likely to result in a material adverse effect on the Mortgaged Property or the development, construction or operation of the Improvements as contemplated by the

Loan Documents or on the financial condition of Borrower or Guarantor, or their respective abilities to perform their obligations under the Loan Documents and the Project Documents.

(e)

<u>Litigation.</u> There are no actions, suits or other legal proceedings pending, or to the actual knowledge of Borrower, threatened in writing, against or affecting Borrower, the Mortgaged Property, or the Guarantor which (i) if adversely determined would materially and adversely affect the ability of Borrower or Guarantor to perform its respective obligations under the Loan Documents or would have a material adverse effect on the use or value of the Mortgaged Property, or (ii) challenge the validity or enforceability of the Loan Documents or the priority of the Lien and security interest created thereby.

Legal Compliance. To Borrower's knowledge, the Improvements and the use and occupancy thereof comply in all material respects with all applicable Legal Requirements. Neither the zoning nor any other right to construct, use or operate the Improvements is to any extent dependent upon or related to any real estate other than the

Property. All approvals, licenses and permits required from Governmental Authorities under applicable Legal Requirements in connection with the use and occupancy of the Improvements have been obtained.

(g) Services and Utilities. All streets, easements, utilities and related services necessary for the operation of the Improvements and the operation thereof for their intended purpose are available to the Property.

(h) Enforceability. Each Loan Document executed by Borrower constitutes a legal and binding obligation of, and is valid and enforceable against, Borrower in accordance with the terms thereof (subject to Debtor Relief Laws and general equitable principles) and is not subject to any right of rescission, set-off, counterclaim or defense.

(i) ERISA. Borrower is not an "employee benefit plan" as defined in Section 3(3) of ERISA or a "plan" as defined in Section 4975(e)(1) of the Code. Each Employee Benefit Plan is in material compliance with all applicable requirements under ERISA and the Code, and, to the extent that such Employee Benefit Plan is also intended to be "qualified" within the meaning of Section 401(a) of the Code, it is in material compliance with the applicable requirements under the Code, except to the extent that any defects can be remedied without material liability to Borrower under Revenue Procedure 2008-50 or any similar procedure. None of the Employee Benefit Plans is subject to the requirements of Section 412 of the Code, Part 3 of Title I of ERISA or Title IV of ERISA or is a "multiemployer plan" as defined in Section 3(37) of ERISA. Borrower has no material liability under Title IV of ERISA or Section 412 of the Code with respect to any Employee Benefit Plan.

(j) Legal Parcel; Separate Tax Parcel. Each Property comprising the Property is taxed separately and does not include any other property, and for all purposes each Property comprising the Property may be mortgaged, conveyed and otherwise dealt with as a separate legal parcel.

(k) Leases and Rents. Borrower has good and marketable title to the Leases and rents free and clear of all claims and Liens other than the Permitted Encumbrances. To the knowledge of Borrower, and except as set forth in any estoppel certificates delivered to Lender by tenants prior to the date hereof, the Leases are valid and unmodified and are in full force and effect and Borrower is not in default of any of the material terms or provisions of the Leases. The rents now due or to become due for any periods subsequent to the date hereof have not been collected and payment thereof has not been anticipated for a period of more than one month in advance, waived or released, discounted, set off or otherwise discharged or compromised. Borrower has not received any funds or deposits from any Lessee for which credit has not already been made on account of accrued rents other than security deposits required by the Leases.

(I) Compliance with Laws and Agreements. Except where the failure to do so, individually or in the aggregate could not reasonably be expected to result in a material adverse effect, the Borrower is in compliance with (i) its charter, by-laws or other organizational documents, (ii) all Requirements of Law applicable to it or its property (including, without limitation, the Mortgaged Property) and (iii) all indentures, agreements and other instruments

binding upon it or its property (including, without limitation, the Mortgaged Property). No Default has occurred and is continuing.

(m) Sanctions Laws and Regulations. None of the Borrower, or any of its directors, officers, brokers or other agents acting or benefiting in any capacity in connection with this Agreement or any other capital raising transaction involving any Lender, or any of its Affiliates is a Designated Person.

6.02 Nature of Representations and Warranties.

All representations and warranties made in this Agreement or any other Loan Document or in any certificate or other document delivered to Administrative Agent pursuant to or in connection with this Agreement shall be deemed to have been relied upon by Administrative Agent and the Lenders notwithstanding any investigation heretofore or hereafter made by Administrative Agent or on its behalf.

ARTICLE VII

INSURANCE AND CONDEMNATION

7.01 Insurance and Casualty.

(a) Required Insurance Coverage. Borrower, at its expense, shall maintain and provide to Administrative Agent copies of policies or other satisfactory evidence of insurance providing the following:

(i) Commercial General Liability Insurance with limits of not less than \$1,000,000 per occurrence combined single limit and \$2,000,000 in the aggregate for the policy period, or in whatever higher amounts as may be required by Administrative Agent from time to time by notice to Borrower (with deductibles acceptable to Lender), and extended to cover: (a) Contractual Liability assumed by Borrower with defense provided in addition to policy limits for indemnities of the named insured, (b) Broad Form Property Damage Liability, (c) Products & Completed Operations for coverage, such coverage to apply for two years following the Closing Date, (d) waiver of subrogation against all parties named additional insured, (e) severability of interest provision, and (f) Personal Injury &

(ii) Automobile Liability including coverage on owned, hired and non-owned automobiles and other vehicles, if used in connection with the performance of the work, with Bodily Injury and Property Damage limits of not less than \$1,000,000.00 per occurrence combined single limit, with a waiver of subrogation against all parties named as additional insured. Umbrella/Excess Liability in excess of Commercial General Liability, Automobile Liability and Employers' Liability coverages which is at least (iii) as broad as these underlying policies with a limit of liability of \$25,000,000.00).

(iv) All-Risk Property (Special Cause of Loss) Insurance including, without limitation, coverage for loss or damage to the Property and Improvements

and other perils including windstorm, earthquake/earth movement and malicious mischief, building ordinance extension endorsement (including cost of demolition, increased costs of construction and the value of the undamaged portion of the building and soft costs coverage), and boiler and machinery coverage (if separate policy, that policy must include loss of rents or business interruption coverage), as specified by the Administrative Agent. The policy shall be in an amount not less than the full insurable value on a replacement cost basis of the insured Property and Improvements and personal property related thereto (without deduction for depreciation). If the policy is a blanket policy covering the Property and Improvements and one or more other properties, the policy must specify the dollar amount of the total blanket limit of the policy that is allocated to each property, and the amount so allocated to the Property and Improvements must not be less than the full insurable value on a replacement cost basis. During any construction period, such policy shall be written in the so-called "Builder's Risk Completed Value Non-Reporting Form" with no coinsurance requirement and shall contain a provision granting the insured permission to occupy prior to completion. Such policy shall not contain an exclusion for terrorist losses. However, if such an exclusion exists in the All-Risk policy, a separate Terrorism policy covering Certified Acts of Terrorism must be evidenced to the Administrative Agent in an amount equal to the full replacement cost of the Property and Improvements, or the amount of the Loan, whichever is less. This policy must also list the Administrative Agent as mortgagee and loss payee. Administrative Agent hereby notifies Borrower that the Loans are subject to California Civil Code Section 2955.5(a), which provides that: "No lender shall require a borrower, as a condition of receiving or maintaining a loan secured by real property, to provide hazard insurance coverage against risks to the improvement

(v) Workers' Compensation and Employer's Liability Insurance in accordance with the applicable laws of the state in which the work is to be performed or of the state in which Borrower is obligated to pay compensation to employees engaged in the performance of the work. The policy limit under the Employer's Liability Insurance section shall not be less than \$1,000,000.00 for any one accident.

(vi) If at any time, the Improvements, or any part thereof, lies within a "special flood hazard area" as designated on maps prepared by the Federal Emergency Management Agency (FEMA), a one hundred year flood plain or other area identified by Administrative Agent as having a high or moderate risk of flooding, a flood insurance policy or policies (whether or not coverage is available from the National Flood Insurance Program and whether or not required by the Flood Laws), in form and substance acceptable to the Administrative Agent and the Lenders covering the Improvements and contents (to the extent the contents secure the Obligations), for the duration of the Loans in an amount at least equal to the full insurable value on a replacement cost basis (without deduction for depreciation) of the Improvements and contents, if applicable, or the amount of the Obligations, whichever is less.

(vii) Rent loss or business interruption insurance against loss of income (including, but not limited to, rent, cost reimbursements and all other amounts payable by tenants under Leases or otherwise derived by Borrower from the operation of the

Property) arising out of damage to or destruction of the Property and Improvements by fire or other peril (including earthquake if required) insured against under each policy. The amount of the policy shall be in the amount equal to one year's projected rentals or gross revenue. A pollution legal liability policy with limits as required by Administrative Agent covering pollution conditions and pollution liability exposures

at the 9755 Distribution Property.

(ix) Such other insurance coverages in such amounts as Administrative Agent may reasonably require, which may include, without limitation, errors and omissions insurance with respect to the contractors, architects and engineers, and earthquake insurance; provided, however, that Administrative Agent shall not require Borrower to maintain earthquake insurance unless a so called "PML" report with respect to the applicable Property reflects a scenario expected loss of twenty percent (20%) or more in the event of an earthquake or other seismic casualty.

(b) Policy Requirements: Insurance Consultant. All insurance policies shall (i) be issued by an insurance company licensed to do business in the state where the Property is located having a rating of "A-" VIII or better by A.M. Best Co., in Best's Rating Guide, (ii) name "JPMorgan Chase Bank, N.A. and any and all subsidiaries and their successors and/or assigns as their interests may appear" as additional insureds on all liability insurance and as mortgagee and loss payee on all All-Risk Property, flood insurance, earthquake insurance and rent loss or business interruption insurance (whether or not required hereunder), (iii) be endorsed to show that Borrower's insurance shall be primary and all insurance carried by Administrative Agent is strictly excess and secondary and shall not contribute with Borrower's insurance, (iv) provide that Administrative Agent is to receive thirty (30) days written notice prior to non-renewal or cancellation, (v) be evidenced by a certificate of insurance to be provided to Administrative Agent along with a copy of the policy for All-Risk Property coverage or such other

evidence of insurance acceptable to Administrative Agent in its reasonable discretion, (vi) include either policy or binder numbers on the ACORD form, and (vii) be in form and amounts acceptable to Administrative Agent; provided, however, that with respect to any flood insurance required hereunder, acceptable proof of coverage shall consist of a copy of the insurance policy, the declarations page of the insurance policy or an application plus proof of premium payment (with a copy of the policy or declarations page provided to Administrative Agent within 30 days thereafter) and shall not include ACORD or other forms of certificates of insurance. Administrative Agent, at its option and upon notice to Borrower, may retain, at Borrower's expense, an insurance consultant to review the insurance for the Property and Improvements to confirm that it complies with the terms and conditions set forth herein.

Evidence of Insurance: Payment of Premiums. Borrower shall deliver to Administrative Agent, at least five (5) days before the expiration of an existing policy, evidence acceptable to Administrative Agent of the continuation of the coverage of the expiring policy. If Administrative Agent has not received satisfactory evidence of such continuation of coverage in the time frame herein specified, Administrative Agent shall have the right, but not the obligation, to purchase such insurance for Administrative Agent's and the Lenders' interest only. Any amounts so disbursed by Administrative Agent pursuant to this Section shall be

repaid by Borrower within 10 days after written demand therefor. Nothing contained in this Section shall require Administrative Agent to incur any expense or take any action hereunder, and inaction by Administrative Agent shall never be considered a waiver of any right accruing to Administrative Agent on account on this Section. The payment by Administrative Agent of any insurance premium for insurance which Borrower is obligated to provide hereunder but which Administrative Agent believes has not been paid, shall be conclusive between the parties as to the legality and amounts so paid. Borrower agrees to pay all premiums on such insurance as they become due, and will not permit any condition to exist on or with respect to the Mortgaged Property which would wholly or partially invalidate any insurance thereon.

(d) Collateral Protection. Unless Borrower provides Administrative Agent with evidence satisfactory to Administrative Agent of the insurance coverage required by this Agreement, Administrative Agent may purchase insurance at Borrower's expense to protect Administrative Agent's and the Lenders' interest in the Mortgaged Property. This insurance may, but need not, protect Borrower's interest in the Mortgaged Property. The coverages that Administrative Agent purchases may not pay any claim that Borrower makes or any claim that is made against Borrower in connection with the Mortgaged Property. Borrower or Administrative Agent (as appropriate) may later cancel any insurance purchased by Administrative Agent purchases insurance for the Mortgaged Property, Borrower will be responsible for the actual costs of that insurance, including any charges imposed by Administrative Agent in connection with the placement of insurance, until the effective date of the cancellation or expiration of such insurance. The costs of the insurance may, at Administrative Agent's discretion, be added to Borrower's total principal obligation owing to Administrative Agent and the Lenders, and in any event shall be secured by the liens on the Mortgaged Property created by the Loan Documents. It is understood and agreed that (i) the costs of insurance obtained by Administrative Agent may be more than the costs of insurance borrower may be able to obtain on its own, and (ii) in the case of flood insurance, the amount of coverage may be more than required by the Flood Laws.

(e) No Liability: Assignment. Administrative Agent shall not by the fact of approving, disapproving, accepting, preventing, obtaining or failing to obtain any such insurance, incur any liability for the form or legal sufficiency of insurance contracts, solvency of insurance contracts, and to the fact of approximate the fact of a

foreclosure or Administrative Agent on behalf of the Lenders or other transferee in the event of such other transfer of title.

(f) No Separate Insurance. Borrower shall not carry any separate insurance on the Mortgaged Property concurrent in kind or form with any insurance required hereunder or contributing in the event of loss without Administrative Agent's prior written consent, and any such policy shall have attached a standard non-contributing mortgagee clause, with loss payable to Administrative Agent for the benefit of the Lenders, and shall otherwise meet all other requirements set forth herein.

Casualty Loss. (g)

- (i) If all or any part of the Mortgaged Property shall be damaged or destroyed by fire or other casualty, Borrower shall give immediate written notice and make a claim to the insurance carrier and Administrative Agent. With respect to any such casualty loss for which Borrower has an insurance claim with a value that exceeds One Million Dollars (\$1,000,000), Borrower hereby authorizes and empowers Administrative Agent, at Administrative Agent's option and in Administrative Agent's sole discretion as attorney-in-fact for Borrower, to make proof of loss, to adjust and compromise any claim under insurance policies, to appear in and prosecute any action arising from such insurance policies, to collect and receive insurance proceeds, and to deduct therefrom Administrative Agent's expenses incurred in the collection of such proceeds; provided, however, that the foregoing authorization and empowerment of Administrative Agent to act as attorney-in-fact for Borrower shall not become effective until the occurrence and during the continuance of a Default or until such time as Borrower fails to diligently pursue the collection of such insurance proceeds in Administrative Agent's reasonable opinion. The foregoing appointment is irrevocable, coupled with an interest and continuing so long as the Commitments or Obligations remain outstanding, and such rights, powers and privileges shall be exclusive in Administrative Agent (for the benefit of the Lenders), its successors and assigns.
- (ii) As sole loss payee on all policies of casualty insurance, Administrative Agent shall receive all insurance proceeds from any casualty loss, and shall hold the same in an interest-bearing account pending disposition in accordance with this Section. Borrower authorizes Administrative Agent to deduct from such insurance proceeds received by Administrative Agent all of Administrative Agent's costs and expenses (including, without limitation, reasonable attorneys' fees) incurred in connection with the collection thereof (the remainder of such insurance proceeds being referred to herein as "Net Casualty Proceeds"); provided, however, that, notwithstanding the foregoing, so long as no Default or Unmatured Default shall have occurred and be continuing, Borrower shall have the exclusive right to receive the proceeds of all insurance for loss of or damage to the Property in cases in which such claim is valued at less than \$1,000,000.
- Administrative Agent shall cause the Net Casualty Proceeds from any casualty loss affecting the Mortgaged Property to be disbursed for the cost of reconstruction of the Mortgaged Property if all of the following conditions are satisfied

within ninety (90) days after the applicable casualty loss: (A) Borrower satisfies Administrative Agent (in Administrative Agent's reasonable judgment) that the reconstruction can be completed within a reasonable period of time after such casualty loss (but in no event later than the Maturity Date) and that after giving effect to such reconstruction the Mortgaged Property will be restored to substantially the same condition immediately prior to the casualty loss; (B) Borrower satisfies Administrative Agent (in Administrative Agent and the Net Casualty Proceeds are sufficient to pay all costs of reconstruction, and if insufficient, Borrower deposits with Administrative Agent additional funds to make up such insufficiency; (C) Borrower delivers to Administrative Agent and specifications and construction contracts for the work of reconstruction and such plans and specifications and construction contracts are in form and content reasonably acceptable to Administrative Agent and with a contractor reasonably acceptable to Administrative Agent and with a contractor reasonably acceptable to Administrative Agent and with a contractor reasonably acceptable to Administrative Agent and with a contractor reasonably acceptable to Administrative Agent and with a contractor reasonably acceptable to Administrative Agent and with a contractor reasonably acceptable to Administrative Agent and with a contractor reasonably acceptable to Administrative Agent and with a contractor reasonably acceptable to Administrative Agent, and (D) Borrower delivers to Administrative Agent and with a contractor reasonably acceptable to Administrative Agent and with a contractor reasonably acceptable to Administrative Agent and with a contractor reasonably acceptable to Administrative Agent and with a contractor reasonably acceptable to Administrative Agent and with a contractor reasonably acceptable to Administrative Agent and with a contractor reasonably acceptable to Administrative Agent and with a contractor reasonable proceeds are made availab

Immediately upon receiving written notice of the institution or threatened institution of any proceeding for the condemnation of the Mortgaged Property or any part thereof, Borrower shall notify Administrative Agent of such fact. Borrower shall then file or defend its rights thereunder and prosecute the same with due diligence to its final disposition; provided, however, that Borrower shall not enter into any settlement of such proceeding without the prior approval of Administrative Agent (which approval shall not be unreasonably withheld, conditioned or delayed). Administrative Agent shall be entitled, at its option, to appear in any such proceeding in its own name on behalf of the Lenders, and upon the occurrence and during the continuation of a Default or if Borrower fails to diligently prosecute such proceeding (in Administrative Agent's reasonable judgment), (a) Administrative Agent shall be entitled, at its option, to appear in and prosecute any such proceeding or to make any compromise or settlement in connection with such condemnation on behalf of Borrower, and (b) Borrower hereby irrevocably constitutes and appoints Administrative Agent as its attorney-in-fact, and such

appointment is coupled with an interest, to commence, appear in and prosecute such action or proceeding or to make such compromise or settlement in connection with any such condemnation on its behalf. The foregoing appointment is irrevocable and continuing so long as the Commitments or Obligations remain outstanding, and such rights, powers and privileges shall be exclusive in Administrative Agent (for the benefit of the Lenders), its successors and assigns. If the Mortgaged Property or any material part thereof is taken or materially diminished in value in connection with such condemnation, or if a consent settlement is entered, by or under threat of such proceeding, the award or settlement payable to Borrower by virtue of its interest in the Mortgaged Property, shall be, and by these presents is, assigned, transferred and set over unto Administrative Agent for the benefit of the Lenders; provided, however, that, so long as no Default or Unmatured Default shall have occurred and be continuing, Borrower shall have the exclusive right to receive the award or settlement payable to Borrower in connection with any such condemnation in cases in which such award or settlement is valued at less than One Million Dollars (\$1,000,000). Any such award or settlement shall be first applied to reimburse Administrative Agent and the Lenders for all out-of-pocket costs and expenses, including reasonable attorneys' fees, incurred in connection with the collection of such award or settlement. The balance of such award or settlement (the "Net Condemnation Proceeds") shall be paid to Administrative Agent for the benefit of the Lenders for application in the manner set forth in Section 7.01(g) as if such award or settlement constituted insurance proceeds from a casualty loss; provided, however, that Administrative Agent shall have no obligation to make Net Condemnation Proceeds available for construction or reconstruction of the Mortgaged Property unless Administrative Agent has determined that the Mortgaged Property as so constructed after giving effect to the condemnation would have a value that is not materially less than its value would have been had there been no such condemnation. Borrower shall have the obligation to promptly and diligently complete the work of reconstruction necessitated by any condemnation and restore the Mortgaged Property to the equivalent of its condition immediately prior to such condemnation provided the applicable Net Condemnation Proceeds are made available to Borrower for such purpose; provided, however, that Borrower shall not be obligated to complete the work of reconstruction necessitated by any condemnation and restore the affected Property to the equivalent of its condition immediately prior to such condemnation unles Administrative Agent shall make available to Borrower all Net Condemnation Proceeds received by Administrative Agent in connection with such condemnation.

ARTICLE VIII

DEFAULTS

8.01 Defaults.

Any of the following events, after passage of the applicable cure period set forth below, shall constitute a "Default" hereunder:

Failure to Make Payment. The failure by Borrower to pay in full any principal of the Loans when due (including, without limitation, any Remargin Payment required under Section 4.16(c) above); the failure by Borrower to pay in full any interest on the Loans, or any fees or any other amounts due under the Loan Documents (other than principal) when due and such failure continues unremedied for a period of ten (10) days after the due date thereof; or

the failure by Borrower to make any other payment or deposit required hereunder or under any of the other Loan Documents within the period set forth in Loan Documents, or if no period is set forth in the Loan Documents, then within ten (10) Business Days after written demand therefor;

Involuntary Proceeding. An involuntary proceeding shall be commenced or an involuntary petition shall be filed seeking (i) liquidation, reorganization or other relief in respect of Borrower or any Guarantor or its debts, or of a substantial part of its assets, under any Federal, state or foreign bankruptcy, insolvency, receivership or similar law now or hereafter in effect or (ii) the appointment of a receiver, trustee, custodian, sequestrator, conservator or similar official for Borrower or any Guarantor or for a substantial part of its assets, and, in any such case, such proceeding or petition shall continue undismissed for ninety (90) days or an order or decree approving or ordering any of the foregoing shall be entered;

Voluntary Proceedings. Borrower or any Guarantor shall (i) voluntarily commence any proceeding or file any petition seeking liquidation, reorganization or other relief (c) under any Federal, state or foreign bankruptcy, insolvency, receivership or similar law now or hereafter in effect, (ii) consent to the institution of, or fail to contest in a timely and appropriate manner, any proceeding or petition described in clause (b) of this Section, (iii) apply for or consent to the appointment of a receiver, trustee, custodian, sequestrator, conservator or similar official for Borrower or any Guarantor or for a substantial part of its assets, (iv) file an answer admitting the material allegations of a petition filed against it in any such proceeding, (v) make a general assignment for the benefit of creditors or (vi) take any action for the purpose of effecting any of the foregoing;

Assignment for Benefit of Creditors. The execution by Borrower or Guarantor of an assignment for the benefit of creditors;

(d) <u>Unable to Pay Debts</u>. The admission in writing in any court proceeding by Borrower or Guarantor that it is unable to pay its debts as they mature or that it is generally (e) not paying its debts as they mature;

(f) <u>Liquidation of Borrower or Guarantor</u>. The liquidation, termination or dissolution of Borrower or Guarantor; Transfer or Encumbrance of Interest in Mortgaged Property or Borrower.

Mortgaged Property. Except for any sales, exchanges, conveyances and transfers at the closing of which the Loan is fully repaid (or the applicable

(i) Minimum Release Price is paid to Administrative Agent pursuant to Section 3.15, as the case may be) and any Permitted Encumbrance, the sale, lease (except as permitted under this Agreement), exchange, conveyance, transfer, mortgage, assignment, pledge or encumbrance, either voluntarily or involuntarily, or the agreement to do so, of any right, title or interest of Borrower in and to the Mortgaged Property or any portion thereof, which occurrence is not rendered ineffective within ten (10) days after occurrence; provided, however, that Borrower shall be permitted to replace defective,

- obsolete or worn out personal property, and Borrower shall be permitted to grant and/or record Permitted Encumbrances;
 (ii) Borrower. The sale, exchange, conveyance, transfer, mortgage, assignment, pledge or encumbrance, either voluntarily or involuntarily, or the agreement to do so, of any direct or indirect ownership interest in Borrower or any portion thereof; or any change (whether voluntary or involuntary) in the management or control of Borrower (collectively, a "Transfer"); provided, however, Borrower shall not be in Default hereunder (and any such Transfer to a Person that is not an Embargoes Person shall be expressly permitted) if, after giving effect to any of the foregoing, Guaranton shall hold, directly or indirectly, at least 51% of the ownership interests in Borrower;
- (h) Levy; Attachment; Seizure. The levy, attachment or seizure pursuant to court order of (i) any right, title or interest of Borrower in and to the Mortgaged Property or any portion thereof or (ii) any direct or indirect ownership interests in Borrower, if such order is not vacated or the proceeding in which it was entered is not dismissed, in each case, within forty-five (45) days of the entry and service of such order:
- Failure of Representations. Any representation or warranty contained herein or in any of the other Loan Documents, or in any certificate or other document executed by Borrower or Guarantor and delivered to Administrative Agent pursuant to or in connection with this Agreement, is not true and correct in all material respects, or omits to state a material fact necessary to make such representation not misleading, in each case, as of the date made or deemed made;
- (j) Claims; Liens; Encumbrances; Stop Notices. Unless Borrower is contesting the same in accordance with the provisions of Section 4.02(c) hereof, the filing of any claim of lien or encumbrance (other than Permitted Encumbrances) against the Mortgaged Property or any part thereof that is not released, bonded over or insured over with a title insurance endorsement (obtained at Borrower's cost and expense) within thirty (30) days after notice thereof from Administrative Agent to Borrower; or the service on Administrative Agent, any Lender or any disburser of funds of a notice or demand to withhold funds, which is not nullified within thirty (30) days after the date of such service;
- Permits; Utilities; Insurance. (i) The neglect, failure or refusal of Borrower to keep in full force and effect any material permit, license, consent or approval required for (k) the construction or operation of any Improvements that is not fully reinstated within thirty (30) days after Administrative Agent gives Borrower written notice of the lapse of effectiveness of such material permit, license, consent or approval; or (ii) the curtailment in availability to the Mortgaged Property of utilities or other public services necessary for the full occupancy and utilization of the Improvements that is not restored to full availability within thirty (30) days after Administrative Agent gives Borrower written notice of such curtailment of availability; or (iii) the failure by Borrower to maintain any insurance required under Section 7.01 hereof that is not cured within five (5) Business Days after Administrative Agent gives Borrower written notice of such lapse;

(l) <u>Cessation of Loan Documents to be Effective</u>. The cessation, for any reason, of any Loan Document to be in full force and effect in all material respects; the failure of any Lien intended to be created by the Loan Documents to exist or to be valid and perfected; the cessation of any such Lien, for any reason, to have the priority contemplated by this Agreement or the other Loan Documents, subject to Borrower's right to contest liens in accordance with the terms of this Agreement; or the revocation by Guarantor of the Limited Guaranty or any other Loan Document executed by Guarantor;

(m) ERISA. Any breach of the provisions of Section 4.13 hereof; Prohibited Distributions. Any breach of the provisions of Section 4.08 hereof shall occur which is not cured by Borrower within five (5) Business Days after (n)

Administrative Agent gives Borrower written notice of such breach; Operations of Borrower. Any breach of the provisions of Section 4.04 hereof shall occur which is not cured by Borrower within twenty (20) days after Administrative

(o)

Judgments. Any judgment or order for the payment of money in excess of \$100,000 is rendered against Borrower or in excess of \$5,000,000 is rendered against (p) Guarantor and either (a) enforcement proceedings have been commenced by a creditor upon such judgment, or (b) there is a period of fifteen (15) days during which a stay of enforcement of such judgment or order, by reason of a pending appeal or otherwise, is not in effect;

Guarantor Cross-Default. Failure by any Guarantor to pay when due any Indebtedness in an outstanding principal amount of \$5,000,000 or more in the aggregate (q) ("Material Guarantor Indebtedness"); or the default by any Guarantor in the performance (beyond the applicable grace period with respect thereto, if any) of any term, provision or condition contained in any loan agreement or other debt instrument, or any other event shall occur or condition exist, the effect of which default, event or condition is to cause, or to permit the holder(s) of such Material Guarantor Indebtedness to cause, such Material Guarantor Indebtedness to become due prior to its stated maturity or any commitment to lend under any such loan agreement or other debt instrument to be terminated prior to its stated expiration date; or any Material Guarantor Indebtedness shall be declared to be due and payable or required to be prepaid or repurchased (other than by a regularly scheduled payment) prior to the stated maturity thereof;

Death of Individual Guarantor. Any Guarantor that is a natural person dies;

Swap Agreements. The occurrence or existence of any default, event of default or other similar condition or event (however described) after the expiration of all (s) applicable notice and cure periods, if any, with respect to any Swap Agreement, whether or not any Lender or Affiliate of any Lender is a party thereto;

Borrower Cross-Default. Failure by Borrower to pay when due any Indebtedness in an outstanding principal amount of \$100,000 or more in the aggregate excluding the (t)

Loans ("Material Borrower Indebtedness"); or the default by Borrower in the performance

Agent gives Borrower notice thereof;

(beyond the applicable grace period with respect thereto, if any) of any term, provision or condition contained in any loan agreement or other debt instrument, or any other event shall occur or condition exist, the effect of which default, event or condition is to cause, or permit the holder(s) of such Material Borrower Indebtedness to cause, such Material Borrower Indebtedness to become due prior to its stated maturity or any commitment to lend under any such loan agreement or other debt instrument to be terminated prior to its stated expiration date; or any Material Borrower Indebtedness shall be declared to be due and payable or required to be prepaid or repurchased (other than by a regularly scheduled payment) prior to the stated maturity thereof;

(u) Borrower written notice thereof; or

Management of the Property. Any breach of the provisions of Section 4.05 hereof shall occur which is not cured by Borrower within fifteen (15) days after Lender gives

(v) Failure to Perform Covenants. The failure of Borrower to fully perform any and all covenants and agreements hereunder or under any of the other Loan Documents, and, with respect to covenants and agreements other than those specifically referenced in this Section 8.01, or for which another cure period is provided, such failure is not cured by Borrower within thirty (30) days after Administrative Agent gives written notice to Borrower thereof, unless (i) such failure, by its nature, is not capable of being cured within such period, (ii) within fifteen (15) days after delivery of such written notice, Borrower commences to cure such failure and thereafter diligently prosecutes the cure thereof, and (iii) Borrower causes such failure to be cured no later than ninety (90) days after the date of such written notice from Administrative Agent.

ARTICLE IX

ACCELERATION AND REMEDIES

9.01 Acceleration.

If any Default described in Section 8.01(b) or (c) hereof occurs with respect to Borrower, the obligations of the Lenders to make Loans hereunder shall automatically terminate and the Obligations (other than Swap Obligations included therein) shall immediately become due and payable without any election or action on the part of Administrative Agent or any Lender. If any other Default occurs and is continuing, the Required Lenders (or Administrative Agent with the consent of the Required Lenders) may declare the Obligations (other than Swap Obligations included therein) to be due and payable, or both, whereupon the Obligations (other than Swap Obligations included therein) shall become immediately due and payable, without presentment, demand, protest or notice of any kind, all of which Borrower hereby expressly waives. If the Required Lenders (in their sole discretion) shall so direct, Administrative Agent shall, by notice to Borrower, rescind and annul such acceleration and/or termination. Rights and remedies with respect to the Swap Obligations are set forth in the Swap Agreements

9.02

Other Remedies,

Upon the occurrence and during the continuance of a Default hereunder, Administrative Agent, on behalf of the Lenders, shall have the right, in person or by agent, in

addition to all other rights and remedies available to Administrative Agent under this Agreement and the other Loan Documents, to the fullest extent permitted by law, to take possession of the Mortgaged Property and perform any and all obligations of Borrower hereunder and under the other Loan Documents, and employ watchmen to protect the Mortgaged Property from injury. All reasonable sums so expended by Administrative Agent shall be deemed to have been paid to Borrower and constitute Obligations. Effective upon the occurrence and during the continuance of a Default, Borrower hereby constitutes and appoints Administrative Agent its true and lawful attorney-in-fact, with full power of substitution, to perform all obligations of Borrower hereunder and under the other Loan Documents in the name of Borrower. Borrower hereby empowers said attorney to: (a) prosecute and defend all actions or proceedings in connection with the Mortgaged Property or the construction of the Improvements and take such action and require such performance as it deems necessary under any bond or guaranty of completion; and (b) do any and every act which Borrower might do on its own behalf. It is further understood and agreed that this power of attorney, which shall be deemed to be a power coupled with an interest, cannot be revoked.

9.03

Curing of Defaults.

Upon the occurrence and during the continuance of a Default hereunder, Administrative Agent, without waiving any right of acceleration or foreclosure under the Loan Documents which Administrative Agent or the Lenders may have by reason of such Default or any other right Administrative Agent or the Lenders may have against Borrower because of said Default, shall have the right (but not the obligation) to take such actions and make such payments as shall be necessary to cure such Default, including, without limitation, the making of Advances. All amounts so expended shall constitute Obligations and shall be payable by Borrower on demand by Administrative Agent.

ARTICLE X THE ADMINISTRATIVE AGENT

10.01

Appointment.

Each of the Lenders hereby irrevocably appoints Administrative Agent as its agent and authorizes Administrative Agent to take such actions on its behalf and to exercise such powers as are delegated to Administrative Agent by the terms hereof, together with such actions and powers as are reasonably incidental thereto.

10.02 Capacity as Lender

The bank serving as the Administrative Agent hereunder shall have the same rights and powers in its capacity as a Lender as any other Lender and may exercise the same as though it were not the Administrative Agent, and such bank and its Affiliates may accept deposits from, lend money to and generally engage in any kind of business with Borrower or any Subsidiary or other Affiliate thereof as if it were not the Administrative Agent hereunder

Duties and Obligations.

Administrative Agent shall not have any duties or obligations except those expressly set forth herein. Without limiting the generality of the foregoing, (a) Administrative Agent shall not be subject to any fiduciary or other implied duties, regardless of whether a Default or Unmatured Default has occurred and is continuing, (b) Administrative Agent shall not have any duty to take any discretionary action or exercise any discretionary powers, except discretionary rights and powers expressly contemplated hereby that Administrative Agent is required to exercise in writing as directed by the Required Lenders (or such other number or percentage of the Lenders as shall be necessary under the circumstances as provided in this Agreement), and (c) except as expressly set forth herein, Administrative Agent shall not have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to Borrower or any of its subsidiaries or Affiliates that is communicated to or obtained by the bank serving as Administrative Agent or any of its Affiliates in any capacity. Administrative Agent shall not be liable for any action taken or not taken by it with the consent or at the request of the Required Lenders (or such other number or percentage of the Lenders as shall be necessary under the circumstances as provided in this Agreement) or in the absence of its own gross negligence or willful misconduct. Administrative Agent shall be deemed not to have knowledge of any Default or Unmatured Default unless and until written notice thereof is given to Administrative Agent by Borrower or a Lender, and Administrative Agent shall not be responsible for or have any duty to ascertain or inquire into (i) any statement, warranty or representation made in or in connection with this Agreement, (ii) the contents of any certificate, report or other document delivered hereunder or in connection herewith, (iii) the performance or observance of any of the covenants, agreements or other terms or conditions set forth in any Loan Document, (iv) the validity, enforceability, effectiveness or genuineness of this Agreement, any other Loan Document, or any other agreement, instrument or document, (v) the satisfaction of any condition set forth in this Agreement, other than to confirm receipt of items expressly required to be delivered to Administrative Agent, (vi) the value, sufficiency, creation, perfection or priority of any lien on the Mortgaged Property, or (vii) the financial condition of Borrower, Guarantor

10.04

Reliance. Administrative Agent shall be entitled to rely upon, and shall not incur any liability for relying upon, any notice, request, certificate, consent, statement, instrument, document or other writing (including electronic communications) believed by it to be genuine and to have been signed or sent by the proper Person. Administrative Agent also may rely upon any statement made to it orally or by telephone and believed by it to be made by the proper Person, and shall not incur any liability for relying thereon. Administrative Agent may consult with legal counsel (who may be counsel for Borrower), independent accountants and other experts selected by it, and shall not be liable for any action taken or not taken by it in accordance with the advice of any such counsel, accountants or experts.

10.05

Sub-Agents

Administrative Agent may perform any and all its duties and exercise its rights and powers by or through any one or more sub-agents appointed by Administrative Agent.

Administrative Agent and any such sub-agent may perform any and all its duties and exercise its rights and powers through their respective Related Parties. The exculpatory provisions of the preceding paragraphs shall apply to any such sub-agent and to the Related Parties of the Administrative Agent and any such sub-agent, and shall apply to their respective activities in connection with the syndication of the credit facilities provided for herein as well as activities as Administrative Agent.

Resignation

10.06

Subject to the appointment and acceptance of a successor Administrative Agent as provided in this paragraph, Administrative Agent may resign at any time by notifying the Lenders and Borrower. Upon any such resignation, the Required Lenders shall have the right, in consultation with Borrower, to appoint a successor. If no successor shall have been so appointed by the Required Lenders and shall have accepted such appointment within 30 days after the retiring Administrative Agent mere and administrative Agent may, on behalf of the Lenders, appoint a successor Administrative Agent which shall be a bank with an office in New York, New York, or an Affiliate of any such bank. Upon the acceptance of its appointment as Administrative Agent hereunder by a successor, such successor shall succeed to and become vested with all the rights, powers, privileges and duties of the retiring Administrative Agent, and the retiring Administrative Agent shall be discharged from its duties and obligations hereunder. The fees payable by Borrower to a successor Administrative Agent shall be the same as those payable to its predecessor unless otherwise agreed between Borrower and such successor. After Administrative Agent, its sub-agents and their respective Related Parties in respect of any actions taken or omitted to be taken by any of them while it was acting as Administrative Agent.

10.07 Independent Credit Analysis.

Each Lender acknowledges and agrees that the extensions of credit made hereunder are commercial loans and letters of credit and not investments in a business enterprise or securities. Each Lender further represents that it is engaged in making, acquiring or holding commercial loans in the ordinary course of its business and has, independently and without reliance upon Administrative Agent or any other Lender and based on such documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement as a Lender, and to make, acquire or hold Loans hereunder. Each Lender shall, independently and without reliance upon Administrative Agent or any other Lender and based on such documents and information (which may contain material, non-public information within the meaning of the United States securities laws concerning Borrower and its Affiliates) as it shall from time to time deem appropriate, continue to make its own decisions in taking or not taking action under or based upon this Agreement, any related agreement or any document furnished hereunder or thereunder and in deciding whether or to the extent to which it will continue as a lender or assign or otherwise transfer its rights, interests and obligations hereunder.

10.08 Lender Actions Against Collateral.

Each Lender agrees that it will not take any action, nor institute any actions or proceedings, with respect to the Obligations, against Borrower, Guarantor, or any other obligor under this Agreement or the other Loan Documents or against any of the Mortgaged Property (including, without limitation, set-off rights) without the consent of the Required Lenders. With respect to any action by Administrative Agent to enforce the rights and remedies of Administrative Agent and the Lenders under this Agreement and the Other Loan Documents, each Lender hereby consents to the jurisdiction of the court in which such action is maintained, and agrees to deliver its Note to Administrative Agent to the extent necessary to enforce the rights and remedies of Administrative Agent for the benefit of the Lenders under the Deed of Trust in accordance with the provisions hereof. Each Lender agrees to indemnify each of the other Lenders for any loss or damage suffered or cost incurred by such other Lender (including without limitation, attorneys fees and expenses and other costs of defense) as a result of the breach of this Section 10.08 by such Lender.

Lender Reply Period.

All communications from Administrative Agent to Lenders requesting Lenders' determination, consent or approval (i) shall be given in the form of a written notice to each Lender, (ii) shall be accompanied by a description of the matter as to which such determination, consent or approval is requested, (iii) shall include a legend substantially as follows, printed in capital letters or boldface type:

"THIS COMMUNICATION REQUIRES IMMEDIATE RESPONSE. FAILURE TO RESPOND WITHIN TEN (10) BUSINESS DAYS AFTER THE DELIVERY OF THIS COMMUNICATION

SHALL CONSTITUTE A DEEMED APPROVAL BY THE ADDRESSEE OF THE MATTER DESCRIBED ABOVE."

and (iv) shall include Administrative Agent's recommended course of action or determination in respect thereof. Each Lender shall reply promptly to any such request, but in any event within ten (10) Business Days after the delivery of such request by Administrative Agent (the "Lender Reply Period"). Unless a Lender shall give written notice to Administrative Agent that it objects to the recommendation or determination of Administrative Agent (together with a written explanation of the reasons behind such objection) within the Lender Reply Period, such Lender shall be deemed to have approved of or consented to such recommendation or determination. With respect to decisions requiring the approval of the Required Lenders or all Lenders, Administrative Agent shall timely submit any required written notices to all Lenders and upon receiving the required approval or consent shall follow the course of action or determination recommended by Administrative Agent or such other course of action recommended by the Required Lenders or all of the Lenders, as the case may be, and each non-responding Lender shall be deemed to have concurred with such recommended course of action.

10.10

In the event that all or any portion of the Mortgaged Property is acquired by Administrative Agent as the result of a foreclosure or acceptance of a deed or assignment in lieu

of foreclosure, or is retained in satisfaction of all or any part of the Obligations, title to any such Mortgaged Property or any portion thereof shall be held in the name of Administrative Agent or a nominee or subsidiary of Administrative Agent, as agent, for the benefit of the Lenders, or in an entity co-owned by the Lenders as determined by the Administrative Agent Administrative Agent shall prepare a recommended course of action for such Mortgaged Property (the "Post-Foreclosure Plan") and submit it to the Lenders for approval by the Required Lenders. In the event that Administrative Agent and all submit any and all such additional Post-Foreclosure Plan(s) to the Lenders for evaluation and the approval by the Required Lenders. In accordance with the approved Post-Foreclosure Plan, Administrative Agent shall manage, operate, repair, administer, complete, construct, restore or otherwise deal with the Mortgaged Property acquired and administer all transactions relating thereto, including, without limitation, employing a management agent agent, leasing agent and other agents, contractors and employees, including agents for the sale of such Mortgaged Property and paying the expenses of such Mortgaged Property. Upon demand therefor from time to time, each Lender will contribute its ratable share (based on their respective Commitments immediately prior to the termination thereof) of all reasonable costs and expenses incurred by Administrative Agent pursuant to the Post-Foreclosure Plan in connection with the construction, operation, management, maintenance, leasing and sale of the Mortgaged Property. In addition, Administrative Agent shall render or cause to be rendered by the managing agent, to each of the Lenders, monthly, an income and expenses statement for such Mortgaged Property, and each of the Lenders shall promptly contribute its ratable share (based on their respective Commitments immediately prior to the termination thereof) of any operating loss for the Mortgaged Property, and such other expenses and

10.11 Defaulting Lender.

Notwithstanding any provision of this Agreement to the contrary, if a Lender becomes a Defaulting Lender, the following provisions shall apply for so long as such Lender is a Defaulting Lender;

(a) Suspension of Voting Rights. Such Defaulting Lender shall not have the right to vote on any issue on which voting is required (other than to the extent expressly

provided in Section 11.02(b)) and the Commitment of such Defaulting Lender shall not be included in determining whether the Required Lenders have taken or may take any action hereunder.

(b) Turn Over of Payments. All amounts payable hereunder to the Defaulting Lender in respect of the Obligations (whether on account of principal, interest, fees or

otherwise, including, without limitation, interest payments from interest reserve allocations to the Defaulting Lender and any amounts that would otherwise be payable to the Defaulting Lender pursuant to Section 3.12, but excluding Section 3.13(b)), shall be paid to Administrative Agent, retained in a segregated account and, subject to any applicable requirements of law, be applied at such time or times as may be determined by Administrative Agent as follows: (i) first, to the payment of any amounts owing by the Defaulting Lender to Administrative Agent hereunder, (ii) second, to the funding of any Advance in respect of which the Defaulting Lender has failed to fund its portion as required by this Agreement, as determined by Administrative Agent, (iii) third, to the payment of any amounts owing by the Defaulting Lender to the Non-Defaulting Lenders hereunder, including without limitation for any Special Advance under paragraph (c) of this Section 10.11, (iv) fourth, if so determined by Administrative Agent and Borrower, held in such account as cash collateral for future funding obligations of the Defaulting Lender under this Agreement, and (v) fifth, to the Defaulting Lender or as otherwise directed by a court of competent jurisdiction; provided that if such payment is a prepayment of the principal amount of any Loans, such payment shall be applied solely to repay the Loans of all Non-Defaulting Lenders pro rata prior to being applied to the prepayment of any Loans owed to the Defaulting Lender.

(c) Special Advances. If a Lender fails to fund its portion of any Advance, in whole or part, within three (3) Business Days after the date required hereunder and Administrative Agent shall not have funded the Defaulting Lender's portion of the Advance under Section 3.02(b), Administrative Agent shall so notify the Lenders, and within [three (3)] Business Days after delivery of such notice, the Non-Defaulting Lender is shall have the right, but not the obligation, in their respective, sole and absolute discretion, to fund all or a portion of such deficiency (the amount so funded by any such Non-Defaulting Lenders being referred to herein as a "Special Advance") to Borrower. In such event, the Defaulting Lender and Borrower severally agree to pay to Administrative Agent for payment to the Non-Defaulting Lenders making the Special Advance, forthwith on demand such amount with interest thereon, for each day from and including the date such amount is made available to Borrower to but excluding the date of payment to Administrative Agent, at (i) in the case of the Defaulting Lender, the greater of the Federal Funds Effective Rate and a rate determined by Administrative Agent in accordance with banking industry rules on interbank compensation or (ii) in the case of Borrower, the interest rate applicable to Floating Rate Loans.

(d) Option to Purchase Future Commitment. The Non-Defaulting Lenders shall have the right, but not the obligation, in their respective, sole and absolute discretion, to acquire for no cash consideration (pro rata, based on the respective Commitments of those Lenders electing to exercise such right), Defaulting Lender's Commitment to fund future Loans (the "Future Commitment"). Upon any such purchase of the Defaulting Lender's Future Commitment, the Defaulting Lender's share in future Advances and its rights under the Loan Documents with respect thereto shall terminate on the date of purchase, and the Defaulting Lender shall promptly execute all documents reasonably requested to surrender and transfer such interest.

(e) <u>Replacement of Defaulting Lender</u>.

(i) By Required Lenders may, upon notice to the Defaulting Lender and Administrative Agent, require the Defaulting Lender to assign and delegate, without recourse (in accordance with and subject to the restrictions contained in Section 11.04 hereof) all its interests, rights and obligations under this Agreement to an assignee that shall assume such obligations (which assignee may be another Lender, if a Lender accepts such assignment); provided that (i) the Defaulting Lender shall have received payment of an amount equal to the outstanding principal of its Loans, accrued interest thereon, accrued fees and all other amounts payable to it hereunder from the assignee (to the extent of such outstanding principal and accrued interest and fees) or Borrower (in the case of all other amounts); and (ii) Administrative Agent shall have received payment of any amounts owing by such Lender to the Administrative Agent or the other Lenders under this Agreement. The Defaulting Lender shall not be required to make any such assignment and delegation if, prior thereto, such Lender shall cease to be a Defaulting Lender.

(ii) By Borrower. If the Lender has become a Defaulting Lender due to a failure to fund its Loans hereunder, Borrower may at its option replace Defaulting Lender under Section 3.13(b).

(f) Indemnification. Each Defaulting Lender shall indemnify Administrative Agent, each Non-Defaulting Lender and Borrower from and against any liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind and nature whatever which may be imposed on, incurred by or asserted against Administrative Agent, any Non-Defaulting Lender or Borrower with respect to the Loan Documents in any way relating to or arising out of such Lender's status as a Defaulting Lender. The obligations of the Defaulting Lender under this clause (f) shall survive the payment of the Obligations, the termination of this Agreement and the Defaulting Lender's reversion to a Non-Defaulting Lender under Paragraph (g) of this Section 10.11.

(g) <u>Ceasing to be a Defaulting Lender.</u> A Lender shall cease to be Defaulting Lender only upon (i) the payment of all amounts due and payable by Defaulting Lender to Administrative Agent or any other Lender under this Agreement; (ii) the payment of any damages suffered by Borrower as a result of such Defaulting Lender's default hereunder; (iii) the confirmation by the Lender to Administrative Agent and Borrower in writing that the Lender will comply with all of its funding obligations under this Agreement; and (iii) the

circumstances described in clause (d) of the definition of "Defaulting Lender" do not exist. An assignment by a Lender of its rights and obligations under this Agreement shall not in and of itself cause the Lender to cease to be a Defaulting Lender.

10.12 Compliance with Flood Laws.

Administrative Agent has adopted internal policies and procedures that address requirements placed on federally regulated lenders under the Flood Laws and will post on the applicable electronic platform (or otherwise distribute to each Lender documents that it receives in connection with the Flood Laws ("Flood Documents"); provided, however that Administrative Agent makes no representation or warranty with respect to the adequacy of the Flood Documents or their compliance with the Flood Laws. Each Lender acknowledges and agrees that it is individually responsible for its own compliance with the Flood Laws and that it shall, independently and without reliance upon Administrative Agent or any other Lender and based on such documents and information as it shall from time to time deem appropriate, including the Flood Documents posted or distributed by Administrative Agent, continue to do its own due diligence to ensure its compliance with the Flood Laws.

10.13 Borrower's Rights.

The provisions of this Article X are solely for the benefit of Administrative Agent and the Lenders, and Borrower shall not have any rights to rely on, enforce or consent to any waiver, modification or amendment of, any of the provisions hereof; provided, however, that Borrower (a) acknowledges and agrees to the limitations set forth in Section 11.02(c) hereof on Administrative Agent's ability to act unilaterally with respect to this Agreement and the other Loan Documents, and (b) agrees that Administrative Agent's inability to deliver any consent to, or approval of, an action requested by Borrower due lack of appropriate Lender consent in accordance with the provisions of Section 11.02(c) hereof shall not constitute an unreasonable withholding or delay by Administrative Agent in the giving of such consent or approval. Notwithstanding the foregoing, Borrower shall be entitled to rely on consents and approvals executed by Administrative Agent without investigation as to the existence of proper Lender authorization.

ARTICLE XI MISCELLANEOUS

<u>11.01</u>	Notices.	
	(a)	Generally. Except in the case of notices and other communications expressly permitted to be given by electronic communication (and subject to paragraph (b) below),
all notices and other communi	cations provided for he	rein shall be in writing and shall be delivered by hand or overnight courier service, mailed by certified or registered mail or sent by telecopy, as follows:
	(i) if to Borrower, to it at c/o Rexford Industrial, 11620 Wilshire Boulevard, Suite 300, Los Angeles, California 90025, Attention Adeel Khan
(Telecopy N	No. (310)), (Email Address: akhan@rexfordindustrial.com); with a copy to Gibson, Dunn & Crutcher LLP, 333 South Grand Avenue, Los Angeles, California

90071, Attention: Dennis B. Arnold, Esq. (Telecopy No. (213) 229-6864), (Email Address: darnold@gibsondunn.com);
(ii) if to Administrative Agent, to it at JPMorgan Chase Bank, N.A., 2029 Century Park East, 38th Floor, Los Angeles, California 90067, Attention

Faina Birger (Email Address: faina.birger@jpmorgan.com); and

(iii) if to any other Lender, to it at its address (or telecopy number or email address) set forth in its Administrative Questionnaire.

Notices sent by hand or overnight courier service, or mailed by certified or registered mail, shall be deemed to have been given when received; notices sent by facsimile shall be deemed to have been given when sent (except that, if not

Notices sent by hand or overnight courier service, or mailed by certified or registered mail, shall be deemed to have been given when received; notices sent by facsimile shall be deemed to have been given when sent (except that, if not given during normal business hours for the recipient, shall be deemed to have been given at the opening of business on the next business day for the recipient). Notices delivered through Electronic Systems, to the extent provided in paragraph (b) below, shall be effective as provided in said paragraph (b).

(b) Electronic Notices. Notices and other communications to the Lenders hereunder may be delivered or furnished by using Electronic Systems pursuant to procedures approved by the Administrative Agent; provided that the foregoing shall not apply to notices for interest rate elections and advances unless otherwise agreed by the Administrative Agent and the applicable Lender. The Administrative Agent are the Borrower may, in its discretion, agree to accept notices and other communications to it hereunder by electronic communications pursuant to procedures approved by it; provided that approval of such procedures may be limited to particular notices or communications.

Unless the Administrative Agent otherwise prescribes, (i) notices and other communications sent to an e-mail address shall be deemed received upon the sender's receipt of an acknowledgement from the intended recipient (such as by the "return receipt requested" function, as available, return e-mail or other written acknowledgement), and (ii) notices or communications posted to an Internet or intranet website shall be deemed received upon the deemed receipt by the intended recipient, at its e-mail address as described in the foregoing clause (i), of notification that such notice or communication is available and identifying the website address therefor; provided that, for both clauses (i) and (ii) above, if such notice, email or other communication is not sent during the normal business hours of the recipient, such notice or communication shall be deemed to have been sent at the opening of business on the next business day for the recipient.

(c) Changes in Address. Any party hereto may change its address or telecopy number or email address for notices and other communications hereunder by notice to the other parties hereto. All notices and other communications given to any party hereto in accordance with the provisions of this Agreement shall be deemed to have been given on the date of receipt.

(d) <u>Electronic Systems.</u>

(i) The Borrower agrees that the Administrative Agent may, but shall not be obligated to, make Communications (as defined below) available to the

other

 $Lenders\ by\ posting\ the\ Communications\ on\ Debt\ Domain,\ Intralinks,\ Syndtrak,\ Clear Par\ or\ a\ substantially\ similar\ Electronic\ System.$

(ii) Any Electronic System used by the Administrative Agent is provided "as is" and "as available." The Administrative Agent does not warrant the adequacy of such Electronic Systems and expressly disclaim liability for errors or omissions in the Communications. No warranty of any kind, express, implied or statutory, including, without limitation, any warranty of merchantability, fitness for a particular purpose, non-infringement of third-party rights or freedom from viruses or other code defects, is made by the Administrative Agent in connection with the Communications or any Electronic System. In no event shall the Administrative Agent have any liability to the Borrower, any Lender, or any other Person or entity for damages of any kind, including, without limitation, direct or indirect, special, incidental or consequential damages, losses or expenses (whether in tort, contract or otherwise) arising out of the Borrower's or the Administrative Agent's transmission of communications through an Electronic System. "Communications" means, collectively, any notice, demand, communication, information, document or other material provided by or on behalf of the Borrower pursuant to any Loan Document or the transactions contemplated therein which is distributed by the Administrative Agent or any Lender by means of electronic communications pursuant to this Section, including through an Electronic System.

Waivers; Amendments, Amendm

(a) No Deemed Waivers; Remedies Cumulative. No failure or delay by Administrative Agent or any Lender in exercising any right or power hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of any such right or power, or any abandonment or discontinuance of steps to enforce such a right or power, preclude any other or further exercise thereof or the exercise of any other right or power. The rights and remedies of Administrative Agent and the Lenders hereunder are cumulative and are not exclusive of any rights or remedies that they would otherwise have. No waiver of any provision of this Agreement or consent to any departure by Borrower therefrom shall in any event be effective unless the same shall be permitted by Section 11.02(b) hereof, and then such waiver or consent shall be effective only in the specific instance and for the purpose for which given. Without limiting the generality of the foregoing, the making of a Loan shall not be construed as a waiver of any Default or Unmatured Default, regardless of whether Administrative Agent or any Lender may have had notice or knowledge of such Default or Unmatured Default at the time.

(b) Waivers and Amendments. No provision of this Agreement or any other Loan Agreement may be waived, amended or modified except pursuant to an agreement or agreements in writing entered into by Borrower and the Required Lenders or by Borrower and the Administrative Agent with the consent of the Required Lenders; provided that no such agreement shall (i) increase the Commitment of any Lender without the written consent of such Lender (including any such Lender that is a Defaulting Lender), (iii) reduce the principal amount of any Loan or reduce the rate of interest thereon, or reduce any fees payable hereunder, without the written consent of each Lender affected thereby (including any such Lender that is a Defaulting Lender), (iii) postpone the scheduled date of payment of the principal amount of any

Loan or any interest thereon, or any fees payable hereunder, or reduce the amount of, waive or excuse any such payment, without the written consent of each Lender (including any such Lender that is a Defaulting Lender), (v) change Section 3.12(b) or (c) in a manner that would alter the pro rata sharing of payments required thereby, without the written consent of each Lender (including any such Lender that is a Defaulting Lender), (v) change any of the provisions of this Section or the definition of "Required Lenders" or any other provision hereof specifying the number or percentage of Lenders required to waive, amend or modify any rights hereunder or make any determination or grant any consent hereunder, without the written consent of each Lender (including any such Lender that is a Defaulting Lender) directly affected thereby, (vi) release Guarantor from any of its obligations under the Loan Documents or release all or substantially all of the Mortgaged Property from the lien of the Loan Documents, without the written consent of each Lender (other than any Defaulting Lender), or (vii) permit an assignment by Borrower of any rights or obligations under the Loan Documents, without the written consent of Administrative Agent hereunder without the prior written consent of Administrative Agent.

(c) Actions by Administrative Agent; Required Consents. Each Lender authorizes Administrative Agent to enter into the Loan Documents (other than this Agreement) on behalf of, and for the benefit of, the Lenders and to take all actions left to the discretion of the Administrative Agent herein and therein on behalf of, and for the benefit of, the Lenders. Each Lender agrees that any action taken by Administrative Agent at the direction of the Required Lenders (or such other number or percentage of the Lenders as shall be necessary under the circumstances as provided in this Agreement), and any action taken by Administrative Agent on trequiring consent by the Required Lenders (or such other number or percentage of the Lenders as shall be necessary under the circumstances as provided in this Agreement) shall be authorized by and binding upon all Lenders. Notwithstanding the foregoing provisions of this Section 11.02(c), Administrative Agent shall not have the authority to bind the Lenders with respect to any of the following matters without Administrative Agent obtaining the prior written consent of the Required Lenders:

sent of the Required Lenders:

(i) the determination as to whether an Appraisal will be required pursuant to Section 3.14(a)(iv);
(ii) the determination as to whether any Major Lease which is less beneficial to Borrower as Landlord than as set forth on Exhibit E hereto is an Approved Lease;

(iii) the approval of any modification or amendment of a Major Lease which would cause such Major Lease to no longer constitute an Approved Lease;

Lease;

(iv) the approval of any action described in Section 4.04(a)(iii) hereof;
(v) the exercise of any rights and remedies against Borrower, Guarantor or the Mortgaged Property; provided, however, that Administrative Agent may, in its discretion but without obligation, in the absence of direction from the Required Lenders, take such interim action as it believes necessary to preserve the rights

of the Lenders hereunder and in and to the Mortgaged Property, including, without limitation, (i) the delivery of default notices to Borrower or any other Person, (ii) petitioning a court for injunctive relief, the appointment of a receiver or preservation of the proceeds of any collateral, (iii) the making of Advances for the payment of interest, and (iv) the exercise of the cure rights of Administrative Agent under this Agreement and the other

(vi) the expenditure of funds by Administrative Agent for which the Lenders are responsible under Section 11.03 hereof to cure Defaults not in excess of Five Hundred Thousand Dollars (\$500,000) in the aggregate; provided, however, that expenditures to pay any of the following items shall not be subject to the foregoing limitation in amount: (i) real estate taxes, assessments, charges and levies owing with respect to the Mortgaged Property, (ii) insurance premiums owing with respect to insurance coverage required by the Loan Documents, (iii) claims of lienholders with priority over the lien of the Deed of Trust, (iv) expenditures necessary to respond to emergency conditions with respect to the Mortgaged Property, and (v) expenditures necessary to preserve the validity and priority of the lien of the Deed of Trust.

11.03

Expenses; Indemnity; Damage Waiver.

Expenses. Borrower shall pay (i) all reasonable out-of-pocket expenses incurred by Administrative Agent and its Affiliates, including appraisal fees, inspection fees, title and escrow charges, the cost of Intralinks or a similar electronic workspace, and the reasonable fees, charges and disbursements of counsel for Administrative Agent, in connection with the closing of the Loan and the syndication of the credit facilities provided for herein, the preparation and administration of this Agreement and the other Loan Documents or any extensions, amendments, modifications or waivers of the provisions hereof or thereof (whether or not the transactions contemplated hereby or thereby shall be consummated), and (ii) all out-of-pocket expenses incurred by Administrative Agent or any Lender, including the fees, charges and disbursements of any counsel for Administrative Agent or any Lender, in connection with the enforcement or protection of its rights in connection with the Loans made hereunder, including its rights under this Section, or in connection with the Loans made hereunder, including all such out-of-pocket expenses incurred during any workout, restructuring or negotiations in respect of such Loans.

(b) Borrower Indemnity. Borrower shall indemnify Administrative Agent and each Lender, and each Related Party of any of the foregoing Persons (each such Person being

called an "Indemnitee") against, and hold each Indemnitee harmless from, any and all losses, claims, damages, judgments, liabilities and related expenses, including the fees, charges and disbursements of any counsel for any Indemnitee, but excluding indirect, consequential, special or punitive damages (collectively, "Losses"), actually incurred by or asserted against any Indemnitee arising out of, in connection with, or as a result of (i) the execution or delivery of this Agreement or any agreement or instrument contemplated hereby, the performance by the parties hereto of their respective obligations hereunder or the consummation of the Transactions or any other transactions contemplated hereby, (ii) the Loan or the use of the proceeds therefrom, or (iv) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing, whether based on contract, tort or any other theory and regardless of whether any Indemnitee is a party thereto; provided that such indemnity shall not, as to any Indemnitee, be available to the extent that such Losses are determined by a court of competent jurisdiction by final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Indemnitee. The foregoing indemnity set forth in this Section 11.03(b) shall not apply (i) with respect to Taxes other than any Taxes that represent losses, claims or damages arising from any non-Tax claim and (ii) to any Losses which are the subject of the Environmental Indemnity Agreement, it being the intention of the parties hereto that Borrower's liability for environmental matters be governed exclusively by the Environmental Indemnity Agreement and not by this Agreement

Reimbursement by Lenders. To the extent that Borrower fails to pay any amount required to be paid by it to Administrative Agent under Section 11.03(a) or (b) hereof, each Lender severally agrees to pay to Administrative Agent such Lender's Applicable Percentage (determined as of the time that the applicable unreimbursed expense or indemnity payment is sought) of such unpaid amount; provided that the unreimbursed expense or indemnified loss, claim, damage, liability or related expense, as the case may be, was incurred by or asserted against Administrative Agent in its capacity as such.

Payment of Amounts Due. All amounts due under this Section shall be payable not later than ten (10) days after written demand therefor. 11.04 Successors and Assigns.

(a) Binding Effect. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, except that (i) Borrower may not assign or otherwise transfer any of its rights or obligations hereunder without the prior written consent of each Lender (and any attempted assignment or transfer by Borrower without such consent shall be null and void) and (ii) no Lender may assign or otherwise transfer its rights or obligations hereunder except in accordance with this Section. Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby, Participants (to the extent provided in Section 11.04(c) hereof) and, to the extent expressly contemplated hereby, the Related Parties of each of Administrative Agent and the Lenders) any legal or equitable right, remedy or claim under or by reason of this Agreement.

(b)

Assignments by Lenders.

Subject to the conditions set forth in Section 11.04(b)(ii) below, any Lender may assign to one or more Persons (other than an Ineligible Institution) (i) all or a portion of its rights and obligations under this Agreement (including all or a portion of its Commitment and the Loans at the time owing to it) with the prior written consent (such consent not to be unreasonably withheld) of:

Borrower, provided that Borrower shall be deemed to have consented to any such assignment unless it shall object thereto by written notice to the Administrative Agent within five (5) Business Days after having received notice thereof, and provided further that no consent of Borrower shall be required

for an assignment to a Lender, an Affiliate of a Lender, an Approved Fund or, if a Default has occurred and is continuing, any other assignee; and
(B) Administrative Agent, provided that no consent of Administrative Agent shall be required for an assignment of any Commitment to an assignee that is a Lender with a Commitment immediately prior to giving effect to such assignment.

(ii) Assignments shall be subject to the following additional conditions:

except in the case of an assignment to a Lender or an Affiliate of a Lender or an assignment of the entire remaining amount of the assigning Lender's Commitment or Loans, the amount of the Commitment or Loans of the assigning Lender subject to each such assignment determined as of the date the Assignment and Assumption with respect to such assignment is delivered to Administrative Agent) shall not be less than \$5,000,000 unless each of Borrower and Administrative Agent otherwise consent, provided that no such consent of Borrower shall be required if a Default has occurred and is continuing;

each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and (B)

the parties to each assignment shall execute and deliver to Administrative Agent an Assignment and Assumption, together

obligations under this Agreement (C)

with a processing and recordation fee of \$3,500; (D) the assignee, if it shall not be a Lender, shall deliver to the Administrative Agent an Administrative Questionnaire in which the assignee designates one or more Credit Contacts to whom all syndicate-level information (which may contain material non-public information about Borrower, the Guarantor, and their

Related Parties or their respective securities) will be made available and who may receive such information in accordance with the assignee's compliance procedures and applicable laws, including Federal and state securities laws;

For the purposes of this Section 11.04(b), the terms "Approved Fund" and "Ineligible Institution" have the following meanings:

"Approved Fund" means any Person (other than a natural person) that is engaged in making, purchasing, holding or investing in bank loans and similar extensions of credit in the ordinary course of its business and that is administered or managed by (a) a Lender, (b) an Affiliate of a Lender or (c) an entity or an Affiliate of an entity that administers or manages a Lender.

"Ineligible Institution" means (a) a natural person, (b) a Defaulting Lender, (c) the Borrower, Guarantor, or any of their respecting Affiliates, or (d) a holding company,

investment vehicle or trust for, or owned and operated for the primary benefit of, a natural person or relative(s) thereof.

Subject to acceptance and recording thereof pursuant to Section 11.04(b)(iv), from and after the effective date specified in each Assignment and Assumption the assignee thereunder shall be a party hereto and, to the extent of the interest assigned by such Assignment and Assumption, have the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder shall, to the extent of the interest assigned by such Assignment and Assumption, be released from its obligations under this Agreement (and, in the case of an Assignment and Assumption covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto but shall continue to be entitled to the benefits of Sections 3.09, 3.10, 3.11, 4.13(c) and 11.03 hereof). Any assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this Section 11.04 shall be treated for purposes of this Agreement as a sale by such Lender of a participation in such rights and obligations in accordance with Section 11.04(c) hereof.

(iv) Administrative Agent, acting for this purpose as an agent of Borrower, shall maintain at one of its offices a copy of each Assignment and Assumption delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Commitment of, and principal amount (and stated interest) of the Loans owing to, each Lender pursuant to the terms hereof from time to time (the "Register"). The entries in the Register shall be conclusive, and Borrower, the Administrative Agent and the Lenders may treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary.

(v) Upon its receipt of a duly completed Assignment and Assumption executed by an assigning Lender and an assignee, the assignee's completed Administrative Questionnaire (unless the assignee shall already be a Lender hereunder), the processing and recordation fee referred to in Section 11.04(b) hereof and any written consent to such assignment required by Section 11.04(b) hereof, Administrative Agent shall accept such Assignment and Assumption and record the information contained therein in the Register; provided that if either the assigning Lender or the assignee shall have failed to make any payment required to be made by it pursuant to Section 3.02(b), 3.12(d) or 11.03(c), Administrative Agent shall have no obligation to accept such Assignment and Assumption and record the information therein in the Register unless and until such payment shall have been made in full, together with all accrued interest thereon. No assignment shall be effective for purposes of this Agreement unless it has been recorded in the Register as provided in this Section.

Participations.

Any Lender may, without the consent of Borrower or Administrative Agent, sell participations to one or more banks or other entities (a "Participant") in all or a portion of such Lender's rights and obligations under this Agreement (including all or a portion of its Commitment and the Loans owing to it);

provided that (A) such Lender's obligations under this Agreement shall remain unchanged, (B) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations and (C) Borrower, Administrative Agent and the other Lenders shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement. Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender will not, without the consent of the Participant, agree to any amendment, modification or waiver described in the first proviso to Section 11.02(b) that affects such Participant. Borrower agrees that each Participant shall be entitled to the benefits of Sections 3.09, 3.10 and 3.11 (subject to the requirements and limitations therein, including the requirements under Section 3.11(f) (it being understood that the documentation required under Section 3.11(f) shall be delivered to the participating Lender) to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to Section 11.04(b) hereof; provided that such Participant (A) agrees to be subject to the provisions of Section 3.13 as if it were an assignee under paragraph (b) of this Section; and (B) shall not be entitled to receive any greater payment under Section 3.09 or 3.11, with respect to any participation, than its participating Lender would have been entitled to receive, except to the extent such entitlement to receive a greater payment results from a Change in Law that occurs after the Participant acquired the applicable participation. Each Lender that sells a participation agrees, at Borrower's request and expense, to use reasonable efforts to cooperate with Borrower to effectuate the provisions of Section 3.12(c) with respect to any Participant. To the extent permitted by law, each Participant also shall be entitled to the benefits of Section 11.08 hereof as though it were a Lender; provided that such Participa

79

(d) <u>Pledges by Lenders.</u> Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement to secure obligations of such Lender, including without limitation any pledge or assignment to secure obligations to a Federal Reserve Bank, and this Section shall not apply to any such pledge or assignment of a security interest; <u>provided</u> that no such pledge or assignment of a security interest shall release a Lender from any of its obligations hereunder or substitute any such pledgee or assignee for such Lender as a party hereto.

<u>11.05</u> Survival.

All covenants, agreements, representations and warranties made by Borrower herein and in the certificates or other instruments delivered in connection with or pursuant to this Agreement shall be considered to have been relied upon by the other parties hereto and shall survive the execution and delivery of this Agreement and the making of any Loans, regardless of any investigation made by any such other party or on its behalf and notwithstanding that Administrative Agent or any Lender may have had notice or knowledge of any Default or Umatured Default or incorrect representation or warranty at the time any credit is extended hereunder, and shall continue in full force and effect as long as the principal of or any accrued interest on any Loan or any fee or any other amount payable under this Agreement is outstanding and so long as the Commitments have not expired or terminated. The provisions of Sections 3.09, 3.10, 3.11, 4.13(c) and 11.03 and ARTICLE X hereof shall survive and remain in full force and effect regardless of the consummation of the transactions contemplated hereby, the repayment of the Loans, the expiration or termination the Commitments or the termination of this Agreement or any provision hereof.

11.06 Counterparts; Integration; Effectiveness.

(a) This Agreement may be executed in counterparts (and by different parties hereto on different counterparts), each of which shall constitute an original, but all of which when taken together shall constitute a single contract. This Agreement and the other Loan Documents and any separate letter agreements with respect to fees paid to the Administrative Agent constitute the entire contract among the parties relating to the subject matter hereof and supersede any and all previous agreements and understandings, oral or written, relating to the subject matter hereof. This Agreement shall become effective when it shall have been executed by Administrative Agent and when Administrative Agent shall have received counterparts hereof which, when taken together, bear the signatures of each of the other parties hereto, and thereafter shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns.

(b) Delivery of an executed counterpart of a signature page of this Agreement by telecopy, emailed pdf. or any other electronic means that reproduces an image of the actual executed signature page shall be effective as delivery of a manually executed counterpart of this Agreement. The words "execution," "signature," "delivery," and words of like import in or relating to any document to be signed in connection with this Agreement and the transactions contemplated hereby shall be deemed to include Electronic Signatures, deliveries or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature, physical delivery

thereof or the use of a paper-based recordkeeping system, as the case may be, to the extent and as provided for in any applicable law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws based on the Uniform Electronic Transactions Act.

11.07 Severability.

Any provision of this Agreement or any other Loan Document held to be invalid, illegal or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such invalidity, illegality or unenforceability without affecting the validity, legality and enforceability of the remaining provisions hereof or thereof; and the invalidity of a particular provision in a particular jurisdiction shall not invalidate such provision in any other jurisdiction.

11.08 Right of Setoff.

NO LENDER SHALL EXERCISE A RIGHT OF SETOFF, LENDER'S LIEN OR COUNTERCLAIM OR TAKE ANY COURT OR ADMINISTRATIVE ACTION OR INSTITUTE ANY PROCEEDING TO ENFORCE ANY PROVISION OF THIS AGREEMENT OR ANY LOAN DOCUMENT UNLESS IT IS TAKEN WITH THE CONSENT OF THE LENDERS REQUIRED BY SECTION 11.02 OF THIS AGREEMENT. THIS PARAGRAPH SHALL BE SOLELY FOR THE BENEFIT OF EACH OF THE LENDERS.

11.09 Governing Law; Jurisdiction; Consent to Service of Process.

(a) Governing Law. This Agreement shall be construed in accordance with and governed by the internal laws (and not the law of conflicts) of the State of California, but giving effect to federal laws applicable to national banks.

(b) Consent to Jurisdiction. Administrative Agent, each Lender and Borrower each hereby irrevocably and unconditionally submits, for itself and its property, to the nonexclusive jurisdiction of any United States Federal or California State court sitting in any County in the State of California in which any of the Property is located, and any appellate court from any thereof, in any action or proceeding arising out of or relating to this Agreement, or for recognition or enforcement of any judgment, and each of the parties hereto hereby irrevocably and unconditionally agrees that all claims in respect of any such action or proceeding may be heard and determined in such California State or, to the extent permitted by law, in such Federal court. Each of the parties hereto agrees that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law. Nothing in this Agreement shall affect any right that Administrative Agent or any Lender may otherwise have to bring any action or proceeding relating to this Agreement against Borrower or its properties in the courts of any jurisdiction.

(c) Waiver of Objection to Venue. Administrative Agent, each Lender and Borrower each hereby irrevocably and unconditionally waives, to the fullest extent it may legally and effectively do so, any objection which it may now or hereafter have to the laying of venue of any suit, action or proceeding arising out of or relating to this Agreement in any court referred to in Section 11.09(b) hereof. Each of the parties hereto hereby irrevocably waives, to the fullest

extent permitted by law, the defense of an inconvenient forum to the maintenance of such action or proceeding in any such court.

(d) Service of Process. Each party to this Agreement irrevocably consents to service of process in the manner provided for notices in Section 11.01. Nothing in this Agreement will affect the right of any party to this Agreement to serve process in any other manner permitted by law.

11.10 WAIVER OF JURY TRIAL

EACH PARTY HERETO HEREBY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY LEGAL PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY). EACH PARTY HERETO (A) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION.

IN THE EVENT ANY LEGAL PROCEEDING IS FILED IN A COURT OF THE STATE OF CALIFORNIA (THE "COURT") BY OR AGAINST ANY PARTY HERETO IN CONNECTION WITH ANY CONTROVERSY, DISPUTE OR CLAIM DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY) (EACH, A "CLAIM") AND THE WAIVER SET FORTH IN THE PRECEDING PARAGRAPH IS NOT ENFORCEABLE IN SUCH ACTION OR PROCEEDING, THE PARTIES HERETO AGREE AS FOLLOWS:

WITH THE EXCEPTION OF THE MATTERS SPECIFIED IN PARAGRAPH 2 BELOW, ANY CLAIM WILL BE DETERMINED BY A GENERAL REFERENCE PROCEEDING IN ACCORDANCE WITH THE PROVISIONS OF CALIFORNIA CODE OF CIVIL PROCEDURE SECTIONS 638 THROUGH 645.1. THE PARTIES INTEND THIS GENERAL REFERENCE AGREEMENT TO BE SPECIFICALLY ENFORCEABLE IN ACCORDANCE WITH CALIFORNIA CODE OF CIVIL PROCEDURE SECTION 638. EXCEPT AS OTHERWISE PROVIDED IN THE LOAN DOCUMENTS, VENUE FOR THE REFERENCE PROCEEDING

WILL BE IN THE STATE OR FEDERAL COURT IN THE COUNTY OR DISTRICT WHERE VENUE IS OTHERWISE APPROPRIATE UNDER APPLICABLE LAW.

2. THE FOLLOWING MATTERS SHALL NOT BE SUBJECT TO A GENERAL REFERENCE PROCEEDING: (A) NON-JUDICIAL FORECLOSURE OF ANY SECURITY INTERESTS IN REAL OR PERSONAL PROPERTY, (B) EXERCISE OF SELF-HELP REMEDIES (INCLUDING, WITHOUT LIMITATION, SET-OFF), (C)

APPOINTMENT OF A RECEIVER AND (D) TEMPORARY, PROVISIONAL OR ANCILLARY REMEDIES (INCLUDING, WITHOUT LIMITATION, WRITS OF ATTACHMENT, WRITS OF POSSESSION, TEMPORARY RESTRAINING ORDERS OR PRELIMINARY INJUNCTIONS). THIS AGREEMENT DOES NOT LIMIT THE RIGHT OF ANY PARTY TO EXERCISE OR OPPOSE ANY OF THE RIGHTS AND REMEDIES DESCRIBED IN CLAUSES (A) - (D) AND ANY SUCH EXERCISE OR OPPOSITION DOES NOT WAIVE THE RIGHT OF ANY PARTY TO A REFERENCE PROCEEDING PURSUANT TO THIS AGREEMENT.

3. UPON THE WRITTEN REQUEST OF ANY PARTY, THE PARTIES SHALL SELECT A SINGLE REFEREE, WHO SHALL BE A RETIRED JUDGE OR JUSTICE. IF THE PARTIES DO NOT AGREE UPON A REFEREE WITHIN TEN (10) DAYS OF SUCH WRITTEN REQUEST, THEN, ANY PARTY MAY REQUEST THE COURT TO APPOINT A REFEREE PURSUANT TO CALIFORNIA CODE OF CIVIL PROCEDURE SECTION 640(B).

4. ALL PROCEEDINGS AND HEARINGS CONDUCTED BEFORE THE REFEREE, EXCEPT FOR TRIAL, SHALL BE CONDUCTED WITHOUT A COURT REPORTER, EXCEPT WHEN ANY PARTY SO REQUESTS, A COURT REPORTER WILL BE USED AND THE REFEREE WILL BE PROVIDED A COURTESY COPY OF THE TRANSCRIPT. THE PARTY MAKING SUCH REQUEST SHALL HAVE THE OBLIGATION TO ARRANGE FOR AND PAY COSTS OF THE COURT REPORTER, PROVIDED THAT SUCH COSTS, ALONG WITH THE REFEREE'S FEES, SHALL ULTIMATELY BE BORNE BY THE PARTY WHO DOES NOT PREVAIL, AS DETERMINED BY THE REFEREE.

5. THE REFEREE MAY REQUIRE ONE OR MORE PREHEARING CONFERENCES. THE PARTIES HERETO SHALL BE ENTITLED TO DISCOVERY, AND THE REFEREE SHALL OVERSEE DISCOVERY IN ACCORDANCE WITH THE RULES OF DISCOVERY, AND MAY ENFORCE ALL DISCOVERY ORDERS IN THE SAME MANNER AS ANY TRIAL COURT JUDGE IN PROCEEDINGS AT LAW IN THE STATE OF CALIFORNIA. THE REFEREE SHALL APPLY THE RULES OF EVIDENCE APPLICABLE TO PROCEEDINGS AT LAW IN THE STATE OF CALIFORNIA AND SHALL DETERMINE ALL ISSUES IN ACCORDANCE WITH APPLICABLE STATE AND FEDERAL LAW. THE REFEREE SHALL BE EMPOWERED TO ENTER EQUITABLE AS WELL AS LEGAL RELIEF AND RULE ON ANY MOTION WHICH WOULD BE AUTHORIZED IN A TRIAL, INCLUDING, WITHOUT LIMITATION, MOTIONS FOR DEFAULT JUDGMENT OR SUMMARY JUDGMENT. THE REFEREE SHALL REPORT HIS DECISION, WHICH REPORT SHALL ALSO INCLUDE FINDINGS OF FACT AND CONCLUSIONS OF LAW.

6. THE PARTIES RECOGNIZE AND AGREE THAT ALL CLAIMS RESOLVED IN A GENERAL REFERENCE PROCEEDING PURSUANT HERETO WILL BE DECIDED BY A REFEREE AND NOT BY A JURY.

WAIVER OF SPECIAL DAMAGES.
TO THE EXTENT PERMITTED BY APPLICABLE LAW, BORROWER SHALL NOT ASSERT, AND HEREBY WAIVES, ANY CLAIM AGAINST ADMINISTRATIVE AGENT AND EACH LENDER, ON ANY THEORY OF LIABILITY, FOR SPECIAL, INDIRECT, CONSEQUENTIAL OR PUNITIVE DAMAGES (AS OPPOSED TO DIRECT OR ACTUAL DAMAGES) ARISING OUT OF, IN CONNECTION WITH, OR AS A RESULT OF, THIS AGREEMENT OR ANY AGREEMENT OR INSTRUMENT CONTEMPLATED HEREBY, THE TRANSACTIONS, THE LOANS OR THE USE OF THE PROCEEDS THEREOF.

Article and Section headings and the Table of Contents used herein are for convenience of reference only, are not part of this Agreement and shall not affect the construction of, or be taken into consideration in interpreting, this Agreement.

11.13 Confidentiality.

Each of Administrative Agent and the Lenders agrees to maintain the confidentiality of the Information (as defined below), except that Information may be disclosed (a) to its and its Affiliates' directors, officers, employees and agents, including accountants, legal counsel and other advisors (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Information and instructed to keep such Information confidential), (b) to the extent requested by any regulatory authority, (c) to the extent required by applicable laws or regulations or by any subpoena or similar legal process, (d) to any other party to this Agreement, (e) in connection with the exercise of any remedies hereunder or any suit, action or proceeding relating to the Obligations or the enforcement of rights under the Loan Documents or any Swap Agreement, (f) subject to an agreement containing provisions substantially the same as those of this Section, to (i) any assignee of or Participant in, or any prospective assignee of or Participant in, any of its rights or obligations under this Agreement or (ii) any actual or prospective counterparty (or its advisors) to any Swap Agreement relating to Borrower and its obligations, (g) with the consent of Borrower, (h) to holders of equity interests in Borrower, or (i) to the extent such Information (i) becomes publicly available other than as a result of a breach of this Section or (ii) becomes available to Administrative Agent or any Lender on a non-confidential basis from a source other than Borrower. For the purposes of this Section, "Information" means all information received from Borrower relating to Borrower or its business, other than any such information that is available to Administrative Agent or any Lender on a non-confidential basis prior to disclosure by Borrower; provided that, in the case of information received from Borrower after the date hereof, such information is clearly identified at the time of delivery as confidential. Any Person required to maintain the confidentiality of Information as provided in this Section shall be considered to have complied with its obligation to do so if such Person has exercised the same degree of care to maintain the confidentiality of such Information as such Person would accord

EACH LENDER ACKNOWLEDGES THAT INFORMATION (AS DEFINED IN THIS SECTION) FURNISHED TO IT PURSUANT TO THIS AGREEMENT MAY INCLUDE MATERIAL NON-PUBLIC INFORMATION CONCERNING BORROWER, THE GUARANTOR, AND THEIR RELATED PARTIES OR THEIR RESPECTIVE SECURITIES, AND CONFIRMS THAT IT HAS DEVELOPED COMPLIANCE PROCEDURES REGARDING THE USE OF MATERIAL NON-PUBLIC INFORMATION AND THAT IT WILL HANDLE SUCH MATERIAL NON-PUBLIC INFORMATION IN ACCORDANCE WITH THOSE PROCEDURES AND APPLICABLE LAW, INCLUDING FEDERAL AND STATE SECURITIES LAWS.

ALL INFORMATION, INCLUDING REQUESTS FOR WAIVERS AND AMENDMENTS, FURNISHED BY BORROWER OR THE ADMINISTRATIVE AGENT PURSUANT TO, OR IN THE COURSE OF ADMINISTERING, THIS AGREEMENT WILL BE SYNDICATE-LEVEL INFORMATION, WHICH MAY CONTAIN MATERIAL NON-PUBLIC INFORMATION ABOUT BORROWER, THE GUARANTOR, AND THEIR RELATED PARTIES OR THEIR RESPECTIVE SECURITIES. ACCORDINGLY, EACH LENDER REPRESENTS TO BORROWER AND THE ADMINISTRATIVE AGENT THAT IT HAS IDENTIFIED IN ITS ADMINISTRATIVE QUESTIONNAIRE A CREDIT CONTACT WHO MAY RECEIVE INFORMATION THAT MAY CONTAIN MATERIAL NON-PUBLIC INFORMATION IN ACCORDANCE WITH ITS COMPLIANCE PROCEDURES AND APPLICABLE LAW.

11.14 Interest Rate Limitation.
Notwithstanding anything herein to the contrary, if at any time the interest rate applicable to any Loan, together with all fees, charges and other amounts which are treated as interest on such Loan under applicable law (collectively the "Charges"), shall exceed the maximum lawful rate (the "Maximum Rate") which may be contracted for, charged, taken, received or reserved by the Lender holding such Loan in accordance with applicable law, the rate of interest payable in respect of such Loan hereunder, together with all Charges payable in respect thereof, shall be limited to the Maximum Rate and, to the extent lawful, the interest and Charges that would have been payable in respect of such Loan but were not payable as a result of the operation of this Section shall be cumulated and the interest and Charges payable to such Lender in respect of other Loans or periods shall be increased (but not above the Maximum Rate therefor) until such cumulated amount, together with interest thereon at the Federal Funds Effective Rate to the date of repayment, shall have been received by such Lender

11.15 USA Patriot Act.

Administrative Agent and each Lender hereby notifies Borrower that pursuant to the requirements of the USA Patriot Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)) (the "Act"), it is required to obtain, verify and record information that identifies Borrower, which information includes the name and address of Borrower and other information that will allow Administrative Agent or such Lender to identify Borrower in accordance with the Act.

11.16

Administrative Agent Approvals.

With respect to matter under this Agreement requiring the approval or consent of Administrative Agent or any other exercise of discretion by Administrative Agent, Administrative Agent shall exercise its judgment reasonably and in good faith without unreasonable delay after receipt of the necessary information to make a fully informed decision.

11.17

Replacement Documentation.

Upon receipt of an affidavit of an officer of Administrative Agent or any of the Lenders as to the loss, theft, destruction or mutilation of a Note or any other security document which is not of public record, and, in the case of any such loss, theft, destruction or mutilation, Borrower will issue, in lieu thereof, a replacement Note or other security document in the same principal amount thereof and otherwise of like tenor. In the event that Borrower issues such replacement Note or other security document, the Lender who is the payee on the lost, destroyed, mutilated or stolen Note or security document shall indemnify and hold harmless Borrower from any liability incurred by Borrower in connection with the lost, stolen, destroyed or mutilated Note or security document.

11.18 Swap Agreements.

All Swap Agreements, if any, between Borrower and any Lender or Affiliate of any Lender are independent agreements governed by the written provisions of said Swap Agreements, which will remain in full force and effect, unaffected by any repayment, prepayment, acceleration, reduction, increase or change in the terms of the Loan Documents, except as otherwise expressly provided in said written Swap Agreements, and any payoff statement from Administrative Agent relating to the Loans shall not apply to said Swap Agreements.

[SIGNATURE PAGE FOLLOWS]

IN WITNESS WHEREOF, the parties hereto have executed this Loan and Security Agreement as of the day and year first above written.

BORROWER:

REXFORD INDUSTRIAL – SDLAOC, LLC, a Delaware limited liability company

By:Rexford Industrial Realty, L.P., a Maryland limited partnership, its Managing Member

> By:Rexford Industrial Realty, Inc., a Maryland corporation, its General Partner

> > By: <u>/s/ Howard Schwimmer</u>
> > Name: <u>Howard Schwimmer</u>
> > Title: Co-Chief Executive Officer

ADMINISTRATIVE AGENT/LENDER:

JPMORGAN CHASE BANK, N.A., a national bank association

By: _/s/ Krista Valley
Name: _Krista Valley
Title: _Authorized Officer

SCHEDULE 1.01

Lenders

Name Address Commit

\$48,500,0

JPMorgan Chase Bank, N.A. JPMorgan Chase Bank, N.A.

JPMorgan Chase Bank, N.A. 2029 Century Park East, 38th Floor Los Angeles, California 90067 Attention: Faina Birger

Schedule 1.01-1

LEGAL DESCRIPTION OF 9404 CABOT PROPERTY

Real property in the City of San Diego, County of San Diego, State of California, described as follows:

PARCEL A:
PARCEL 2 OF PARCEL MAP NO. 4947, IN THE CITY OF SAN DIEGO, COUNTY OF SAN DIEGO, STATE OF CALIFORNIA, ACCORDING TO MAP THEREOF ON FILE IN THE OFFICE OF COUNTY RECORDER OF SAN DIEGO COUNTY, BEING A CONSOLIDATION AND DIVISION OF PARCELS 2 AND 3 OF PARCEL MAP NO. 2647, WHICH IS A DIVISION OF LOT 6 OF MIRAMAR MESA INDUSTRIAL PARK UNIT NO. 1, ACCORDING TO MAP THEREOF NO. 7715, FILED IN THE OFFICE OF COUNTY RECORDER OF SAN DIEGO COUNTY.

PARCEL B:
A NON-EXCLUSIVE EASEMENT APPURTENANT TO PARCEL 2 OF PARCEL MAP NO. 4947, FOR INGRESS AND EGRESS OVER AND ACROSS THAT PORTION OF THE CABOT EASEMENT PREMISES AS DESCRIBED IN THAT CERTAIN RECIPROCAL EASEMENT DEED RECORDED AUGUST 19, 1977, AS FILE/PAGE NO. 77-339425, OF OFFICIAL RECORDS OF SAN DIEGO COUNTY.

LEGAL DESCRIPTION OF 9455 CABOT PROPERTY

Real property in the City of San Diego, County of San Diego, State of California, described as follows:

PARCEL A:
PARCEL 1 OF PARCEL MAP NO. 2641, IN THE CITY OF SAN DIEGO, COUNTY OF SAN DIEGO, STATE OF CALIFORNIA, ACCORDING TO MAP THEREOF ON FILE IN THE OFFICE OF COUNTY RECORDER OF SAN DIEGO COUNTY, BEING A DIVISION OF LOT 5 OF MIRAMAR MESA INDUSTRIAL PARK UNIT NO. 1, ACCORDING TO MAP THEREOF NO. 7715, FILED IN THE OFFICE OF COUNTY RECORDER OF SAN DIEGO COUNTY.

COUNTY.

PARCEL B:
AN EASEMENT TOGETHER WITH THE RIGHT TO GRANT AND TRANSFER THE SAME, FOR RAILROAD AND RELATED PURPOSES, INCLUDING ALL MAINTENANCE ACTIVITIES RELATED THERETO, OVER
THE EASTERLY 12.50 FEET OF PARCEL 2 OF PARCEL MAP NO. 4004, IN THE CITY OF SAN DIEGO, COUNTY OF SAN DIEGO, STATE OF CALIFORNIA, ACCORDING TO MAP THEREOF ON FILE IN THE OFFICE OF
COUNTY RECORDER OF SAN DIEGO COUNTY, BEING A DIVISION OF PARCEL 2 OF PARCEL MAP NO. 2641, FILED IN THE OFFICE OF COUNTY RECORDER OF SAN DIEGO COUNTY, SAID PARCEL MAP NO.
2641, BEING A DIVISION OF LOT 5 OF MIRAMAR MESA INDUSTRIAL PARK UNIT NO. 1, ACCORDING TO MAP THEREOF NO. 7715, FILED IN THE OFFICE OF COUNTY RECORDER OF SAN DIEGO COUNTY.

LEGAL DESCRIPTION OF 9340 CABOT PROPERTY

Real property in the City of San Diego, County of San Diego, State of California, described as follows:
PARCEL 1 OF PARCEL MAP NO. 3549, IN THE CITY OF SAN DIEGO, COUNTY OF SAN DIEGO, STATE OF CALIFORNIA, ACCORDING TO MAP THEREOF ON FILE IN THE OFFICE OF COUNTY RECORDER OF SAN DIEGO COUNTY, BEING A DIVISION OF PARCEL 1 OF PARCEL MAP NO. 2647, BEING A DIVISION OF A PORTION OF LOT 6 OF MIRAMAR MESA INDUSTRIAL PARK UNIT NO. 1, ACCORDING TO MAP THEREOF NO. 7715, FILED IN THE OFFICE OF COUNTY RECORDER OF SAN DIEGO COUNTY.

LEGAL DESCRIPTION OF 9855 DISTRIBUTION PROPERTY

Real property in the City of San Diego, County of San Diego, State of California, described as follows:

PARCEL A:
PARCEL 2 OF PARCEL MAP NO. 12403, IN THE CITY OF SAN DIEGO, COUNTY OF SAN DIEGO, FILED IN THE OFFICE OF THE COUNTY RECORDER OF SAN DIEGO COUNTY OCTOBER 29, 1982.

PARCEL B:
A PERPETUAL, NON-EXCLUSIVE EASEMENT FOR THE PURPOSE OF VEHICULAR AND PEDESTRIAN ACCESS, INGRESS AND EGRESS OVER THE EXISTING DRIVEWAY, BEING MORE PARTICULARLY DESCRIBED IN THE RECIPROCAL EASEMENT AGREEMENT DATED AUGUST 7, 2000 AND RECORDED AUGUST 7, 2000 AS INSTRUMENT NO. 2000-0416402 OF OFFICIAL RECORDS. PARCEL C:

A PERPETUAL, NON-EXCLUSIVE EASEMENT FOR THE PURPOSE OF VEHICULAR AND PEDESTRIAN ACCESS, INGRESS AND EGRESS, BEING MORE PARTICULARLY DESCRIBED IN THE RECIPROCAL EASEMENT AGREEMENT DATED DECEMBER 10, 2001 AND RECORDED DECEMBER 17, 2001 AS INSTRUMENT NO. 2001-0925651 OF OFFICIAL RECORDS.

LEGAL DESCRIPTION OF 9755 DISTRIBUTION PROPERTY

Real property in the City of San Diego, County of San Diego, State of California, described as follows:

LOT 4 OF G-W INDUSTRIAL PARK ANNEX, IN THE CITY OF SAN DIEGO, COUNTY OF SAN DIEGO, STATE OF CALIFORNIA, ACCORDING TO MAP THEREOF NO. 7813, FILED IN THE OFFICE OF THE COUNTY RECORDER OF SAN DIEGO COUNTY, DECEMBER 17, 1973.

LEGAL DESCRIPTION OF 14955-14791 E. SALT LAKE PROPERTY

Real property in the City of Industry, County of Los Angeles, State of California, described as follows:

ALL OF LOT 7 AND THAT PORTION OF LOT 6, BOTH IN BLOCK 22 OF TRACT NO. 1343, IN THE CITY OF INDUSTRY, AS PER MAP RECORDED IN BOOK 20 PAGES 10 AND 11 OF MAPS, IN THE OFFICE OF THE COUNTY RECORDER OF SAID COUNTY, LYING SOUTHERLY OF THE CENTERLINE OF THAT CERTAIN STRIP OF LAND, 200.00 FEET WIDE, DESCRIBED IN EASEMENT TO LOS ANGELES COUNTY FLOOD CONTROL DISTRICT, RECORDED IN BOOK 18176 PAGE 173 OFFICIAL RECORDS OF SAID COUNTY.

EXCEPT THEREFROM THAT PORTION OF SAID LOT 6 LYING NORTHWESTERLY OF THE NORTHEAST PROLONGATION OF THE NORTHWESTERLY LINE OF SAID LOT 7.

ALSO EXCEPT THEREFROM ALL MINERALS AND MINERAL RIGHTS OF EVERY KIND AND CHARACTER NOW KNOWN TO EXIST OR HEREAFTER DISCOVERED, INCLUDING, WITHOUT LIMITING THE GENERALITY OF THE FOREGOING, OIL AND GAS RIGHTS THERETO, TOGETHER WITH THE SALE, EXCLUSIVE AND PERPETUAL RIGHT TO EXPLORE FOR, REMOVE AND DISPOSE OF SAID MINERALS BY ANY MEANS OR METHODS SUITABLE TO THE GRANTOR, ITS SUCCESSORS AND ASSIGNS, BUT WITHOUT ENTERING UPON OR USING THE SURFACE OF THE LANDS HEREBY CONVEYED, AND IN SUCH A MANNER AS NOT TO DAMAGE THE SURFACE OF SAID LANDS OR TO INTERFERE WITH THE USE THEREOF BY THE GRANTEE, ITS SUCCESSORS OR ASSIGNS, AS RESERVED BY E.C. WOODWARD AND CITIZENS NATIONAL TRUST & SAVINGS BANK OF LOS ANGELES, AS TRUSTEE AND THE WILL OF AUGUST V. RANDORF, DECEASED, IN A DEED RECORDED FEBRUARY 20, 1957 AS INSTRUMENT NO. 1304 IN BOOK 53698 PAGE 126 OF OFFICIAL RECORDS.

A PORTION OF SAID LAND IS ALSO KNOWN AS PARCEL 4 OF PARCEL MAP NO. 120 FILED IN BOOK 97, PAGE 72 OF PARCEL MAPS, IN THE OFFICE OF THE COUNTY RECORDER OF SAID COUNTY.

LEGAL DESCRIPTION OF 5235 E. HUNTER PROPERTY

Real property in the City of Anaheim, County of Orange, State of California, described as follows:

PARCEL 2 OF PARCEL MAP NO. 86-188, IN THE CITY OF ANAHEIM, COUNTY OF ORANGE, STATE OF CALIFORNIA, AS SHOWN ON A MAP FILED IN BOOK 213, PAGES 11 AND 12 OF PARCEL MAPS, IN THE OFFICE OF THE COUNTY RECORDER OF ORANGE COUNTY, CALIFORNIA.

EXCEPT THEREFROM ALL OIL, GAS, MINERALS AND OTHER HYDROCARBON SUBSTANCES, IN, UPON AND UNDER SAID SUBJECT PROPERTY BELOW A DEPTH OF 500 FEET FROM THE SURFACE THEREOF, WITHOUT THE RIGHT OF SURFACE ENTRY FOR THE PURPOSE OF REMOVING OR EXTRACTING THE SAME, AS RESERVED IN DEEDS RECORDED JUNE 21, 1965 IN BOOK 7563, PAGES 252, 257 AND 262 OF OFFICIAL RECORDS, BY AUGUST BIEGER AND OTHERS.

APN: 346-211-27, 346-211-18 AND 29

LEGAL DESCRIPTION OF 3880 W. VALLEY PROPERTY

Real property in the City of Pomona, County of Los Angeles, State of California, described as follows:

PARCEL 1: Assessor's Parcel Number: 8709-026-005

THAT PORTION OF SECTION 33 AND SECTION 34, TOWNSHIP 1 SOUTH, RANGE 9 WEST, SAN BERNARDINO MERIDIAN, AND THAT PORTION OF RANCHO LOS NOGALES, IN THE CITY OF POMONA, COUNTY OF LOS ANGELES, STATE OF CALIFORNIA, AS SHOWN ON MAPS RECORDED IN BOOK 3, PAGES 164 AND 165 OF PATENTS, AND IN BOOK 6, PAGES 248 AND 249 OF MISCELLANEOUS RECORDS, IN THE OFFICE OF THE COUNTY RECORDER OF SAID COUNTY, AS FOLLOWS:

COMMENCING AT THE INTERSECTION OF THE SOUTHEASTERLY RIGHT-OF-WAY LINE OF VALLEY BOULEVARD (66 FEET IN WIDTH) WITH THE NORTHWESTERLY RIGHT-OF-WAY LINE OF THE SOUTHERN PACIFIC RAILROAD (100 FEET IN WIDTH) AS SAID RIGHT-OF-WAY LINE AS SHOWN ON RECORD OF SURVEY FILED IN BOOK 76, PAGES 51 TO 56 INCLUSIVE OF RECORD OF SURVEYS, IN THE OFFICE OF THE COUNTY RECORDER OF SAID COUNTY, SAID POINT OF COMMENCEMENT ALSO BEING THE NORTHHEASTERLY TERMINUS OF THAT CERTAIN COURSE SHOWN ON SAID PROVED SURVEY AS HAVING A BEARING OF NORTH 32° 29' 32.55" EAST AND A LENGTH OF 4325.36 FEET; THENCE SOUTH 32° 29' 32.55" WEST ALONG SAID RAILROAD RIGHT-OF-WAY, A DISTANCE OF 1128.69 FEET TO THE POINT OF BEGINNING, SAID POINT BEING ON THE CENTER LINE OF A 30.00 FOOT EASEMENT IN FAVOR OF THE OWNER OF THE 105.42 ACRE AND THE 144.56 ACRE TRACTS OF LAND CONVEYED TO CHARLES M. WRIGHT BY DEED RECORDED IN BOOK 1705, PAGE 148 OF DEEDS; THENCE NORTH 53° 00' 55" WEST ALONG A CENTER LINE 346.08 FEET TO THE SOUTHEASTERLY RIGHT-OF-WAY LINE OF SAID VALLEY BOULEVARD;

THENCE SOUTH 49° 52' 58" WEST ALONG SAID RIGHT-OF-WAY LINE A DISTANCE OF 418.81 FEET TO THE INTERSECTION OF THAT CERTAIN COURSE DESCRIBED IN DEED RECORDED DECEMBER 14, 1962 IN BOOK D-1856, PAGE 352, OFFICIAL RECORDS AS INSTRUMENT NO. 5608 AS HAVING A BEARING OF NORTH 40° 08' 31" WEST AND A LENGTH OF 50.00 FEET, WITH SAID SOUTHEASTERLY RIGHT-OF-WAY LINE OF VALLEY BOULEVARD; THENCE ALONG THE EASTERLY LINE OF THAT PARCEL DESCRIBED IN THE AFOREMENTIONED DEED RECORDED IN BOOK D-1856 THE FOLLOWING COURSES; SOUTH 40° 08' 31" EAST 50.00 FEET; SOUTH 19° 51' 19" WEST 108.27 FEET, SOUTH 03° 51' 29" EAST 251.38 FEET, SOUTH 86° 08' 31" WEST 5.00 FEET, SOUTH 03° 51' 29" EAST 148.69 FEET TO THE BEGINNING OF A TANGENT CURVE CONCAVE WESTERLY AND HAVING A RADIUS OF 850.00 FEET; THENCE SOUTHERLY ALONG SAID CURVE 539.13 FEET TO A POINT, SAID POINT BEING THE SOUTHERLY TERMINUS OF SAID CURVE, A RADIAL LINE THROUGH SAID POINT BEARS SOUTH 57° 31' 02" EAST; THENCE RADIAL TO SAID CURVE AND

LEAVING SAID EASTERLY LINE SOUTH 57° 31' 02" EAST 0.29 FEET TO A POINT IN THE AFOREMENTIONED NORTHWESTERLY RIGHT-OF-WAY LINE OF THE SOUTHERN PACIFIC RAILROAD; THENCE ALONG SAID NORTHWESTERLY LINE NORTH 32° 29' 32.55" EAST 1321.98 FEET TO THE POINT OF BEGINNING.

EXCEPT THEREFROM ALL OIL, GAS AND OTHER HYDROCARBONS AND MINERALS NOW OR AT ANY TIME HEREAFTER SITUATED BELOW A DEPTH OF 500 FEET (BUT WITH NO RIGHT TO USE OF THE SURFACE AND FROM THE SURFACE TO A DEPTH OF 500 FEET BELOW THE SURFACE) MEASURED VERTICALLY FROM THE CONTOUR OF THE SURFACE THEREOF, RECORDED OCTOBER 14, 1975 AS INSTRUMENT NO. 3234, OFFICIAL RECORDS.

PARCEL 2:

AN EASEMENT FOR "RANCH ROAD" PURPOSES, OVER A STRIP OF LAND 15 FEET WIDE, NORTHERLY OF AND PARALLEL WITH THE NORTHEASTERLY BOUNDARY OF PARCEL 1, AS CREATED BY THAT CERTAIN INSTRUMENT RECORDED IN BOOK 1705 PAGE 148, OF DEEDS.

EXHIBIT B

PROMISSORY NOTE

REXFORD INDUSTRIAL - SDLAOC, LLC, a Delware limited liability company ("Borrower"), promises to pay to the order of

[Date]

This Note is one of the Notes issued pursuant to, and is entitled to the benefits of, the Loan and Security Agreement dated as of June 24, 2014 (which, as it may be amended or modified and in effect from time to time, is herein called the "Agreement"), among Borrower, the lenders referenced therein, including Lender, and JPMorgan Chase Bank, N.A., as Administrative Agent, to which Agreement reference is hereby made for a statement of the terms and conditions governing this Note, including the terms and conditions under which this Note may be prepaid or its maturity date accelerated. This Note is secured and guaranteed pursuant to the Loan Documents, all as more specifically described in the Agreement, and reference is made thereto for a statement of the terms and provisions thereof. Capitalized terms used herein and not otherwise defined herein are used with the meanings attributed to them in the Agreement. BORROWER: REXFORD INDUSTRIAL - SDLAOC, LLC,

a Delaware limited liability company

By:Rexford Industrial Realty, L.P., a Maryland limited partnership, its Managing Member

By:Rexford Industrial Realty, Inc., a Maryland corporation, its General Partner

Title: Co-Chief Executive Officer

EXHIBIT C

ASSIGNMENT AND ASSUMPTION

This Assignment and Assumption (the "Assignment and Assumption") is dated as of the Effective Date set forth below and is entered into by and between [Insert name of Assigner"]. Capitalized terms used but not defined herein shall have the meanings given to them in the Loan Agreement identified below (as amended, the "Loan Agreement"), receipt of a copy of which is hereby acknowledged by the Assignee. The Standard Terms and Conditions set forth in Annex 1 attached hereto are hereby agreed to and incorporated herein by reference and made a part of this Assignment and Assumption as if set forth herein in full.

For an agreed consideration, the Assignor hereby irrevocably sells and assigns to the Assignee, and the Assignee hereby irrevocably purchases and assumes from the Assignor, subject to and in accordance with the Standard Terms and Conditions and the Loan Agreement, as of the Effective Date inserted by Administrative Agent as contemplated below (i) all of the Assignor's rights and obligations in its capacity as a Lender under the Loan Agreement and any other documents or instruments delivered pursuant thereto to the extent related to the amount and percentage interest identified below of all of such outstanding rights and obligations of the Assignor under the respective facilities identified below and (ii) to the extent permitted to be assigned under applicable law, all claims, suits, causes of action and other rights of the Assignor (in its capacity as a Lender) against any Person, whether known or unknown, arising under or in connection with the Loan Agreement, any other documents or instruments delivered pursuant thereto or the loan transactions governed thereby or in any way based on or related to any of the foregoing, including contract claims, malpractice claims, statutory claims and all other claims at law or in equity related to the rights and obligations sold and assigned pursuant to clauses (i) above (the rights and obligations sold and assigned pursuant to clauses (i) and (ii) above being referred to herein collectively as the "Assigned Interest"). Such sale and assignment is without recourse to the Assignor and, except as expressly provided in this Assignment and Assumption, without representation or warranty by the Assignor.

2.	Assignee: [and is an Affiliate/Approved Fund of [identify Lender]]
3.	Borrower:Rexford Industrial – SDLAOC, LLC
4.	Administrative Agent: JPMorgan Chase Bank, N.A., as the administrative agent under the Loan Agreement

C-1

Credit Agreement: The \$48,500,000.00 Loan and Security Agreement dated as of June 24, 2014 among Rexford Industrial – SDLAOC, LLC, the Lenders parties

thereto, and JPMorgan Chase Bank, N.A., as Administrative Agent

Assigned Interest:

[SIGNATURES ON THE FOLLOWING PAGE]

he terms set forth in this Assignmer	nt and Assumption are hereby agreed to:
	ASSIGNOR:
	a,
	By:
	ASSIGNEE:
	a
	By:Name:
Consented to and] Accepted:	
ADMINISTRATIVE AGENT:	
by:	

[Consented to:]	
[BORROWER:	

______,

ANNEX 1

STANDARD TERMS AND CONDITIONS FOR ASSIGNMENT AND ASSUMPTION

Representations and Warranties.

1.

1.1 Assignor. The Assignor (a) represents and warrants that (i) it is the legal and beneficial owner of the Assigned Interest, (ii) the Assigned Interest is free and clear of any lien, encumbrance or other adverse claim and (iii) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and Assumption and to consummate the transactions contemplated hereby; and (b) assumes no responsibility with respect to (i) any statements, warranties or representations made in or in connection with the Loan Agreement or any other Loan Document, (ii) the execution, legality, enforceability, genuineness, sufficiency or value of the Loan Documents or any collateral thereunder, (iii) the financial condition of Borrower, any of its subsidiaries or any other Person obligated in respect of any Loan Document or (iv) the performance or observance by Borrower, any of its subsidiaries or affiliates or any other Person of any of their respective obligations under any Loan Document.

Assignee. The Assignee (a) represents and warrants that (i) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and Assumption and to consummate the transactions contemplated hereby and to become a Lender under the Loan Agreement, (ii) it satisfies the requirements, if any, specified in the Loan Agreement that are required to be satisfied by it in order to acquire the Assigned Interest and become a Lender, (iii) from and after the Effective Date, it shall be bound by the provisions of the Loan Agreement as a Lender thereunder and, to the extent of the Assigned Interest, shall have the obligations of a Lender thereunder, (iv) it has received a copy of the Loan Agreement, together with copies of the most recent financial statements delivered pursuant to Section 4.11 thereof, as applicable, and such other documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into this Assignment and Assumption and to purchase the Assigned Interest on the basis of which it has made such analysis and decision independently and without reliance on Administrative Agent or any other Lender, and (v) attached to the Assignment and Assumption is any documentation required to be delivered by it pursuant to the terms of the Loan Agreement, duly completed and executed by the Assignee; and (b) agrees that (i) it will, independently and without reliance on Administrative Agent, the Assignor or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under the Loan Documents, and (ii) it will perform in accordance with their terms all of the obligations which by the terms of the Loan Documents are required to be performed by it as a Lender.

2. Payments. From and after the Effective Date, Administrative Agent shall make all payments in respect of the Assigned Interest (including payments of principal, interest, fees and other amounts) to the Assignor for amounts which have accrued to but excluding the

Effective Date and to the Assignee for amounts which have accrued from and after the Effective Date.

3. <u>General Provisions.</u> This Assignment and Assumption shall be binding upon, and inure to the benefit of, the parties hereto and their respective successors and assigns. This Assignment and Assumption may be executed in any number of counterparts, which together shall constitute one instrument. Acceptance and adoption of the terms of this Assignment and Assumption by the Assignment and Assumption by the Assignment and Assumption. This Assignment and Assumption shall be governed by, and construed in accordance with, the law of the State of California.

EXHIBIT D

LIMITED RECOURSE AND GUARANTY PROVISIONS

A. Notwithstanding anything herein to the contrary, at all times prior to the payment in full of the Obligations, Borrower and Guarantor shall have unlimited liability for the payment and performance of the Obligations if:

there occurs any sale, exchange, conveyance, transfer, mortgage, assignment, pledge or encumbrance, either voluntarily or involuntarily, of any right, title or interest of Borrower in and to the Property or any portion thereof or in a direct or indirect interest in Borrower which, in any case, is either (i) not permitted under the Loan Documents (other than as the result of any enforcement actions taken by Administrative Agent in writing in each case, or (ii) any Lien other than Liens securing or resulting from (aa) claims of Persons supplying labor or materials to the Property ("Mechanics Liens"), (bb) non-delinquent taxes, assessments and governmental charges levied upon, assessed or charged against the Property ("Tax Liens"), (cc) the Loan Documents (including, without limitation, any enforcement actions taken by Administrative Agent or any Lender pursuant to the Loan Documents), (dd) matters disclosed in the Title Policy, (ee) matters being contested pursuant to and in accordance with the Loan Documents; (ff) Permitted Encumbrances (including for purposes of this section any Leases); or (gg) matters specifically approved in writing by Administrative Agent in each case; or

Borrower voluntarily files a bankruptcy petition or the assignment for the benefit of creditors (or any similar state law insolvency proceeding); or
 Any involuntary bankruptcy proceeding is filed against Borrower:
 by Guarantor or by any general partner, managing member or other constituent entity or affiliate of Borrower or Guarantor (each a "Restricted")

Entity");
(b) by any Person acting at the direction or request of, or in collusion or by agreement with, Borrower, Guarantor or any Restricted Entity of either of them, or

(c) by any other Person if Borrower fails to oppose in court such filing in good faith by asserting factually and legally appropriate defenses available as a defense to an involuntary bankruptcy proceeding, or Borrower, Guarantor or any Restricted Entity of either of them otherwise consents to, acquiesces in, agrees with, files court papers in any way supportive of, or joins in such filling.

In addition, notwithstanding anything herein to the contrary, at all times prior to the payment in full of the Obligations, the Guarantor guaranties to the Administrative Agent for the benefit of the

B. In addition, notwithstanding anything herein to the contrary, at all times prior to the payment in full of the Obligations, the Guarantor guaranties to the Administrative Agent for the benefit of th Lenders, the full and prompt payment of, and agrees to pay protect, guarantee,

indemnify, defend and hold harmless the Administrative Agent and the Lenders from and against, any liability, loss, damage, costs and expenses (including legal fees and disbursements) actually suffered by the Administrative Agent or the Lenders, and caused by or related to or as a result of the following:

- any willful misconduct, fraud or material misrepresentation in any of the Loan Documents by Borrower, Guarantor, or any Affiliate, or any of their respective partners. shareholders, members, officers, or directors;
 - 2.
- any intentional physical waste by Borrower, Guarantor or any of their respective Affiliates with respect to any portion of the Property; the removal or disposal by, or at the direction or request of, or in collusion or by agreement with, Borrower, Guarantor or any Restricted Entity of either of them, of any property in which Administrative Agent on behalf of the Lender has a Lien, in violation of the terms of the Loan Documents;
- any failure of Borrower to maintain the insurance coverages required under the Loan Documents, provided, however, that Guarantor shall not be liable to the extent (i) Borrower is unable to pay for such insurance coverage because there are not sufficient funds generated by operations of the Property available to Borrower under the terms of the Loan Documents to make such payments, and (ii) Borrower provides at least thirty (30) days' notice to Administrative Agent that Borrower will not maintain the required insurance coverages as a result of the circumstances described in clause (i) above; provided, further, that Guarantor shall have no liability hereunder for Borrower's failure to pay insurance premiums during any period in which Administrative Agent has required Borrower to deposit insurance premiums into an impound account with Administrative Agent (unless Administrative Agent has disbursed to Borrower an amount not less than the required insurance premiums from such impound account for the payment of such premiums);
- any misapplication or misappropriation by Borrower or Guarantor of any insurance proceeds or condemnation awards in a manner not permitted by the Loan Documents; any failure by Borrower to pay to Administrative Agent all revenues received by or on behalf of Borrower from the operation or ownership of the Property during the 6. continuance of any Default, less only that portion of such revenues which is actually used by Borrower to operate the Property in the ordinary course of business pursuant to the terms of the Loan Documents (except to
- the extent that Borrower did not have the legal right to pay such revenues to Administrative Agent because of bankruptcy, receivership or similar judicial proceeding); any Mechanics Liens arising in connection with labor or materials performed or supplied at the request or direction of Borrower, Guarantor or any Restricted Entity of either
- of them, that are either not (i) being contested pursuant to and in accordance with the Loan Documents, or (ii) bonded in a manner reasonably

- acceptable to Administrative Agent or discharged within thirty (30) days after Borrower first receives notice of the recording of such Mechanics Liens;

 8. any Tax Liens that are either not (i) being contested pursuant to and in accordance with the Loan Documents, or (ii) bonded in a manner reasonably acceptable to Administrative Agent or discharged within thirty (30) days after Borrower first receives notice of the recording of such Tax Liens; or
- the execution by Borrower after the date hereof of any Lease that is not an Approved Lease; any and all rent abatement and/or offsets taken or asserted by any Lessee under any Lease (whether pursuant to the express terms of such Lease or applicable law) (i) against a receiver appointed over any portion of the Property (a "Receiver"), or (ii) in the event Administrative Agent, any Lender or any Affiliate or nominee of Administrative Agent or any Lender (as the case may be, a "Successor Owner") shall acquire all or any portion of the Property pursuant to a foreclosure (whether judicial or non-judicial), a receivership sale, or by acceptance of a deed-in-lieu of foreclosure (a "Foreclosure").
- Transfer"), against any such Successor Owner as a result of any failure by Borrower to perform any obligations of Borrower under such Lease prior to the appointment of such Receiver or the consummation of such Foreclosure Transfer (as evidenced by the recordation of the applicable trustee's deed or grant deed in the official records of the applicable county); or

 11. any security deposit paid to the lessee pursuant to that certain lease dated June 11, 2011 between Borrower (or Borrower's predecessor in title), as landlord, and Miramar Transportation, Inc. with respect to the 9340 Cabot Property, but only to the extent that such security deposit is not actually delivered by Borrower to Administrative Agent or the applicable Successor Owner in connection with any Foreclosure Transfer of such 9340 Cabot Property.
- These provisions are in addition to, and not in limitation of, the obligations of the Guarantor under the Environmental Indemnity.

EXHIBIT E

APPROVED LEASE TERMS

A Major Lease shall be an Approved Lease if, and only if, (a) such Major Lease is approved by Administrative Agent in writing, or (b) if such Major Lease is "deemed" approved by Administrative Agent, (ii) is entered into in the ordinary course of business with a bona fide unrelated third party tenant, and Borrower, acting in good faith and exercising due diligence, has determined that the tenant is financially capable of performing its obligations under the Lease; (iii) is received by Administrative Agent, together with any guaranty(ies) and, if requested by Administrative Agent, all financial information received by Borrower regarding the tenant and any guarantor(s), within fifteen (15) days after execution; (iv) reflects an arm's length transaction; (v) contains no option or right of first refusal to purchase all or any portion of the affected Property or any present or future interest therein; (vi) requires the tenant to execute and deliver to Administrative Agent an estoppel certificate in form and substance acceptable to Administrative Agent within ten (10) Business Days after notice from Administrative Agent; (vii) does not cover in excess of fifty percent (50%) of the net rentable square feet of the applicable Improvements or have an effective rental rate that is less than the applicable Minimum Rental Rate; and (viii) is subject to a subordination, non-disturbance and attornment agreement in form and substance satisfactory to Administrative Agent. Prior to execution, Borrower shall provide to Administrative Agent via nationally recognized overnight courier a written request for approval (which request shall specifically reference this Section of this Agreement) of each Lease that does not meet the foregoing requirements for "deemed" approval by Administrative Agent, which request shall be accompanied by a correct and complete copy of the applicable Lease, including any exhibits, and any guaranty(ies) thereof. Borrower shall pay all reasonable costs incurred by Administrative Agent in reviewing and appro

[FORM OF]

U.S. TAX COMPLIANCE CERTIFICATE

(For Foreign Lenders That Are Not Partnerships For U.S. Federal Income Tax Purposes)

Reference is hereby made to the Loan and Security Agreement dated as of [] (as amended, supplemented or otherwise modified from time to time, the "Credit Agreement"), among [], and each lender from time to time party thereto.

Pursuant to the provisions of Section 3.11 of the Credit Agreement, the undersigned hereby certifies that (i) it is the sole record and beneficial owner of the Loan(s) (as well as any Note(s) evidencing such Loan(s)) in respect of which it is providing this certificate, (ii) it is not a bank within the meaning of Section 881(c)(3)(A) of the Code, (iii) it is not a ten percent shareholder of Borrower within the meaning of Section 881(c)(3)(B) of the Code and (iv) it is not a controlled foreign corporation related to Borrower as described in Section 881(c)(3)(C) of the Code.

The undersigned has furnished the Administrative Agent and Borrower with a certificate of its non-U.S. Person status on IRS Form W-8BEN. By executing this certificate, the undersigned agrees that (1) if the information provided on this certificate changes, the undersigned shall promptly so inform Borrower and the Administrative Agent, and (2) the undersigned shall have at all times furnished Borrower and the Administrative Agent with a properly completed and currently effective certificate in either the calendar year in which each payment is to be made to the undersigned, or in either of the two calendar years preceding such payments.

Unless otherwise defined herein, terms defined in the Credit Agreement and used herein shall have the meanings given to them in the Credit Agreement.

[NAME OF LENDER]

By: Name:

[FORM OF]

U.S. TAX COMPLIANCE CERTIFICATE

(For Foreign Participants That Are Not Partnerships For U.S. Federal Income Tax Purposes)

Reference is hereby made to the Loan and Security Agreement dated as of [] (as amended, supplemented or otherwise modified from time to time, the "Credit Agreement"), among [], and each lender from time to time party thereto.

Pursuant to the provisions of Section 3.11 of the Credit Agreement, the undersigned hereby certifies that (i) it is the sole record and beneficial owner of the participation in respect of which it is providing this certificate, (ii) it is not a bank within the meaning of Section 881(c)(3)(A) of the Code, (iii) it is not a ten percent shareholder of Borrower within the meaning of Section 881(c)(3)(B) of the Code, and (iv) it is not a controlled foreign corporation related to Borrower as described in Section 881(c)(3)(C) of the Code.

The undersigned has furnished its participating Lender with a certificate of its non-U.S. Person status on IRS Form W-8BEN. By executing this certificate, the undersigned agrees that (1) if the information provided on this certificate changes, the undersigned shall promptly so inform such Lender in writing, and (2) the undersigned shall have at all times furnished such Lender with a properly completed and currently effective certificate in either the calendar year in which each payment is to be made to the undersigned, or in either of the two calendar years preceding such payments.

Unless otherwise defined herein, terms defined in the Credit Agreement and used herein shall have the meanings given to them in the Credit Agreement.

By:

Name:
Title:
Date: ______, 20[]

[FORM OF]

U.S. TAX COMPLIANCE CERTIFICATE

(For Foreign Participants That Are Partnerships For U.S. Federal Income Tax Purposes)

Reference is hereby made to the Loan and Security Agreement dated as of [] (as amended, supplemented or otherwise modified from time to time, the "Credit Agreement"), among [], and each lender from time to time party thereto.

Pursuant to the provisions of Section 3.11 of the Credit Agreement, the undersigned hereby certifies that (i) it is the sole record owner of the participation in respect of which it is providing this certificate, (ii) its direct or indirect partners/members are the sole beneficial owners of such participation, (iii) with respect such participation, neither the undersigned nor any of its direct or indirect partners/members is a bank extending credit pursuant to a loan agreement entered into in the ordinary course of its trade or business within the meaning of Section 881(c)(3)(A) of the Code, (iv) none of its direct or indirect partners/members is a ten percent shareholder of Borrower within the meaning of Section 881(c)(3)(B) of the Code and (v) none of its direct or indirect partners/members is a controlled foreign corporation related to Borrower as described in Section 881(c)(3)(C) of the Code.

The undersigned has furnished its participating Lender with IRS Form W-8IMY accompanied by one of the following forms from each of its partners/members that is claiming the portfolio interest exemption: (i) an IRS Form W-8BEN or (ii) an IRS Form W-8IMY accompanied by an IRS Form W-8BEN from each of such partner's/member's beneficial owners that is claiming the portfolio interest exemption. By executing this certificate, the undersigned agrees that (1) if the information provided on this certificate changes, the undersigned shall promptly so inform such Lender and (2) the undersigned shall have at all times furnished such Lender with a properly completed and currently effective certificate in either the calendar year in which each payment is to be made to the undersigned, or in either of the two calendar years preceding such payments.

Unless otherwise defined herein, terms defined in the Credit Agreement and used herein shall have the meanings given to them in the Credit Agreement.

[NAME OF PARTICIPANT]

By:

Name: Title: _____, 20[]

Date: _______, 20[]

[FORM OF]

U.S. TAX COMPLIANCE CERTIFICATE

(For Foreign Lenders That Are Partnerships For U.S. Federal Income Tax Purposes)

Reference is hereby made to the Loan and Security Agreement dated as of [] (as amended, supplemented or otherwise modified from time to time, the "Credit Agreement"), among [], and each lender from time to time party thereto.

Pursuant to the provisions of Section 3.11 of the Credit Agreement, the undersigned hereby certifies that (i) it is the sole record owner of the Loan(s) (as well as any Note(s) evidencing such Loan(s)) in respect of which it is providing this certificate, (ii) its direct or indirect partners/members are the sole beneficial owners of such Loan(s) (as well as any Note(s) evidencing such Loan(s)), (iii) with respect to the extension of credit pursuant to this Credit Agreement or any other Loan Document, neither the undersigned nor any of its direct or indirect partners/members is a bank extending credit pursuant to a loan agreement entered into in the ordinary course of its trade or business within the meaning of Section 881(c)(3)(A) of the Code, (iv) none of its direct or indirect partners/members is a controlled foreign corporation related to Borrower as described in Section 881(c)(3)(C) of the Code.

The undersigned has furnished the Administrative Agent and Borrower with IRS Form W-8IMY accompanied by one of the following forms from each of its partners/members that is claiming the portfolio interest exemption: (i) an IRS Form W-8BEN or (ii) an IRS Form W-8IMY accompanied by an IRS Form W-8BEN from each of such partner's/member's beneficial owners that is claiming the portfolio interest exemption. By executing this certificate, the undersigned agrees that (1) if the information provided on this certificate changes, the undersigned shall promptly so inform Borrower and the Administrative Agent, and (2) the undersigned shall have at all times furnished Borrower and the Administrative Agent with a properly completed and currently effective certificate in either the calendar year in which each payment is to be made to the undersigned, or in either of the two calendar years preceding such payments.

Unless otherwise defined herein, terms defined in the Credit Agreement and used herein shall have the meanings given to them in the Credit Agreement.

[NAME OF LENDER]

Bv.

Name:

F4-2

EXHIBIT G FORM OF COMPLIANCE CERTIFICATE COMPLIANCE CERTIFICATE

This Compliance Certificate ("Certificate") is delivered with reference to that certain Loan and Security Agreement dated as of June ____, 2014, between REXFORD INDUSTRIAL – SDLAOC, LLC, a Delaware limited liability company ("Borrower"), JPMORGAN CHASE BANK, N.A., in its capacity as Administrative Agent ("Administrative Agent"), and the Lenders from time to time party thereto (as amended, restated or otherwise modified from time to time, the "Loan Agreement"). Terms defined in the Loan Agreement and not otherwise defined in this Certificate shall have the meanings defined for them in the Loan Agreement. Section references in this Certificate relate to the Loan Agreement unless stated otherwise.

(the "Test	Period").	This Certificate is delivered by Borrower in accordance with Section 4.11(a)(iii) of the Loan Agreement. This Certificate is delivered with respect to the twelve-month period ended June 30, 20
I.	Section 4.16 – Debt Yie	dd. As of the end of the Test Period, the Debt Yield was%.
		<u>Debt Yield Calculation</u> : Debt Yield equals:
		NOI: \$
		Divided by:
		Aggregate Commitment \$
		Equals:%
		Pursuant to Section 4.16, Debt Yield must be no less than 9.25%.
		A review of the activities of Borrower during the Test Period has been made under the supervision of the undersigned with a view to determining whether during such Test Period Borrower and all of their respective obligations under the Loan Documents. To the best knowledge of the undersigned, during the Test Period, all covenants and conditions have been so performed and observed and no occurred and is continuing, with the exceptions set forth below in response to which Borrower and/or Guarantor has taken or proposes to take the following actions (if none, so state):

The undersigned certifies that the calculations made and the information contained herein are derived from and correctly reflect the books and records of Borrower.

[Signature on the Following Page]

Dated:	_, 20
	BORROWER: REXFORD INDUSTRIAL – SDLAOC, LLC, a Delaware limited liability company

EXHIBIT H ENVIRONMENTAL REPORTS

- Phase I Environmental Site Assessment dated May 23, 2014, prepared by ADR Environmental Group, Inc., with respect to the 9340 Cabot Property. Phase I Environmental Site Assessment dated May 23, 2014, prepared by ADR Environmental Group, Inc., with respect to the 9404 Cabot Property. Phase I Environmental Site Assessment dated May 23, 2014, prepared by ADR Environmental Group, Inc., with respect to the 9455 Cabot Property.
- Phase I Environmental Site Assessment dated May 23, 2014, prepared by ADR Environmental Group, Inc., with respect to the 9755 Distribution Property. Phase I Environmental Site Assessment dated May 14, 2014, prepared by ADR Environmental Group, Inc., with respect to the 9855 Distribution Property.
- 1. 2. 3 4. 5. 6. 7. 8. 9.

 - Phase I Environmental Site Assessment dated May 14, 2014, prepared by ADR Environmental Group, Inc., with respect to the 9855 Distribution Property. Phase I Environmental Group, Inc., with respect to the 5235 E. Hunter Property. Subsurface Investigation Report dated June 12, 2014, prepared by ADR Environmental Group, Inc., with respect to the 5235 E. Hunter Property. Phase I Environmental Site Assessment dated May 23, 2014, prepared by ADR Environmental Group, Inc., with respect to the 14955-14791 E. Salt Lake Property. Subsurface Investigation Report dated June 12, 2014, prepared by ADR Environmental Group, Inc., with respect to the 14955-14791 E. Salt Lake Property. Phase I Environmental Site Assessment dated May 23, 2014, prepared by ADR Environmental Group, Inc., with respect to the 3880 W. Valley Property. Subsurface Investigation Report dated June 16 2014, prepared by ADR Environmental Group, Inc., with respect to the 3880 W. Valley Property.

EXHIBIT I PHYSICAL CONDITION REPORTS

- PHYSICAL CONDITION REPORTS
 Physical Condition Report dated May 13, 2014, prepared by Andersen Environmental with respect to the 9340 Cabot Property.
 Physical Condition Report dated May 13, 2014, prepared by Andersen Environmental with respect to the 9404 Cabot Property.
 Physical Condition Report dated May 14, 2014, prepared by Andersen Environmental with respect to the 9455 Cabot Property.
 Physical Condition Report dated May 14, 2014, prepared by Andersen Environmental with respect to the 9755 Distribution Property.
 Physical Condition Report dated May 14, 2014, prepared by Andersen Environmental with respect to the 9855 Distribution Property.
 Physical Condition Report dated May 13, 2014, prepared by Andersen Environmental with respect to the 5235 E. Hunter Property.
 Physical Condition Report dated May 15, 2014, prepared by Andersen Environmental with respect to the 14955-14791 E. Salt Lake Property.
 Physical Condition Report dated May 12, 2014, prepared by Andersen Environmental with respect to the 3880 W. Valley Property. 1. 2. 3 4. 5. 6. 7. 8.



REXFORD INDUSTRIAL ACQUIRES NINE-BUILDING INDUSTRIAL PORTFOLIO FOR \$88.5 MILLION

- A Total of 817,166 Square Feet Located in High-Barrier Infill Markets -
 - Year-to-Date Investment Volume Totals \$185.6 Million -
 - 50.3% Growth in Square Footage of Owned Portfolio Since IPO -

Los Angeles June 30, 2014 – Rexford Industrial Realty, Inc. (the "Company" or "Rexford Industrial") (NYSE: REXR), a real estate investment trust focused on owning and operating industrial properties located in Southern California infill markets, today announced that it has acquired a nine-property industrial portfolio, containing an aggregate of 817,166 square feet for \$88.5 million, or approximately \$108 per square foot.

"We are pleased to announce the acquisition of this high quality portfolio comprised of nine industrial properties in class "A" locations within our target Southern California infill markets" said Howard Schwimmer and Michael Frankel, Co-Chief Executive Officers of the Company. "The properties are strategically located within Los Angeles County, Orange County and San Diego County with convenient access to key regional, interstate, rail, and airport infrastructure to support local and regional distribution. The portfolio is currently 87% occupied. We plan to execute on a range of opportunities to drive occupancy while enhancing functionality, cash flow and value through strategic repositioning."

Portfolio Summary

The properties have been institutionally owned, and offer favorable locational attributes, functionality and opportunities for value-add improvements that maximize their competitive position within each respective submarket. Eight of the nine properties are 96% leased, in aggregate. The remaining property is 40% leased, with an immediate value-add opportunity to reposition the property in order to deliver a high-demand product capable of achieving higher rents .

Collectively, the nine properties contain 24 units leased to 17 tenants with staggered lease expirations, mitigating vacancy risk. As leases expire, select properties and units can be repositioned to increase loading and warehousing functionality, fire sprinkler capacity, among other enhancements. In addition, the properties generally offer flexibility to convert the existing units (which range from 10,303 SF – 108,703 SF) into smaller units with dock-high loading. The San Diego assets are currently configured with in-place but unused storefronts, existing dividing walls and restrooms, among other attributes, which will expedite and decrease the cost of a divisibility strategy.

The properties are further detailed below:

Name	SF	County	Submarket	Occupancy
Salt Lake	126,036	Los Angeles	San Gabriel Valley	100%
Valley	108,703	Los Angeles	San Gabriel Valley	100%
Hunter	119,692	Orange	North Orange County	100%
Alton	124,000	Orange	Orange County Airport	40%
9340 Cabot	86,564	San Diego	Miramar	84%
9404 Cabot	46,846	San Diego	Miramar	100%
9455 Cabot	96,840	San Diego	Miramar	84%
Distribution I	47,666	San Diego	Miramar	100%
Distribution II	60,819	San Diego	Miramar	100%
	817,166			87%

Property Information

San Gabriel Valley Submarket

Salt Lake is a 126,036 square foot industrial building located in City of Industry, California. The property, situated on 6.62 acres, was built in 1979, and includes clear heights of 24 to 30 feet, and dock-high loading. The property is currently 100% occupied with four tenants.

Valley is a 108,703 square foot industrial building located in Pomona, California. The property, situated on 7.50 acres, was built in 1980, and includes clear heights of 22 feet and excess land. The property is currently 100% occupied with one tenant.

Salt Lake and Valley are located in the San Gabriel Valley submarket of Los Angeles. The San Gabriel Valley submarket continues to experience steadily increasing levels of demand, with nearly 1.04 million square feet of gross activity recorded, and 535,786 square feet of positive net absorption in the first quarter of 2014. As a result of this activity, the vacancy rate in the San Gabriel Valley submarket stands at 1.8%, marking a 100 basis point decrease since the beginning of 2013.

North Orange County Submarket

Hunter is a 119,692 square foot industrial building located in Anaheim, California. The property, situated on 6.00 acres, was built in 1987, and includes clear heights of 24 feet. The property is currently 100% occupied with three tenants. Hunter is located in the North Orange County submarket. The North Orange County submarket is the largest submarket in Orange County and comprises over 110 million square feet of net rentable area. The North Orange County submarket was the most active Orange County submarket in the first quarter of 2014 with over 800,000 square feet of gross activity and 31,850 square feet of positive net absorption. The vacancy rate within the North Orange County submarket is 2.3%.

Orange County Airport

Alton is a 124,000 square foot industrial building located in Irvine, California. The property, situated on 5.82 acres, was built in 1974, and includes clear heights of 28 feet. The property is currently 40% occupied with one tenant. Alton is located in the Orange County Airport submarket, within the Irvine Business Complex. During the first quarter of 2014, the Orange County Airport submarket had 756,708 square feet of gross activity and 158,366 square feet of positive net absorption. The vacancy rate within the Orange County Airport submarket is 2.7%.

Central San Diego Submarket

9340 Cabot is an 86,564 square foot industrial building located in San Diego, California. The property, situated on 4.88 acres, was built in 1975/1976, and includes clear heights of 22 to 24 feet. The property is currently 84% occupied with two tenants.

9404 Cabot is a 46,846 square foot industrial building located in San Diego, California. The property, situated on 2.53 acres, was built in 1975/1976, and includes clear heights of 22 to 24 feet. The property is currently 100% occupied with one tenant.

9455 Cabot is a 96,840 square foot industrial building located in San Diego, California. The property, situated on 5.06 acres, was built in 1975/1976, and includes clear heights of 22 to 24 feet. The property is currently 84% occupied with one tenant.

Distribution I is a 47,666 square foot industrial building located in San Diego, California. The property, situated on 2.12 acres, was built in 1974, and includes clear heights of 27 feet. The property is currently 100% occupied with two tenants.

Distribution II is a 60,819 square foot industrial building located in San Diego, California. The property, situated on 3.08 acres, was built in 1983, and includes clear heights of 22 to 24 feet. The property is currently 100% occupied with two tenants.

The Cabot, Distribution I and Distribution II properties are located within the Miramar submarket of San Diego, the County's largest and most centrally located submarket for warehouse distribution. The buildings benefit from convenient access to the 805, 5 and 15 freeways and are less than 15 miles away from San Diego International Airport.

The Greater San Diego County market has experienced positive net absorption for the past seven consecutive quarters, with 1.2 million square feet being absorbed in the first quarter of 2014. The Central San Diego County submarket recorded 1,756,584 square feet of leasing activity and 554,046 square feet of positive net absorption in 2014, with a vacancy rate of 7.3% in the first quarter 2014.

About Rexford Industrial

Rexford Industrial is a real estate investment trust focused on owning and operating industrial properties in Southern California infill markets. The Company owns interests in 85 properties with approximately 9.1 million rentable square feet and manages an additional 20 properties with approximately 1.2 million rentable square feet. Additional information about the acquisition and the properties in the portfolio can be found in a Current Report on Form 8-K that the Company expects to file today with the Securities and Exchange Commission.

For additional information, visit www.rexfordindustrial.com

Market Data

Unless otherwise noted, all market data provided by CBRE.

Forward Looking Statements

This press release may contain forward-looking statements within the meaning of the federal securities laws, which are based on current expectations, forecasts and assumptions that involve risks and uncertainties that could cause actual outcomes and results to differ materially. Forward-looking

statements relate to expectations, beliefs, projections, future plans and strategies, anticipated events or trends and similar expressions concerning matters that are not historical facts. In some cases, you can identify forward-looking statements by the use of forward-looking terminology such as "may," "will," "should," "expects," "intends," "plans," "anticipates," "believes," "estimates," "predicts," or "potential" or the negative of these words and phrases or similar words or phrases which are predictions of or indicate future events or trends and which do not relate solely to historical matters. While forward-looking statements reflect the Company's good faith beliefs, assumptions and expectations, they are not guarantees of future performance. For a further discussion of these and other factors that could cause the Company's future results to differ materially from any forward-looking statements, see the reports and other filings by the Company with the U.S. Securities and Exchange Commission, including the Company's Annual Report on Form 10-K for the year ended December 31, 2013. The Company disclaims any obligation to publicly update or revise any forward-looking statement to reflect changes in underlying assumptions or factors, of new information, data or methods, future events or other changes

Contact:

Investor Relations:

Stephen Swett

424 256 2153 ext 401

investorrelations@rexfordindustrial.com